THE ROLE OF BLOOD GROUPS IN PREVENTING OR ENHANCING HIV INFECTION IN BOTSWANA

MODISA SEKHAMO MOTSWALEDI

2019

CAPE PENINSULA UNIVERSITY OF TECHNOLOGY

The role of blood groups in preventing or enhancing HIV infection in Botswana

By

Modisa Sekhamo Motswaledi

Thesis submitted in fulfilment of the requirements for the

Doctor of Philosophy: Biomedical Science

In the Faculty of Health and Wellness Sciences

At

Cape Peninsula University of Technology

Supervisor: Professor Oluwafemi O Oguntibeju

Co-supervisor: Professor Ishmael Kasvosve

Bellville

January 2019

CPUT copyright information

The dissertation/thesis may not be published either in part (in scholarly, scientific or technical journals), or as a whole (monograph), unless permission has been obtained from the University

DECLARATION

I, Modisa Sekhamo Motswaledi declare that the contents of this dissertation/thesis represent my own unaided work, and that the dissertation/thesis has not been previously submitted for academic examination towards any qualification. Furthermore, it represents my own opinions and not necessarily those of Cape Peninsula University of Technology.

from w

Signed:

Date: 2nd January 2019

ABSTRACT

Knowledge of population vulnerabilities to infectious diseases is key in managing many public health problems and for mapping appropriate strategies for prevention or intervention. A number of genes associated with resistance to HIV infection, such as the double deletion of 32 base pairs in the CCR5 gene , have been described and potentially account for lower HIV infections in some populations. The magnitude of the HIV pandemic in Sub-Saharan Africa warrants an investigation of the peculiar genetic factors that may have exacerbated its spread. An understanding of the genetic factors that are involved may aid in the development of specific strategies for prevention such as vaccine development, genetic counselling as well as gene therapy. The aim of this project was therefore to study the relationship between blood groups and HIV-infection in Botswana. HIV infection in Africa has not been linked to particular blood groups.

The project was undertaken in two phases from December 2012 to December 2017. In the first phase, 346 subjects of known HIV status (negative or positive) were phenotyped for 23 erythrocyte antigens via standard scientific procedures. A Chi-square analysis was used to determine those antigens associated with increased or reduced risk of HIV infection. In the second phase, 120 samples were phenotyped for the protective blood group (RhC) and the risk-associated groups (Lu^b and P₁). The samples were also characterized according to their laboratory results for viral load, lymphocyte sub-populations, complete blood count and blood chemistry, including total cholesterol. Some of the samples were also assessed for erythrocyte-associated viral RNA.

Generally, the prevalence of the blood groups in the general population in Botswana did not differ with the known prevalence for Africans broadly. Three novel findings were established. First, the blood group Rh(C) was associated with a 40% risk reduction for HIV infection. Immunologically, carriage of the C antigen was associated with a more robust cell-mediated immunity as evidenced by enhanced cytotoxic T cell counts. Moreover, this antigen occurred with a frequency lower than 30% in all countries where HIV prevalence was high. There was therefore an inverse relationship between Rh(C) frequency and HIV prevalence. An examination of reports from previous studies revealed that the pattern was consistent in Africa, Europe, Asia, South America and Caribbean countries. It appears that the population frequency of this antigen explains, at least in part, a genetic factor that puts some African populations at higher risk for HIV infection. These results are novel in that Rh antigens have not been previously associated with immunity in any reports.

Novel findings regarding the P₁ blood group was its association with a double risk for HIV infection. While the plasma viral load did not differ between P₁-positive and P₁-negative subjects, P₁-positive erythrocyte lysate yielded more viral RNA than P₁-negative cells, implying more intracellular HIV RNA. Intra-erythrocytic viral RNA was detected even in patients with an undetectable plasma viral load. Glycosphingolipids, of which P₁ is an example, have been documented to promote viral fusion to cells independent of CD4 receptors or other ligands. In at least one report, the presence of sphingolipids in lipid rafts was considered to be

sufficient for viral fusion. The presence of viral RNA even in erythrocyte lysates corroborates this phenomenon and potentially explains the double risk of HIV infection observed. The occurrence of HIV RNA in erythrocyte lysate is a novel finding that suggests a new viral reservoir. Apparently, P_1 has a high frequency among Africans and low in other races.

The Lu^b blood group was associated with triple risk for HIV infection and a lower absolute count of circulating monocytes. This monocytopenia was accentuated in the presence of higher cholesterol levels and in the presence of the Lu^b antigen. The occurrence of monocyte depletion in a setting of high cholesterol plus an adhesion molecule suggests conditions that favour atherosclerotic plagues. The increased risk of HIV-infection in Lu^b positive individuals is therefore likely linked to tethering and emigration into tissues, thus disseminating the virus to distant organs.

Our results suggest that blood group C protects against HIV infection by enhancing cytotoxic T-cell counts. Furthermore, the P₁ blood group potentially promotes HIV infection by creating viral reservoirs within the erythrocytes that can later be released when aging erythrocytes are trapped in the spleen and engulfed by splenic macrophages. The Lutheran blood group appears to enhance monocyte depletion in a setting of hypercholesterolemia and chronic inflammation created by HIV infection. So, while it is generally accepted that cardiovascular disease has a genetic predisposition, the results of the current study suggest that Lutheran blood group antigens may be a novel part of that genetic milieu of atherogenic factors in HIV-positive persons. The results further suggest the prophylactic possibility of suppressing endothelial expression of Lutheran antigens or modulating the expression of integrins on monocytic cells to reduce the adhesion risk. Finally, the ability of HIV to fuse to glycosphingolipids independent of CD4 receptors raises the possibility that HIV virions could be targeted by use of RNase-laden glycosphingolipid liposomes in post-exposure prophylaxis or where there is need to rapidly reduce the plasma viral load.

ACKNOWLEDGEMENTS

I would like to express sincere gratitude to the management and staff of the Botswana National Blood Transfusion Service, Julia Molefe Clinic, Botswana-Harvard HIV Reference Laboratory, and the Botswana-Harvard AIDS Institute, where samples were obtained and/or some of the tests performed.

Gratitude is also expressed to Professor Keoagile Thaga, Department of Statistics at the University of Botswana for his guidance with statistical analysis. Gratitude is also expressed to the Office of Research and Development, University of Botswana and Cape Peninsula University of Technology for supporting the Project financially.

The study would not have been successful without the kind permission of Botswana's Ministry of Health and Wellness as well the guidance provided by my supervisor and co-supervisor Professors O. O. Oguntibeju and Ishmael Kasvosve, respectively.

Finally, thanks to all my colleagues at the University of Botswana and the Botswana Institute for Clinical Laboratory Professionals who provided a listening ear and helpful critiques throughout the duration of this study. Special thanks to our very efficient librarians, Neo Kgotlhang and Dineo Ketshogileng for sourcing all the articles I ever requested for.

DEDICATION

To my dear wife Agnes and the children who endured my substantial absence in pursuit of this qualification and to my students, past, present and future who need to know that perseverance and hard work are normally and ultimately rewarded. Above all, to God the Creator of all things, in whom are hidden all the treasures of wisdom and knowledge.

TABLE OF CONTENTS

DECLARATION	iii
ABSTRACT	iv
ACKNOWLEDGEMENTS	vi
DEDICATION	vii
TABLE OF CONTENTS	/iii
List of Tables	xi
LIST OF FIGURES	xii
GLOSSARY	ciii
PREFACE	xv
CHAPTER ONE	. 1
INTRODUCTION	. 1
The HIV Pandemic in Botswana	.1
The Aim of the Study	. 2
References	.3
LITERATURE REVIEW	. 5
Summary:	.6
Summary: Introduction	.6
Summary: Introduction Physiologic Functions of RBC Antigens	.6 .6 .7
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations	.6 .6 .7 .8
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections.	.6 .6 .7 .8
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections. Blood Groups in HIV Infection	.6 .6 .7 .8 .9
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections. Blood Groups in HIV Infection Blood Groups in Other Viral Infections	.6 .6 .7 .8 .9 .9
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections. Blood Groups in HIV Infection Blood Groups in Other Viral Infections	.6 .6 .7 .8 .9 .9 .2
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections. Blood Groups in HIV Infection Blood Groups in Other Viral Infections Conclusion	.6 .6 .7 .8 .9 .9 L2 L2
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections Blood Groups in HIV Infection Blood Groups in Other Viral Infections Conclusion CONFLICT OF INTEREST.	.6 .6 .7 .8 .9 .9 L2 L2 L2
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections. Blood Groups in HIV Infection Blood Groups in Other Viral Infections Conclusion CONFLICT OF INTEREST. References.	.6 .7 .8 .9 .9 L2 L2 L3 L3
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections Blood Groups in HIV Infection Blood Groups in Other Viral Infections Conclusion CONFLICT OF INTEREST References CHAPTER THREE	.6 .7 .8 .9 .9 L2 L3 L3 L3 L3
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections Blood Groups in HIV Infection Blood Groups in Other Viral Infections Conclusion Conclusion CONFLICT OF INTEREST References CHAPTER THREE BLOOD GROUP ANTIGENS C, Lu ^b AND P ₁ MAY HAVE A ROLE IN HIV INFECTION IN AFRICANS	.6 .7 .8 .9 .9 .2 .12 .13 .13 .13 .13
Summary: Introduction Physiologic Functions of RBC Antigens Blood Group Disease Associations Blood Groups in HIV and other viral Infections Blood Groups in HIV Infection Blood Groups in Other Viral Infections Conclusion CONFLICT OF INTEREST References CHAPTER THREE BLOOD GROUP ANTIGENS C, Lu ^b AND P ₁ MAY HAVE A ROLE IN HIV INFECTION IN AFRICANS Abstract	.6 .7 .8 .9 .9 12 13 13 13 19 20

Methods	21
Results	23
Discussion	26
Conclusion	27
Acknowledgement:	28
References:	28
CHAPTER FOUR	32
Abstract:	33
Introduction	34
Materials and Methods	35
Results:	35
Discussion	37
Conclusion	38
Acknowledgments	38
Funding	38
Competing interests:	38
Author Contributions	38
References	38
CHAPTER FIVE	41
RHCE*C PHENOTYPE MAY EXPLAIN THE GEOGRAPHICAL DISTRIBUTION OF HIV INFECTIONS	41
Introduction	43
Materials and methods	44
Results	44
Discussion	47
Evidence from empirical HIV prevalence data	49
Conclusion	49
Acknowledgements	49
References	50
CHAPTER SIX	55
HIV-1 RNA IS DETECTABLE IN P1 ERYTHROCYTE LYSATE	55
Abstract	56
Introduction	57
Methods	58
Results	59

Discussion	60
Conclusion	61
Authors' disclosure	61
Acknowledgement	61
References	61
CHAPTER 7	64
GENERAL DISCUSSION AND CONCLUSION	64
Aims and Objectives	64
ABO Blood Groups	64
Blood Group C (rh', Rh2 or RHCE*C)	65
Duffy Blood Group	66
Lutheran Blood Group	66
The P ₁ Blood Group	67
Combination of risk of HIV-1 Infection based on the ratio of the risk to protective block	ood group 68
Conclusion	69
References	70
CHAPTER 8	74
RECOMMENDATIONS	74
Blood Group C	74
Lutheran Blood group (Lu/BCAM, LU2 or Lu ^b)	74
P1 Blood Group	74
APPENDICES	76
APPEDNDIX A – University of Botswana IRB Permit	76
APPENDIX B – Ministry of Health Permit	77
APPENDIX C – CPUT ETHICS PERMIT	78
APPENDIX D – National Blood Transfusion Permit	79
APPENDIX E – BOTSWANA-HARVARD HIV REFERENCE LABORATORY PERMIT	80
ADDENDUM: RESEARCH OUTPUT	81
Published articles	81
Articles submitted to journals	81
Conference Presentations	81

List of Tables

Chapter Three

Table 3. 1 – Prevalence of blood group antigens in HIV-infected and uninfected subjects
Table 3. 2 – Logistic regression model showing the effect of blood group antigens on the risk of HIV
infection
Table 3. 3 – Blood group antigens associated with HIV infection in multivariate logistic regression
analysis25

Chapter Four

No table of figures entries found.

Chapter Five

Table 5.1: Comparison of CD4 and CD8 in untreated, HIV-infected individuals and normal controls
Table 5.2: Comparison of mean Lymphocyte sub-populations in untreated HIV-infected patients 45
Table 5.3: Comparison of RHCE*C with HIV prevalence across geographical regions 47
Figure 5.1: Relationship between viral load and CD8% in C-positive subjects46
Chapter Six
Table 6.1 - Comparison of RNA-PCR results in untreated HIV-1-infected patients.599 Table 6.2 : - Comparison of P1 frequency with HIV prevalence in various countries60

Chapter Seven

Table 7.1: Country comparison of	of the risk of HIV-1	Infection based o	n the ratio of the risk	y
to protective blood group				69

LIST OF FIGURES

CHAPTER FIVE

GLOSSARY

ABBREVIATION	DEFINITION / EXPLANATION
АСНАР	African Comprehensive HIV/AIDS Partnerships
AHG	Antihuman Globulin antibody
ART	Anti-retroviral Therapy
BHHRL	Botswana-Harvard HIV reference Laboratory
CCL5	Cysteine-Cysteine Ligand Type 5
CCR5	Cysteine-cysteine Receptor Type 5
CCR5Δ32/Δ32	CCR5 mutation with double deletion of 32 base pairs
CHD	Coronary Heart Disease
CR1	Complement Receptor Type 1
CTL	Cytotoxic T-Lymphocytes
CVD	Cardio-vascular Disease
CXCR4	Cysteine-X-Cysteine Receptor Type 4
DARC	Duffy Antigen Receptor for Chemokines
DNA	Deoxyribonucleic acid
EDTA	Ethylene diamine-tetraacetic acid
EGF	Epidermal Growth Factor
E-HIV	Erythrocyte-associated Human Immunodeficiency Virus
Fy ^a or Fy ^b	antithetical Duffy "a" or "b" antigens
GSL	Glycosphingolipid
HAART	Highly-active anti-retroviral therapy
HDL	High Density Lipoprotein (Good cholesterol)
HIV	Human Immunodeficiency Virus
IQR	Interquartile Range
ISBT	International Society for Blood Transfusion
Jk ^a /Jk ^b	Antithetical antigens of the Kidd blood group system
LDL	Low Density Lipoprotein (Bad cholesterol)

Lewis "a"/Lewis "b" antithetical blood groups
Lu ^a
Lu ^b
Lutheran "a" blood group antigen antithetical to Lu ^b
Lutheran antigen antithetical to Lu ^a
Lutheran/Basal Cell Adhesion Molecule
Landsteiner-Weiner
Monocyte Chemo-attractant Protein-1
Major Histocompatibility Complex Class I (or II)
Polymerase Chain Reaction
Regulated on Activation Normal T-Cell Expressed and Secreted
Red Blood Cell
Weiner designation for Rh (C) and (E) antigens and genes
Rh-associated Antigen
"C" antigen in the Rhesus blood group
ISBT name for Rh(C)
Another designation of the "C" antigen in the Rh system
Ribonucleic acid
Signal-regulatory Protein α
Total Cholesterol
Joint United Nations Programme on HIV/ Acquired Immune Deficiency Syndrome
Viral Load
HIV strains that use the CXCR4 co-receptor
HIV strains that utilize CCR5 co-receptor

PREFACE

This dissertation is presented in article format. The chapters appear in the format of the journals in which they have been published or submitted for publication.

Chapter 1 is the general Introduction, which basically puts the problem in perspective. Here the attention of the reader is drawn to the magnitude of the HIV and AIDS problem globally as well as in Botswana, which is the particular setting of the current research project. Epidemiologic data from UNAIDS on both the global and local situations is presented to underscore the importance of the subject. The role of erythrocyte antigens as binding sites for infectious agents is discussed with examples of other infectious agents that specifically bind to blood group antigens. The chapter then makes a brief reference to the role of red blood cells (erythrocytes) in HIV-infection and singles out red cell antigens as potential binding entities in the transmission of HIV to susceptible cells, thus justifying the relevance of the blood groups in HIV research.

Chapter 2 is a more detailed review of literature on the role of the erythrocytes in HIV and other diseases. A review article from this thesis was published in BioMed Research International journal in 2013. The paper points to the fact that frequencies of blood groups differ among populations partly due to breeding practices as well as disease-induced selection pressure. Individuals whose cell antigens provide binding sites for pathogenic microorganisms will naturally be susceptible to negative selection. The paper further discusses a number of erythrocyte antigens that have been linked to diseases, prototype of which is the Duffy antigen that has been eliminated to a significant level in individuals living in endemic malaria areas. Controversies in this area of research are also addressed.

The physiologic function of erythrocyte antigens is discussed to highlight relevance to normal function in general, and to innate immune responses. A detailed discussion of the role of blood groups in specific diseases, including potential mechanisms of cytopathic effects in HIV-infection concludes the paper.

Chapter 3 (This part of the research work has been published as an article in the journal: PLOS One). The paper reports the screening of 346 HIV-infected and uninfected controls for 23 antigens, these being A, B, "O", D, C, c, E, e, Fy^a, Fy^b, Jk^a, Jk^b, Kp^a, Kp^b, Le^a, Le^b, Lu^a, Lu_b, M,N, P₁, S, and s. In this study, one blood group, Rh(C), also designated as rh' or Rh2, was found to be protective while two blood groups, P₁ and Lu^b, were associated with significantly increased risk for HIV-infection. Our data also contributed to the hotly contested role of the Duffy blood groups as risk factor for HIV infection as none was found in our study. The rest of the blood groups investigated were not associated with significant risk or protection. Another important finding of this investigation was that the protective blood group Rh(C) occurred much less frequently among Africans than it did among Caucasians and Northern Indians while the risky blood groups, P₁ and Lu^b had prevalence in excess of 90%. Information available in the literature was helpful in relating the biochemical nature of the risky blood groups to the study observations. However, the protective mechanism for Rh(C) appeared inexplicable in the

absence of information on the biological role of this antigen. Mechanisms were therefore suggested for the risk-association of these antigens.

Chapter 4 is an article that is under review by the African Journal for Laboratory Medicine. The substance of this article sought to gain insight into the mechanisms accounting for the risk observed in the Lutheran blood group. The Lutheran blood group antigens are members of the immunoglobulin superfamily and act as adhesion molecules expressed on monocytes and endothelial cells. Carriage of the Lu^b antigen (also designated LU2) was associated with a lower monocyte count, and the count diminished with increasing cholesterol levels. This cholesterol-associated monocyte depletion was accentuated in Lu^b-positive individuals but virtually absent in Lu^b-negative subjects. The depletion of monocytes in a setting of high cholesterol and chronic inflammation points to an atherogenic environment and potentially explains why HIV-infected individuals are prone to cardiovascular disease. Moreover, if this depletion truly represents extravasation of monocytes, this could explain the increased risk of infection as infected monocytes migrate and introduce the virus to other anatomical sites. Novel findings here include both the increased risk of infection in LU2 individuals as well as the monocyte-depleting effect of this blood group.

Chapter 5 presents the findings as they relate to blood group C (or Rh2). The main finding here was the association of Rh2 expression with both a lowered risk for HIV-infection in the context of enhanced cytotoxic T-cells counts. These cells are important for the control of viraemia and are known to confer resistance in long term progressors and non-progressors. This is the first report linking an antigen in the Rh system with immune function.

Chapter 6 reports on the findings relating to the P_1 blood group. Since P_1 is a glycosphingolipid capable of promoting viral fusion with cell membranes, evidence for intra-erythrocytic viral RNA was sought. The study confirmed the presence of erythrocyte-associated HIV RNA in patients with or without detectable plasma viral load. However, this viral pool was not detectable in every patient with a detectable or even high viral load. P_1 blood group was associated with increased intra-erythrocytic HIV-1 RNA. This too was consistent with the role of this blood group, a glycosphingolipid, which enhances the fusion of the virus to cells, thus promoting infection.

Based on these studies, it can be concluded that there is adequate scientific evidence for the role of blood groups C, P_1 and Lu^b in enhancing or preventing HIV-infection, thus the purpose of this study was achieved.

Chapter 7 presents a general discussion and conclusions. Here all the main novelties are highlighted with potential mechanisms. The prevalence data is compared with HIV prevalence in various countries to show that the findings in this study are consistent with severity of the disease in those countries or communities. Uganda and Botswana stand out in that respect.

Chapter 8 provides recommendations for further enquiries to elucidate specific research questions emanating from this study.

CHAPTER ONE

INTRODUCTION

The HIV Pandemic in Botswana

The HIV and AIDS pandemic is one of the world's worst health-related catastrophes in recent history. UNAIDS estimates that since its start in the early 1980's, the pandemic has infected more than 70 million people and claimed at least half the lives of those infected as at the end of 2016 (UNAIDS, 2017a). It is further estimated that 36.7 million people are living with the virus world-wide, with 1.8 million new infections experienced in 2016. The report further takes cognizance of the notable variation between countries and regions, but Sub-Saharan Africa is the most severely affected region and bearing 2/3 of global infections (UNAIDS, 2017a).

On the economic front, the virus has prompted the commitment of massive resources for prevention and treatment programmes, with US\$19.1 billion having been committed to support low and middle income countries in 2016 alone (UNAIDS, 2017a).

Botswana is one of the worst-affected countries by the HIV pandemic. During the period 1992 to 1994, the country experienced a dramatic increase in the HIV prevalence with urban areas such as Gaborone and Francistown jumping from 15% to 28% and from 23% to 30% (Macdonald, 1996), respectively.

In 2002, the Government of Botswana introduced the Masa Project, in which anti-retroviral treatment was offered to eligible individuals (Farahani et al., 2014). Initially, the criterion for initiation of combination anti-retroviral treatment (cART) included a cut-off CD4 cell count (initially 200 cells/ μ L). This programme radically reduced mortality rates from 63% in 2002 to 0.8% in 2010 (Farahani et al., 2014). In 2004, the government of Botswana scaled up the anti-retroviral (ARV) programme and began to offer HIV-testing at health service points as well as voluntary testing and counselling centres around the country. The clinical efforts were also accompanied by intense clinical trials and other forms of research that developed robust interventions such as the Prevention of mother-to-child (PMTCT) study (Shapiro et al., 2013) and the micronutrient studies that added to the quality of the clinical interventions. Based on continuous research, strategies were modified to obtain best practices (Zash et al., 2016). Through sustained and focused clinical and laboratory-based studies (Novitsky et al., 2010), the Botswana Programme ultimately graduated to a "Treatment as Prevention" or "Treat-all" programme (Botswana Ministry of Health, 2016) based on the observation that individuals who achieved viral suppression were less likely to transmit the virus to their partners(Hayes et al., 2011).

In 2010, the country was reported to be the world's second worst hit country with prevalence rates then at 17.6% and as high as 20% in women (Kandala et al., 2012). The current UNAIDS reports indicate that at least 350,000 individuals in Botswana are living with the virus in a population barely exceeding 2 million, with a prevalence of 22.2% in the age group 15-49 years of age (UNAIDS, 2017b). This astounding spread of the virus may have been due to peculiar genetics of the population in addition to other socio-economic factors.

Statement of the Problem

Population genetics is an important consideration in studying the epidemiology of infectious diseases. Blood groups are genetically-determined antigens carried on red blood cells or erythrocytes that are inherited as co-dominant Mendelian traits. Blood groups are mainly important in blood transfusion because they determine compatibility of donor and recipient cells. However, some blood group antigens function as receptors for infectious agents such as malaria (Hadley and Peiper, 1997), parvo virus (Brown et al., 1993) and *Helicobacter pylori* (Anstee, 2010, Aspholm-Hurtig et al., 2004). In recent years, the binding of the human immunodeficiency virus to erythrocytes and the apparent increased infectivity of such bound virus has sparked interest in this potential viral reservoir (Puri et al., 1998, Hammache et al., 1999, Beck and Alving, 2011). Blood groups therefore emerge as logical binding sites for the virus on erythrocytes and candidate molecules for scientific enquiry.

The aim of this research project was therefore to identify blood group antigens associated with increased or decreased HIV infection and, where possible, elucidate the underlying mechanisms. The results of the study were intended to help clarify the genetic basis of the HIV epidemic as it relates to blood groups in the Botswana population. The discovery of HIV-resistant blood groups could open a number of frontiers for prevention and treatment of HIV. Couples seeking artificial fertilization may have the option of semen of the appropriate blood group to produce HIV-resistant offspring. Furthermore, research efforts could be focussed on the use of hematopoietic stem cells as these are better tolerated in the developing foetus. Consideration could also be made for using blood transfusion to treat infections or as pre- or post-exposure prophylaxis.

The Aim of the Study

The overarching aim of this study is to identify candidate antigens that may predispose to or impact on the risk for HIV infection, and if any are identified, to investigate potential mechanisms by which protection or risk is conferred.

Specific objectives

The objectives of the study were:

- a. Estimate the population frequencies of 23 blood group antigens in Botswana.
- b. Compare the occurrence of blood groups in HIV-infected versus uninfected individuals.

c. Investigate the mechanism responsible for increased or reduced susceptibility to HIV infection.

Hypothesis:

Ho: Blood groups are not associated with HIV infection.

Ha: Some blood groups are associated with HIV infection.

In the light of the above, the research sought to answer the following questions:

- 1. What are the frequencies of the 23 blood group antigens in the Botswana population?
- 2. Is there any difference in occurrence of HIV among the different blood groups?
- 3. What mechanisms account for the difference in susceptibility to HIV infection among the blood groups?

References

ANSTEE, D. J. 2010. The relationship between blood groups and disease. *Blood*, 115, 4635-4643

Aspholm-Hurtig, M., Dailide, G., Lahmann, M., Kalia, A., Ilver, D., Roche, N., et al. 2004. Functional adaptation of BabA, the H. pylori ABO blood group antigen binding adhesin. *Science*, 305, 519-22.

Beck, Z. & Alving, C. R. 2011. Interactions of infectious HIV-1 virions with erythrocytes: implications for HIV-1 pathogenesis and infectivity. *In:* CHANG, T. L. (ed.) *HIV-Host Interactions*. Rockville: In-Tech.

Brown, K. E., Anderson, S. M. & Young, N. S. 1993. Erythrocyte P antigen: cellular receptor for B19 parvovirus. *Science*, 262, 114-7.

Farahani, M., Vable, A., Lebelonyane, R., Seipone, K., Anderson, M., Avalos, A., et al. 2014. Outcomes of the Botswana national HIV/AIDS treatment programme from 2002 to 2010: a longitudinal analysis. *The Lancet Global Health*, 2, e44-e50.

Hadley, T. J. & Peiper, A. C. 1997. From Malaria to chemokine receptor; the emerging physiologic role of the Duffy blood group antigen. *Blood Reviews*, 89, 3077-91.

Hammache, D., Yahi, N., Maresca, M., Pie'Roni, G. & Fantini, J. 1999. Human erythrocyte glycosphingolipids as alternative cofactors for human immunodeficiency virus type 1 (HIV-1) entry: evidence for CD4-induced interactions between HIV-1 gp120 and reconstituted membrane microdomains of glycosphingolipids (Gb3 and GM3). *Journal of Virology*, 73, 5244–5248.

Hayes, R., Sabapathy, K. & Fidler, S. 2011. Universal Testing and Treatment as an HIV Prevention Strategy: Research Questions and Methods. *Current HIV Research*, 9, 429-445.

Kandala, N.-B., Campbell, E. K., Rakgoasi, S. D., Madi-Segwagwe, B. C. & Fako, T. T. 2012. The geography of HIV/AIDS prevalence rates in Botswana. *HIV AIDS (Auckl)*, 4, 95-102.

Macdonald, D. S. 1996. Notes on the socio-economic and cultural factors influencing the transmission of HIV in Botswana. *Soc Sci Med*, 42, 1325-1333. Novitsky, V., Wang, R., Bussmann, H., Lockman, S., Baum, M., Shapiro, R., et al. 2010. HIV-1 Subtype C-Infected Individuals Maintaining High Viral Load as Potential Targets for the "Test-and-Treat" Approach to Reduce HIV Transmission. *PLoS ONE*, *5*, e10148.

Puri, A., Hug, P., Mun^oz-Barroso, I. & Blumenthal, R. 1998. Human erythrocyte glycolipids promote HIV-1 envelope glycoprotein-mediated fusion of CD4+ cells. *Biochemical and Biophysical Research Communications*, 242, 219–225.

Shapiro, R. L., Kitch, D., Ogwu, A., Hughes, M. D., Lockman, S., Powis, K., et al. 2013. HIV transmission and 24-month survival in a randomized trial of HAART to prevent MTCT during pregnancy and breastfeeding in Botswana (The Mma Bana Study). *AIDS (London, England)*, 27, 1911-1920.

UNAIDS 2017a. Fact sheet - Latest statistics on the status of the AIDS epidemic.

UNAIDS. 2017b. *Fact Sheet - Latest statistics on the status of the AIDS epidemic* [Online]. Available: <u>http://www.unaids.org/en/regionscountries/countries/botswana/</u>.

Zash, R., Souda, S., Leidner, J., Ribaudo, H., Binda, K., Moyo, S., et al. 2016. HIV-exposed children account for more than half of 24-month mortality in Botswana. *BMC Pediatrics*, 16, 103.

CHAPTER TWO

LITERATURE REVIEW

THE ROLE OF RED BLOOD CELLS IN ENHANCING OR PREVENTING HIV INFECTION AND OTHER DISEASES

^{1,2}Modisa S Motswaledi, ¹Ishmael Kasvosve & ²Oluwafemi O Oguntibeju

¹University of Botswana, Department of Medical Laboratory Sciences, Faculty of Health Sciences, Gaborone, Botswana. ²Oxidative Stress Research Centre, Department of Biomedical Sciences, Faculty of Health & Wellness Sciences, Cape Peninsula University of Technology, Faculty of Health and Wellness Sciences, South Africa, Bellville, 7535.

Address correspondence to: <u>oguntibejuo@cput.ac.za</u>, <u>bejufemi@yahoo.co.uk</u> Tel: +27219538495, Fax: +27219538495

Modisa S Motswaledi, MT (ASCP), MSc, BSc, (modisa.motswaledi@mopipi.ub.bw)

Ishmael Kasvosve, PhD, MSc, BSc (ishmael.kasvosve@mopipi.ub.bw)

Oluwafemi O Oguntibeju, DTech, FIMLS, FACBS, FIBMS, CSci (<u>oguntibejuo@cput.ac.za</u>, bejufemi@yahoo.co.uk)

Published in BioMed Research International, 2013, Volume 2013, Article ID 758682, 7 pages, http://dx.doi.org/10.1155/2013/758682.

Summary:

AIM: To highlight the apparently neglected role of red blood cell antigens in the epidemiology of infectious diseases, with particular focus on HIV, with the prime objective of stimulating research in this area.

METHOD: PubMed database was searched for relevant articles from 1984 to 2013, this being the era covering active HIV research. However, the review was limited by the relative scarcity of articles on the subject, and only articles written in English were reviewed.

RESULTS: Substantial documentary evidence attests to the ability of infectious agents to bind to erythrocyte-borne molecules, thereby facilitating infection.

CONCLUSION: Mature erythrocytes, by their lack of a nucleus are not candidates for HIV infection. Emerging information, however, suggests that red blood cell antigens may be important in the pathogenesis of many diseases, including HIV as they bind free viruses and viral immune complexes and are instrumental in transmitting the virus to susceptible cells. The role of erythrocyte surface molecules acting as viral binding sites is discussed in light of current scientific knowledge.

Key Words: Blood groups, HIV infection, Erythrocyte-bound HIV, chemokines, disease associations

Introduction

Blood groups have been known to exist for centuries ever since man entertained the possibility of replacing blood loss through transfusion. They are responsible for determining compatibility of blood in transfusion medicine and are also responsible for foetal loss in haemolytic disease affecting the foetus or new born (Bowman, 1996).

Blood groups are inherited as Mendelian co-dominant(Neil et al., 2003) traits and should be expected to occur in somewhat comparable frequencies in the human race. However, differences do exist in the distribution of blood groups in various human populations. This has been largely attributed to selection pressure as endemic diseases appear to have a predilection for selected blood groups leading to the demise of individuals bearing those susceptible blood group antigens(Anstee, 2010, Le Pendu et al., 2006). But certainly, there must be more to blood groups than just causing problems in blood transfusion and predisposing to myriad self-annihilation prospects.

On the other hand, HIV has emerged as one of the major public health concerns of the 21st century. Some genetic factors have been cited as contributors to HIV susceptibility or resistance, among them blood groups such as ABO/Rh(Sayal et al., 1996, Arendrup et al., 1991, Abdulazeez et al., 2008, John et al., 2011, Neil et al., 2003), Duffy(Bolton and Garry, 2011, Lachgar et al., 1998, Ramsuran et al., 2011, He et al., 2008) and P^k(Lund et al., 2009). However, the matter has not been without controversy as some investigators have reported findings to the contrary(Ukaejiofo and Nubila, 2006, Winkler et al., 2009) against what appears to be conclusive evidence of red cell interaction with the virus(Bolton and Garry, 2011, Beck

et al., 2009, Puri et al., 1998, Hammache et al., 1999). Blood groups therefore appear to have a contribution to public health, at least in the area of infectious disease, which makes it imperative to synthesize available knowledge in an attempt to decipher the extent to which red blood cell (RBC) antigens are involved in HIV epidemiology and to unveil avenues for future research; here follows a synthesis of current research on the beneficial and detrimental effects of RBC antigens with probable molecular explanations.

Methods:

A literature search was performed on the PubMed database. This was achieved by using the phrases "erythrocyte blood groups AND HIV" (81 articles) or "red cell antigen, blood groups AND HIV" (60 articles). A manual search was also done using Google Scholar and supplemented by other original articles referenced by various authors.

Physiologic Functions of RBC Antigens

The physiologic functions of most blood group antigens are unknown (Greenwell, 1997). Few antigens, however, such as the Kidd (urea transporter), Diego (anion exchanger), Colton (water channel), Cromer (Decay Accelerating Factor) and Duffy (chemokine receptor) have had their physiologic functions somewhat elucidated (Mudad and Telen, 1996, Telen, 1995). The importance of such functions remains a matter of debate though, since individuals who are negative for these blood group antigens live an apparently normal life (Daniels, 1999).

A lot has been documented about the Duffy blood group as a chemokine receptor, a function that has led to its designation as the Duffy antigen receptor for chemokines (DARC), and to the investigation of its potential role in modulating the inflammatory response (Hadley and Peiper, 1997, Lachgar et al., 1998, Nibbs et al., 2003, Gardner et al., 2004, Bolton and Garry, 2011). Chemokines are important for recruiting white blood cells to the site of infection or inflammation by binding to their receptors. However, HIV also uses these chemokine receptors as co-receptors for its entry into permissive cells. It is known that R5 viruses utilize CCR5 while R4 viruses use CXCR4 as a co-receptor. The binding of HIV to the co-receptor and its subsequent entry into the cell is inhibited by chemokines which act as competitors for the binding site (Lachgar et al., 1998, Bolton and Garry, 2011). The Duffy antigen receptor for chemokines binds a wide range of chemokines and thus competes with white blood cells for chemo-attractants as well as increase cell exposure to HIV infection. DARC also binds other inflammatory mediators such as interleukin 8 (IL-8), RANTES and monocyte chemo-attractant protein-1 (MCP-1) (Telen et al., 1990, Reich et al., 2009, Cartron, 2010), all of which are important in contributing to a successful defence against pathogens.

Red blood cells also bear the CD44 antigen, a ligand for hyaluronic acid. Binding of hyaluronic acid to this RBC antigen would have the significance of dampening its inflammatory effect. As in the case of the Duffy antigen binding to chemokines, this erythrocytic property may be important in absorbing excess stimulant signals intended for inflammatory cells, thereby modulating immune responses (Telen, 1995). In cases where the stimulant is produced in small amounts, it is plausible that erythrocyte antigens, by their sheer numbers, may out-compete the intended cells, leading to an immunological hypo-response.

Human red blood cells not only transport oxygen and carbon dioxide, but are important in the elimination of many molecules from circulation, including products of immunologic reactions. RBCs possess adhesion molecules that bind complement fragments(Horakova et al., 2004) and are therefore important for capturing and delivering such complexes to the reticulo-endothelial system for elimination from circulation. In this regard, Horakova et al.(Horakova et al., 2004) investigated the binding of HIV immune complexes to erythrocytes. Their investigations led to the conclusion that at least three mechanisms are involved, these being; the binding of complement-opsonized immune complexes via CR1, direct binding of the virus in a complement-dependent manner but in which specific antibodies were not required, and a third mode in which complement was not required at all. Their conclusion was that the third mode of binding might be instrumental in the spread of infection at the time of primary infection when specific antibodies would not have been formed yet.

Some RBC antigens are also known to bind inflammatory mediators with potential to affect innate immune responses (Telen, 1995, Hadley and Peiper, 1997). These RBC antigens absorb and limit the amount of chemokine stimulant activity (Reich et al., 2009). Indeed it has been demonstrated that mice lacking the Duffy antigens exhibit a greater inflammatory response to CXC and CC chemokines compared to homozygotes or heterozygotes (Dawson et al., 2000). This is consistent with the role of Duffy as a "chemokine sink" created by chemokine-binding sites on DARC(+) erythrocytes. But, paradoxically, the absence of the Duffy antigens has also been associated with a low neutrophil count (Reich et al., 2009, Ramsuran et al., 2011). Given the observation that DARC also binds HIV and is capable of transfecting CD4 cells (Lachgar et al., 1998), it is probable that HIV-binding to erythrocytes represents an action of various molecules acting in synergy.

Blood Group Disease Associations

On the pathological front, however, certain blood groups have been associated with diseases. Individuals of blood group A, for example, are known to be more susceptible to coronary heart disease (CHD) independent of known risk factors than other ABO blood groups (Wazirali, 2005). This is further corroborated by the observation that group A individuals also have higher levels of low density lipoprotein (LDL) cholesterol (Greenwell, 1997). Although the molecular aspects of this observation still remains to be elucidated, the observation implies that this antigen has potential to either influence synthesis of, or inhibit the natural metabolism of these lipids, thus predisposing individuals to CHD. The same study, on the other hand, reported a lower risk for this condition among group O individuals.

Additionally, blood group A and B are known to be highly susceptible to thrombotic disorders in contrast to group O individuals who are more at risk for bleeding than thrombotic events(Clark and Wu, 2011, Schleef et al., 2005, Lourenço et al., 1996). This observation has been attributed to higher levels of Factor VIII and von Willebrand factor in A and B individuals while the levels may be as low as 30% in group O individuals. Blood groups A and B antigens have been demonstrated on these coagulation molecules and are thought to prolong their half-life, leading to higher concentration in non-O individuals (Jenkins, 2006).

Blood Group A has also been associated with malignancies such as cancer of the ovary, cervix, rectum, breast, stomach and leukemia (Greenwell, 1997). This is thought to be due to an abundance of high-affinity binding sites for epidermal growth factor (EGF) on blood group A red blood cells compared to blood groups O and B (Engelmann et al., 1992). Since blood groups may be expressed on tissues other than red cells(Telen, 1995), it is plausible that the binding of EGF to these binding sites may indeed promote cancer development. Conversely, some malignancies have been observed to positively or negatively modulate blood group antigen expression to varying degrees depending on the anatomical site (Greenwell, 1997). Thus, gastric carcinomas were found to be associated with increased expression of ABH and Le^a antigens while colonic cancers were associated with expression of Le^b and low expression of the other antigens. In other studies, enhanced fucosyltransferase activity was reported in endometrial cancer leading to unexpected expression of Le^a, Le^b and H antigens (Ravn, 1992, Ravn et al., 1994).

The Duffy blood group is probably one of the most studied blood groups in relation to disease association. Individuals who are homozygotes or heterozygotes for the Duffy antigens, Fy^{a} or Fy^{b} , are vulnerable to malaria parasites *Plasmodium vivax* and *Plasmodium knowlesi*. Individuals that lack both Duffy antigens, Fy(a-b-) and therefore lack the receptor for malaria parasites are constitutively resistant to these forms of malaria (Barnwell et al., 1989, Telen, 1995).

Blood Groups in HIV and other viral Infections

Blood Groups in HIV Infection

Most research on HIV has focused on cells of the immune system to the exclusion of viral interactions with red blood cells. However, emerging evidence, suggests that red blood cells may be important in the pathogenesis of HIV as they enhance viral infectivity by binding free viruses (Beck et al., 2009) as well as viral immune complexes (Hess et al., 2002, Horakova et al., 2004, Beck et al., 2009), and through such binding transfect HIV-susceptible cells (Hess et al., 2002, Horakova et al., 2002, Horakova et al., 2004).

HIV infection has been reported to occur in select blood groups in some regions of the world. A study by Sayal *et al.* in India reported a preponderance for infection in group O Rh(D)-positive men and least among Group B positive and D-negative ones. However, a close examination of the results reveals insufficient statistical analysis rendering the differences statistically insignificant. Similar studies by Nneli *et al* (Nneli et al., 2004) and John et al (John et al., 2011) suffered similar deficiencies. In these studies, group O positive individuals were thought to be highly susceptible, but again the studies lacked the statistical rigor to indicate the level of significance and have been contradicted by other investigators (Ukaejiofo and Nubila, 2006). A statistical interrogation of the data in each of the above case does not support the conclusion of statistical significance.

It would then appear that current scientific information does not support a potential role for ABO blood groups in HIV infection. In fact, evidence from other studies would suggest the contrary for group O individuals. Since HIV virions have been shown to acquire the blood group antigens of the infected individuals(Rachkewich et al., 1978, Neil et al., 2003, Arendrup

et al., 1991), such virions would be neutralized by naturally-occurring antibodies in group O individuals, thus offering protection in blood group-discordant couples (Neil et al., 2003). It is noted, however, that this protection will not be available if the source of infection was of a similar blood group. Moreover, given the apparently uniform risk of infection among ABO blood groups, it is doubtful if this neutralization is of any clinical consequence, especially with reference to HIV-1 infection. It remains to be demonstrated then whether HIV from blood group A or B is able to infect group O CD4 cells.

In a separate study, Abdulazeez and colleagues (Abdulazeez et al., 2008) working with HIV-1 and 2 reported a higher prevalence of HIV-2 (71.4%) compared to HIV-1 (7.1%) in the AB blood group and that Rh(D) positive (97.8%) was more susceptible than D negative (2.2%). However, these findings are yet to be confirmed by other investigators.

The lack of direct empirical evidence for ABO blood groups does not, however, obliterate the possibility of associations in other blood groups. Since secreted blood group substances can be adsorbed onto lymphocyte membranes (Rachkewich et al., 1978, Arendrup et al., 1991, Neil et al., 2003), the presence of these antigens could potentially alter cell behavior. Blood group antigens, being glycoproteins and glycolipids, are highly charged molecules that are bound to affect their molecular micro-environment, including protein conformation and receptor/CD4 localization and function (Greenwell, 1997, Lund et al., 2009). Glycosphingolipids and glycoproteins have in fact been demonstrated to facilitate fusion of HIV-1 with CD4 cells by independent investigators, thus acting as alternative co-receptors for the virus (Puri et al., 1998, Hammache et al., 1999). There exists a theoretical possibility therefore, that membrane-bound blood group substances on CD4-positive cells may thus affect the affinity of viral-binding proteins and viral infectivity, promoting or diminishing cell susceptibility to infection. Apparently, both the promotive and inhibitory effects of blood groups in HIV infection have been documented. The Duffy antigen receptor for chemokines (DARC) has been reported as a binding site for HIV-1 on red blood cells, which binding increases infectivity for permissive cells(Lachgar et al., 1998, He et al., 2008). Although some investigators have refuted the role of this antigen in HIV-1 infection(Winkler et al., 2009, Horne et al., 2009), the preponderance of evidence is heavily in favor of this role, including recent findings of sequence similarities between HIV-1 V3 loop and the Plasmodium vivax Duffy-binding protein, both of which bind to DARC.

Furthermore, the Duffy Null phenotype has been associated with increased susceptibility to HIV infection among African Americans, elevating the odds of acquiring HIV by as much as 40%, while those negative for the antigen were observed to exhibit slower progression to AIDS (He, 2008). Although these findings were later refuted by other investigators (Winkler, 2009, Horne et al., 2009), there is persuasive evidence suggesting the role of DARC in binding the virus and facilitating transfection of permissive cells(Ramsuran et al., 2011, Lachgar et al., 1998, Bolton and Garry, 2011). These reports were given further credibility by findings that the Duffy-binding protein of *Plasmodium vivax* bore sequence similarity with the V3 loop of HIV1(Bolton and Garry, 2011).

He et al (He et al., 2008) reported that the DARC+ phenotype was associated not only with higher rates of HIV-1 infection, but also that DARC+ red cells transfected HIV-susceptible

cells with X4 variants more efficiently than it did with X5 variants. Blocking of the DARC with monoclonal antibodies or other natural ligands significantly reduced viral binding, thus confirming the role of DARC in the interaction. However, the blocking of the DARC did not totally obliterate HIV binding to erythrocytes, thus suggesting possibilities of involvement of other binding molecules as hypothesized by Beck et al(Beck et al., 2009). Similar findings were also reported from a study involving commercial sex workers in South Africa, where investigators found that a Duffy-null phenotype with low neutrophil counts was predictive of HIV-1 sero-conversion (Ramsuran et al., 2011).

Another such molecule is the P^k RBC antigen, which has been reported to confer resistance to HIV infection in lymphocytes *in vivo* and *in vitro*. In this Canadian study, the level of expression of P^k antigen on peripheral mononuclear cells (PBMC) was positively correlated with anti-viral protection (Lund et al., 2009). The P^k antigen has a low frequency (one in a million)(Lund et al., 2009) among Caucasians and is thought to have contributed to reduction of HIV spread among such populations. The CCR5 Δ 32/ Δ 32 is another mutation that is prevalent in the same population at 1-3% (He et al., 2008) and is known to confer resistance to HIV infection. While it has been argued that the lack of P^k among Africans may not account for the difference in the epidemiology of HIV in sub-Saharan Africa (given its low frequency among Caucasians), one could nevertheless consider the combined effect of both genes in the same population and the number of contacts that could have been generated if such genes were not operational, compared to a population that lacks both.

In addition, Beck et al demonstrated that human red blood cells bind HIV virions independent of antibody or complement. The RBC selectively bound infectious HIV virions as opposed to other viral components and removed all infectivity from the suspension medium. The HIVbinding to erythrocytes was calcium-dependent but complement and CD4-independent. Moreover, RBC-bound virions were 100-fold more infectious to CD4 cells than cell-free particles(Beck et al., 2009), leading them to conclude that red blood cells could constitute a viral reservoir for trans-infection of CD4 cells. The Beck report(Beck et al., 2009) essentially confirmed earlier findings by Hess et al. (Hess et al., 2002) of a pool of HIV viral RNA associated with RBC in HIV-1-infected patients. These included patients in whom plasma viraemia had been suppressed to undetectable levels for several months with anti-retroviral drugs. Their investigations revealed higher numbers of RNA copies per mL of whole blood in the RBC than white cell preparations of the same patients, suggesting that the bulk of the cellbound virus circulates on RBC rather than white blood cells (WBC). RBC-bound HIV was also able to infect HeLaT4 cells. The number of erythrocyte-bound HIV copies significantly correlated with disease severity(Hess et al., 2002). The sum total of these findings is that RBC are important elements for consideration in HIV infection and could potentially influence epidemiology and clinical progression.

In order to deal with this reservoir of infection, there is a need to understand the nature of the RBC-HIV binding, especially now that there is evidence linking certain blood groups to susceptibility or resistance to HIV(Lund et al., 2009, Nneli et al., 2004, Neil et al., 2003, Puri et al., 1998).

Blood Groups in Other Viral Infections

The binding of viruses by red blood cells, including specific binding to blood group antigens is not unique to HIV, although the mechanisms involved may differ. Noroviruses have been observed to selectively bind to group A, H and difucosylated Lewis blood groups (Nilsson et al., 2009, Shirato-Horikoshi et al., 2007), and only volunteers who were secretors of these antigens were found to be susceptible to norovirus gastroenteritis(Le Pendu et al., 2006). Although this binding was not directly related to red blood cells, it nevertheless demonstrated viral affinity for a red blood cell antigen expressed on gastric mucosal cells and led to the conclusion that the group A, H and Le^b antigens act as viral receptors for the virus(Le Pendu et al., 2005).

Anaemia

Anaemia is the most important prognostic indicator in AIDS patients (Mocroft et al., 1999, Harris et al., 2008, Olayemi et al., 2008). A large scale study involving over 4000 patients established an increasing hazard ratio of 1.42, 2.56 and 5.26 for mild, moderate and severe anaemia, respectively (Harris et al., 2008). Various mechanisms have been suggested, including bone marrow suppression by various cytokines, toxic depletion by the virus, and immune destruction following sensitization with viral proteins (Volberding et al., 2003, Weiss and Goodnough, 2005, Beck and Alving, 2011). However, the degree of anaemia has not been correlated with the amount of erythrocyte-bound HIV (E-HIV), a phenomenon that might account for the autoimmune haemolytic anaemia that commonly occurs in HIV-infected individuals. Of interest is the observation by Martins-Silva *et al* that HIV infection alters red cell and lymphocyte membrane fluidity, membrane protein activity and brings about changes in trans-membrane calcium transportation (Martins-Silva et al., 2006). These changes may disrupt erythrocyte membrane stability, thus promoting haemolysis and ultimately anaemia.

Blood group antigens have also been reported to occur on trans-membrane transport channels (Telen, 1995). These transport channels have an influence on membrane structure and integrity and HIV binding to these transport channels may explain some of the mechanisms that contribute to membrane instability with consequent haemolysis and cytopaenias in HIV patients. Binding of R5 viruses to CCR5, for example, has been shown to elicit phospholipase C production (Fantuzzi et al., 2008). Furthermore, others have reported that phospholipase C is translocated to the periphery of activated natural killer (NK) cells with possible role in cell-mediated cytotoxicity (Ramoni et al., 2001). However, the role of this enzyme in haemolytic processes in HIV patients remains a matter for further enquiry.

The haemolytic effects of phospholipase C could be further aggravated by the presence of lipolytic enzymes emanating from the various opportunistic infections. Many bacteria produce this lytic enzyme including *Mycobacterium tuberculosis*, a common opportunistic pathogen in HIV patients (Johansen et al., 1996, Côtes et al., 2008, Bakala N'Goma et al., 2010, Schué et al., 2010).

Conclusion

Finally, HIV continues to take a toll on the health and economies of the world especially the developing countries. To date a cure for this virus continues to be elusive. However, a virtually complete remission has been attained in one patient following a bone marrow transplant of a CCR5 Δ 32/ Δ 32 mutant cell line (Hütter, 2009). However, despite this milestone, donors with this mutation are still far too rare and limited to European populations. The P^k blood group has also been demonstrated to offer such protection (Lund et al., 2009). These findings indicate that there could be additional blood groups that can be explored in the fight against HIV. Very few studies have focused on the role of blood groups in preventing or enhancing HIV infection, and these few had the limited scope of only looking at the ABH and Rh(D) blood groups. Africa suffers from a lot of malignancies that require total body irradiation and bone marrow reconstitution, some of them linked to HIV infection. The occurrence of HIV-resistant blood groups may avail bone marrows that may well be an answer to HIV eradication in such cancer patients.

CONFLICT OF INTEREST

The authors have nothing to disclose and declare that there is no potential conflict of interest.

FUNDING SOURCE- This investigation was supported by a research grant from the Office of Research and Development of the University of Botswana, Cape Peninsula University of Technology and National Research Foundation (NRF), South Africa.

References

Abdulazeez, A., Alo, E. & Rebecca, S. 2008. Carriage rate of human immunodeficiency virus (HIV) infection among different ABO and Rhesus blood groups in Adamawa state, Nigeria. *Biomedical Research*, 19, 41-44.

Anstee, D. J. 2010. The relationship between blood groups and disease. *Blood*, 115, 4635-4643

Arendrup, M., Hansen, J. E., Clausen, H., Nielsen, C., Mathiesen, L. R. & Nielsen, J. O. 1991. Antibody to histo-blood group A antigen neutralizes HIV produced by lymphocytes from blood group A donors but not from blood group B or O donors. *AIDS*, 5, 441-4.

Bakala N'goma, J. C., Schué, M., Carrière, F., Geerlof, A. & Canaan, S. 2010. Evidence for the cytotoxic effects of Mycobacterium tuberculosis phospholipase C towards macrophages. *BBA-Molecular Cell Biology L* 1801, 1305-1313.

Barnwell, J. W., Nichols, M. E. & Rubinstein, P. 1989. In vitro evaluation of the role of the Duffy blood group in erythrocyte invasion by Plasmodium vivax. *J Exp Med*, 169, 1795-1802.

Beck, Z. & Alving, C. R. 2011. Interactions of infectious HIV-1 virions with erythrocytes: implications for HIV-1 pathogenesis and infectivity. *In:* CHANG, T. L. (ed.) *HIV-Host Interactions*. In-Tech.

Beck, Z., Brown, B. K., Wieczorek, L., Peachman, K. K., Matyas, G. R., et al. 2009. Human erythrocytes selectively bind and enrich infectious HIV-1 virions. *PLoS One*, 4.

Bolton, M. J. & Garry, R. F. 2011. Sequence similarity between the erythrocyte binding domain of the Plasmodium vivax Duffy binding protein and the V3 loop of HIV-1 strain MN reveals a functional heparin binding motif involved in binding to the Duffy antigen receptor for chemokines. *Virology Journal*, *8*, 523.

Bowman, J. M. 1996. Hemolytic Disease of the Newborn. Vox Sanguinis, 70, 62-67.

Cartron, J.-P. 2010. Blood groups: genetics and physiology. ISBT Science Series, 5, 27-45.

Clark, P. & Wu, O. 2011. ABO blood groups and thrombosis: a causal association, but is there value in screening? *Future Cardiology*, 7, 191-201.

Côtes, K., Bakala N'goma, J., Dhouib, R., Douchet, I., Maurin, D., Carrière, F. & Canaan, S. 2008. Lipolytic enzymes in Mycobacterium tuberculosis. *Applied Microbiology and Biotechnology*, 78, 741-749.

Daniels, G. 1999. Functional aspects of red cell antigens. Blood Rev, 13, 14-35.

Dawson, T. C., Lentsch, A. B., Wang, Z., Cowhig, J. E., Rot, A., Maeda, N. et al. 2000. Exaggerated response to endotoxin in mice lacking the Duffy antigen/receptor for chemokines (DARC). *Blood*, 96, 1681-1684.

Engelmann, B., Schumacher, U. & Haen, E. 1992. Epidermal growth factor binding sites on human erythrocytes in donors with different ABO blood groups. *American Journal of Hematology*, 39, 239-241.

Fantuzzi, L., Spadaro, F., Purificato, C., Cecchetti, S., Podo, F., Belardelli, F., et al. 2008. Phosphatidylcholine-specific phospholipase C activation is required for CCR5-dependent, NF-kB-driven CCL2 secretion elicited in response to HIV-1 gp120 in human primary macrophages. *Blood*, 111, 3355-63.

Gardner, L., Patterson, A. M., Ashton, B. A., Stone, M. A. & Middleton, J. 2004. The human Duffy antigen binds selected inflammatory but not homeostatic chemokines. *Biochemical and Biophysical Research Communications*, 321, 306-312.

Greenwell, P. 1997. Blood group antigens; molecules seeking a function? *Glycoconjugate Journal*, 14, 159-173.

Hadley, T. J. & Peiper, A. C. 1997. From Malaria to chemokine receptor; the emerging physiologic role of the Duffy blood group antigen. *Blood Reviews*, 89, 3077-91.

Hammache, D., Yahi, N., Maresca, M., Pie'Roni, G. & Fantini, J. 1999. Human erythrocyte glycosphingolipids as alternative cofactors for human immunodeficiency virus type 1 (HIV-1) entry: evidence for CD4-induced interactions between HIV-1 gp120 and reconstituted membrane microdomains of glycosphingolipids (Gb3 and GM3)

Journal of Virology, 73, 5244–5248.

Harris, R. J., Sterne, J. A., Abgrall, S., Dabis, F., Reiss, P., Saag, M., et al. 2008. Prognostic importance of anaemia in HIV type-1-infected patients starting antiretroviral therapy: collaborative analysis of prospective cohort studies. *Antiviral Therapy*, 13, 959-67.

He, W., Neil, S., Kulkarni, H., Wright, E. & Agan, B. K., Et Al 2008. Duffy antigen receptor for chemokines mediates trans-infection of HIV-1 from red blood cells to target cells and affects HIV-AIDS susceptibility. *Cell Host Microbe*, 4, 52–62.

HE, W., NEIL, S., KULKARNI, H., WRIGHT ,E., AGAN, B.K., ET AL 2008. Duffy antigen receptor for chemokines mediates trans-infection of HIV-1 from red blood cells to target cells and affects HIV-AIDS susceptibility. *Cell Host Microbe*, 52–62.

Hess, C., Klimkait, T., Schlapbach, L., Zenero, V. D., Sadallah, S. & Horakova, E. 2002. Association of a pool of HIV-1 with erythrocytes in vivo: a cohort study. *Lancet*, 359, 2230-4.

Horakova, E., Gasser, O., Sadallah, S., Inal, J., Bourgeois, G., Ziekau, I., et al. 2004. Complement mediates the binding of HIV to erythrocytes. *J Immunol*, 173, 4236-41.

Horne, K., Li, X., Jacobson, L., Palella, F., Jamieson, B., Margolick, J., et al. 2009. Duffy antigen polymorphisms do not alter progression of HIV in African Americans in the MACS cohort. *Cell Host Microbe*, *5*, 415.

Hütter, G., Nowak, D., Mossner, M., Et Al. 2009. Long-Term Control of HIV by CCR5 Delta32/Delta32 Stem-Cell Transplantation. *New England Journal of Medicine*, 360, 692-8.

Jenkins, P. V., O'donnelL, J.S. 2006. ABO blood group determines plasma von Willebrand factor levels: a biologic function after all? *Transfusion*, 46, 1836-44.

Johansen, K. A., Gill, R. E. & Vasil, M. L. 1996. Biochemical and molecular analysis of phospholipase C and phospholipase D activity in mycobacteria. *Infection and Immunity*, 64, 3259-3266.

John, O. D., Terry, O. A., Adeolu, O. A. & Favour, O. 2011. A Survey of prevalence of serum antibodies to human immunodeficiency deficiency virus (HIV), hepatitis B virus (HBV) and hepatitis C virus (HCV) among blood donors. *N Am J Med Sci*, *3*, 35-38.

Lachgar, A., Jaureguiberry, G., Le Buenac, H., Bizzini, B., Zagury, J. F., Rappaport, J., et al. 1998. Binding of HIV-1 to RBCs involves the Duffy antigen receptors for chemokines (DARC). *Biomedicine and Pharmacotherapy*, 52, 436-439.

Le Pendu, J., Ruvoën-Clouet, N., Kindberg, E. & Svensson, L. 2006. Mendelian resistance to human norovirus infections. *Semininars in Immunology*, 18, 375-386. Lourenço, D., Miranda, F. & Lopes, L. H. C. 1996. ABO Blood Groups as Risk Factors for Thrombosis. *Clinical Applications in Thrombosis and Hemostasis*, 2, 196-199.

Lund, N., Olsson, M. L., Ramkumar, S., Sakac, D., Yahalom, V., Levene, C., et al. The human Pk histo-blood group antigen provides protection against HIV-1 infection. *Blood Reviews*, 113, 4980-4991.

Martins-Silva, J., Santos, N. C., Doroana, M., Duarte, N., Tavares, L., Antunes, F. et al. 2006. Changes in blood cell membrane properties in HIV type-1-Infected patients. *AIDS Research Human Retrovirology*, 22, 849–853.

Mocroft, A., Kirk, O., Barton, S. E., Dietrich, M., Proenca, R., Colebunders, R., et al. 1999. Anaemia is an independent predictive marker for clinical prognosis in HIV-infected patients from across Europe. EuroSIDA study group. *AIDS*, 138, 943-50.

Mudad, R. & Telen, M. J. 1996. Biologic functions of blood group antigens. *Current Opinion in Hematology*, 3, 473-9.

Neil, S. J., Magre, S., Mcknight, A. & Weiss, R. A. 2003. A-B-O blood groups and HIV-1 infection. *Antiviral Therapy*, Abstract no. 134.

Nibbs, R., Graham, G. & Rot, A. 2003. Chemokines on the move: control by the chemokine "interceptors" Duffy blood group antigen and D6. *Seminars in Immunology*, 15, 287-294.

Nilsson, J., Rydell, G. E., Le Pendu, J. & Larson, G. 2009. Norwalk virus-like particles bind specifically to A, H and difucosylated Lewis but not to B histo-blood group active glycosphingolipids. *Glycoconjugate Journal*, 26, 1171–1180.

Nneli, R. O., Ekpo, B. O., Egene, J. & Ohaeri, O. C. 2004. Prevalence of Rh and ABO blood groups in HIV seropositive pregnant women in Enugu, Nigeria. *Nigerian Journal of Physiological Science*, 19, 7-9.

Olayemi, E., Awodu, A. O. & Bazuaye, G. N. 2008. Autoimmune hemolytic anemia in HIV-infected patients: a hospital based study. *Annals of African Medicine*, 72-6.

Puri, A., Hug, P., Mun^oz-Barroso, I. & Blumenthal, R. 1998. Human erythrocyte glycolipids promote HIV-1 envelope glycoprotein-mediated fusion of CD4+ cells. *Biochemical and Biophysical Research Communications*, 242, 219–225.

Rachkewich, R. A., Crookston, M. C., Tilley, C. A. & Wherrett, J. R. 1978. Evidence that blood group A antigen on lymphocytes is derived from the plasma. *International Journal of Immunogenetics*, 5, 25-29.

Ramoni, C., Spadaro, F., Menegon, M. & Podo, F. 2001. Cellular localization and functional role of phosphatidylcholine-specific phospholipase C in NK Cells. *Journal of Immunology*, 167, 2642-2650.

Ramsuran, V., Kulkarni, H., He, W., Mlisana, K., Wright, E. J., Werner, L., et al. 2011. Duffy-Null–Associated Low Neutrophil Counts Influence HIV-1 Susceptibility in High-Risk South African Black Women. *Clinical Infectious Disease*, 52, 1248–1256.

Ravn, V., Mandelt, U., Svenstrup, B. & Dabelsteen, E. 1994. N-Acetyllactosamine and sialosyl N-acetyllactosamine in normal and malignant human endometrium. *Glycosylation & Disease*, 1, 271-278.

Ravn, V., Teglbjaerg, C.S., Mandel, U., Dabelsteen, E. 1992. The distribution of type-2 chain histo-blood group antigens in normal cycling human endometrium. *Cell Tissue Research*, 270, 425-433.

Reich, D., Nalls, M. A., Kao, W. H. L., Akylbekova, E. L., Tandon, A., et al. 2009. Reduced neutrophil count in people of African descent is due to a regulatory variant in the Duffy antigen receptor for chemokines gene. *PLoS Genetics*, 5, e1000360. doi:10.1371/journal.pgen.1000360.

Sayal, S. K., Das, A. L. & Nema, S. K. 1996. Study of blood groups in HIV seropositive patients. *Indian Journal of Dermatology, Venereology and Leprology*, 62, 295-7.

Schleef, M., Strobel, E., Dick, A., Frank, J., Schramm, W. & Spannagl, M. 2005. Relationship between ABO and Secretor genotype with plasma levels of factor VIII and von Willebrand factor in thrombosis patients and control individuals. *British Journal of Haematology*, 128, 100-107.

Schué, M., Maurin, D., Dhouib, R., N'goma, J. C. B., Delorme, V., Lambeau, G., et al. 2010. Two cutinase-like proteins secreted by Mycobacterium tuberculosis show very different lipolytic activities reflecting their physiological function. *FASEB Journal*, 24, 1893-1903.

Shirato-Horikoshi, H., Ogawa, S., Wakita, T., Takeda, N. & Hansman, G. S. 2007. Binding activity of norovirus and sapovirus to histo-blood group antigens. *Archives in Virology*, 152, 457–461.

Tan, M. & Jiang, X. 2005. Norovirus and its histo-blood group antigen receptors: an answer to a historical puzzle. *Trends in Microbiology*, 13, 285-293.

Telen, M. J. 1995. Erythrocyte blood group antigens: not so simple after all. *Blood*, 85, 299-306.

Telen, M. J., Roberts, K. B. & Bartlett, J. A. 1990. HIV-associated autoimmune hemolytic anemia: report of a case and review of the literature. *Journal of Acquired Immune Deficiency Syndrome*, **3**, 933-7.

Ukaejiofo, E. O. & Nubila, T. 2006. Association between ABO, Rhesus blood group systems and hemoglobin genotype among confirmed HIV/AIDS TB-coinfected patients in Enugu Urban, Nigeria. *West African Journal of Medicine*, 25, 61-64.

Volberding, P. A., Baker, K. R. & Levine, A. M. 2003. Human Immunodeficiency virus hematology. *Hematology*, 294-313.

Wazirali, H., Ashfaque, R.A., Herzig, J.W. 2005. Association of Blood Group A with Increased Risk of Coronary heart Disease in the Pakistani Population. *Pakistani Journal of Physiology*, 1.

Weiss, G. & Goodnough, L. T. 2005. Anemia of Chronic Disease. N Engl J Med, 352, 1011-1023.

Winkler, C. A., An, P., Johnson, R., Nelson, G. W. & Kirk, G. 2009. Expression of Duffy antigen receptor for chemokines (DARC) has no effect on HIV-1 acquisition or progression to AIDS in African Americans. *Cell Host Microbe*, *5*, 411-413.

Winkler, C. A., An, P., Johnson, R., Nelson, G.W., KirK, G. 2009. Expression of Duffy Antigen Receptor for Chemokines (DARC) Has No Effect on HIV-1 Acquisition or Progression to AIDS in African Americans. *Cell Host & Microbe*, *5*, 411-413.

CHAPTER THREE

BLOOD GROUP ANTIGENS C, Lu^b AND P1 MAY HAVE A ROLE IN HIV INFECTION IN AFRICANS

^{1,2}Modisa S Motswaledi, ¹Ishmael Kasvosve, ²Oluwafemi O Oguntibeju

¹Department of Medical Laboratory Sciences, Faculty of Health Sciences, University of Botswana, Gaborone, Botswana.

²Nutrition & Chronic Diseases Research Group, Oxidative Stress Research Centre, Department of Biomedical Sciences, Faculty of Health & Wellness Sciences, Cape Peninsula University of Technology, Cape Town South Africa

Address correspondence to: Department of Biomedical Sciences, Faculty of Health & Wellness Sciences, Cape Peninsula University of Technology, Bellville, 7535, Cape Town, South Africa.

Email: <u>oguntibejuo@cput.ac.za</u>, <u>bejufemi@yahoo.co.uk</u> Tel: +27219538495, Fax: +27219538495

Modisa S Motswaledi, MT (ASCP), MSc, BSc, (modisa.motswaledi@mopipi.ub.bw)

Ishmael Kasvosve, PhD, MSc, BSc (ishmael.kasvosve@mopipi.ub.bw)

Oluwafemi O Oguntibeju, DTech, FIMLS, FACBS, FIBMS, CSci, Pr.Sci.Nat (oguntibejuo@cput.ac.za, bejufemi@yahoo.co.uk)

Paper published in PLoS ONE, 2016, 11(2): e0149883. doi:10.1371/journal.pone.0149883

Abstract

BACKGROUND: Botswana is among the world's countries with the highest rates of HIV infection. It is not known whether or not this susceptibility to infection is due to genetic factors in the population. Accumulating evidence, however, points to the role of erythrocytes as potential mediators of infection. We therefore sought to establish the role, if any, of some erythrocyte antigens in HIV infection in a cross-section of the population.

METHODS: 348 HIV-negative samples were obtained from the National Blood Transfusion Service as residual samples, while 196 HIV-positive samples were obtained from the Botswana-Harvard HIV Reference Laboratory. Samples were grouped for twenty three antigens. Chi-square or Fischer Exact analyses were used to compare the frequencies of the antigens in the two groups. A stepwise, binary logistic regression was used to study the interaction of the various antigens in the light of HIV-status.

RESULTS: The Rh antigens C and E were associated with HIV-negative status, while blood group Jk^a, P₁ and Lu^b were associated with HIV-positive status. A stepwise binary logistic regression analysis yielded group C as the most significant protective blood group while Lu^b and P₁ were associated with significantly higher odds ratio in favour of HIV-infection. The lower-risk-associated group C was significantly lower in Africans compared to published data for Caucasians and might explain the difference in susceptibility to HIV-1.

CONCLUSION: The most influential antigen C, which also appears to be protective, is significantly lower in Africans than Caucasians. On the other hand, there are multiple antigens associated with increased risk that may override the protective role of C. A study of the distribution of these antigens in other populations may shed light on their roles in the HIV pandemic.

Keywords:

Blood group, HIV risk, Blood group C, Blood Group P_1 , Blood group Lu^b, Blood group frequencies

Conflict of Interest: The authors declare no potential conflict of interest.

Funding: This study was supported by a research grant from the Office of Research, University of Botswana and Cape Peninsula University of Technology, granted to M Motswaledi and Prof OO Oguntibeju respectively.
Background

A number of investigators have alluded to the role of blood groups in HIV epidemiology. Arendrup and colleagues, for example, reported that HIV from lymphocytes of blood group A individuals was neutralized by anti-A, implying that this mechanism could potentially reduce the likelihood of infection in ABO discordant couples (Arendrup et al., 1991). These findings were later corroborated by Neil and colleagues (Neil et al., 2003). Furthermore, other investigators have reported higher prevalence of HIV-2 in blood group AB (Abdulazeez et al., 2008), in blood group O (Sayal et al., 1996) as well as a protective function associated with blood group P^k (Lund et al., 2009, Lund et al., 2005).

Many studies have demonstrated the promiscuity of red blood cell antigens in binding to bacterial toxins, viruses and other molecules (Branch, 2010, Nilsson et al., 2009, Horakova et al., 2004, Lachgar et al., 1998). The Duffy antigen receptor for chemokines (DARC) is the receptor for *Plasmodium vivax (Barnwell et al., 1989, Hadley and Peiper, 1997)* and has stimulated debate on its role in HIV-infection as a candidate for HIV-binding and infection of susceptible cells (Lachgar et al., 1998, Horne et al., 2009, He et al., 2008, Ramsuran et al., 2011, Bolton and Garry, 2011). Other reports have suggested that certain erythrocyte lipids promote viral membrane fusion with CD4 cells and thus facilitate infection (Puri et al., 1998, Puri et al., 2004, Hammache et al., 1999). The predilection of HIV for any particular blood group could therefore explain the epidemiology among populations expressing that antigen to varying degrees.

Botswana is one of the countries in the world most hard-hit by the HIV pandemic with estimated national prevalence of 17.6%(Agency, 2008) and as high as 33.7% among pregnant women aged 15-49 years(Gomez et al., 2007). It is not known whether the susceptibility of this population was purely attributable to its social habits or whether there exists a genetic predisposition to HIV infection that exacerbated its vulnerability. If indeed erythrocyte surface molecules bind and increase the transfection of CD4+ cells, we hypothesize that this should reflect, in the prevalence of HIV associated with those erythrocyte surface molecules. Blood groups are probably the most abundant molecules on the red blood cell surface and are therefore logical candidates for this investigation. We thus sought to find out if the presence of any particular blood group antigen could be linked to increased HIV infection risk in Botswana.

Methods

Study Population

Three hundred and forty eight (348) HIV-negative blood samples were obtained from the National Blood Transfusion Service (NBTS) as residual samples. The National Blood Transfusion Service is the main center for donor recruitment as well as screening for suitability of blood donors and for ensuring the safety of donated blood and its components. The NBTS blood collections cover the southern part of the country and therefore represents a wide range of communities in the Southern part of Botswana. Moreover, Gaborone is a cosmopolitan city comprising of individuals from all over the country. Blood donors who tested positive for HIV were added to the HIV-positive pool. One hundred and ninety six (196) HIV-positive samples were obtained from the Botswana-Harvard HIV Reference Laboratory (BHHRL). This

laboratory is a referral laboratory in the management of HIV/AIDS patients and functions mainly to support the anti-retroviral treatment programme by providing CD4 counts and viral load testing services for prospective and ARV-enrolled patients. All samples were obtained as aliquots and assigned new study identification numbers. HIV-positive samples included both new cases and those from patients already on anti-retroviral therapy. The total number of samples tested for each blood group was determined by the availability of the specific antiserum. The HIV-positive samples from the HIV Reference Laboratory covered a wide range of communities in the southern eastern part of the country. As stated above, the city of Gaborone comprises of individuals from various parts of the country and therefore representative of the Botswana population.

Ethical clearance was obtained from the Ethics Committee of the University of Botswana's Office of Research and Development, Ministry of Health's Research Unit, BHHRL, Princess Marina Hospital Research Committee, and from the Research Ethics Committee of the Faculty of Health and Wellness Sciences, Cape Peninsula University of Technology. The study was thus approved by the ethics committees of Cape Peninsula University of Technology and Botswana's Ministry of Health. The collection of data was achieved between June 2013 and January 2015. Informed consent was not necessary since the research utilized residual samples that were rendered anonymous in accordance with the requirements of ISO 15189 (Standardisation, 2007).

Determination of Blood Groups

Patient and donor red cells were phenotyped using specific antisera (Fortress Diagnostics, Antrim, United Kingdom). Blood grouping was done following routine tube-grouping technique as outlined in the manufacturer's instructions for all blood groups, with the exception of anti-Jk^a and Jk^b which failed quality control unless subjected to the indirect antihuman globulin procedure (AHG). It was reasonable to subject them to AHG since they are also known to be non-agglutinating antibodies. The AHG procedure was used to determine the reactivity of the following antibodies: anti-Fy^a, anti-Fy^b anti-Jk^a, anti-Jk^b, anti-Kp^a, anti-Lu^a, anti-Lu^b, Anti-S and anti-s. The AHG was also used to confirm negative results for anti-D. The specificity and reactivity of the antisera was confirmed by use of an antibody identification panel, DiaPanel[®], Lot 16211.66.x-16311.66.x (BioRad[®], Japan). The blood antigens tested for in this study included: A, B, "O", D, C, c, E, e, Fy^a, Fy^b, Jk^a, Jk^b, Kp^a, Kp^b, Le^a, Le^b, Lu^a, Lu^b, M, N, P₁, S, and s.

Statistical Analysis

Data analysis was conducted using IBM SPSS version 23 software. Frequencies were used to report blood group prevalence. Comparison of proportions of blood group antigens between HIV-infected and non-infected individuals were conducted using Pearson's chi-square test (χ 2) or Fisher's exact test where appropriate. We explored the combined effect of various blood groups on HIV prevalence using a stepwise binary logistic regression analysis. Only blood group antigens with statistically significant association with HIV status in bivariate analyses

were included in the model and the results were reported as odds ratio with 95% confidence interval. Results were considered significant at p < 0.05.

Results

In total, 348 HIV-negative blood samples and 196 blood samples from HIV-infected patients were studied. The results comparing the prevalence of each antigen in the HIV group versus uninfected controls are summarized in **Table 3.1**. The frequency of C Rh antigen was lower in HIV-infected patients than in HIV-negative subjects, 17.1% (95% CI 10.7 – 23.5) versus 29.7% (95% CI 24.1 – 35.1), respectively p = 0.006. Similarly, the E Rh antigen was lower in HIV-infected patients than in HIV-negative subjects, 12.1% (95% CI 6.6 – 17.6) versus 21.2% (95% CI 16.2 – 26.2), respectively p = 0.024. Only these Rh-antigens raised the prospects of a protective function against HIV-1 infection.

In contrast, the HIV-infected group exhibited a significantly higher prevalence of blood groups Jk^a, Lu^b and P₁ compared to HIV-negative individuals, suggesting a potential infection risk associated with carriage of these antigens. Among the HIV-infected patients the prevalence of Jk^a was 85.3% ((95% CI 78.7 – 91.9) compared to 51.4% (95% CI 41.8 – 61.0) in non-infected individuals, p = 0.024. The prevalence of Lu^b was 86.7% (95% CI 80.3 – 93.1) among HIV patients versus 67.8% (95% CI 61.8 – 73.8) in non-HIV infected subjects, p = 0.0001. Among the HIV-infected patients, P₁ prevalence was 89.2% (95% CI 83.1 – 95.3) compared to 74.6% (95% CI 69.1 – 80.1), p = 0.003. All the other 19 red cell antigens we tested were not significantly different when stratified according to HIV status.

Antigen	HIV-p	-positive Subjects HIV-		V-negative Subjects			
		Antigen Frequency		Antigen Frequency			
	Ν	% (95% CI)	Ν	% (95% CI)	P-value		
A	155	34.1 (26.5 – 41.7)	239	42.3 (35.9 - 48.7)	0.109		
В	155	32.3 (24.8 - 39.8)	239	25.9 (20.2 - 31.6)	0.174		
0	239	41.3 (47.3 – 65.9)	336	41.0 (35.6 - 46.4)	1.000		
D	155	95.5 (92.2 - 98.8)	239	97.1 (94.9 - 99.3)	0.406		
С	140	17.1 (10.7 – 23.5)	269	29.7 (24.1 - 35.3)	0.006		
c	140	99.3 (97.9 - 100.7)	269	97.4 (95.5 - 99.3)	0.273		
Е	140	12.1 (6.6 – 17.6)	269	21.2 (16.2 - 26.2)	0.024		
e	140	99.3 (97.9 - 100.7)	269	98.5 (97.0 - 100)	0.665		
Fy ^a	146	10.3 (5.3 – 15.3)	251	9.6 (5.9 - 13.3)	0.818		
Fy^b	146	12.3 (6.9 – 17.7)	254	9.8 (6.1 – 13.5)	0.440		
Jk ^a	116	85.3 (78.7 - 91.9)	109	51.4 (41.8 - 61.0)	< 0.0001		
Jk ^b	117	25.6 (17.5 - 33.7)	109	31.2 (22.3 - 40.1)	0.355		
Kp ^a	123	2.5 (-0.3 – 5.3)	264	1.1 (-0.2 – 2.4)	0.387		
Kp ^b	97	45.4 (35.3 - 55.5)	266	36.5 (30.6 - 42.4)	0.124		
Le ^a	82	30.5 (20.3 - 40.7)	257	25.3 (19.9 - 30.7)	0.354		
Le ^b	66	50.0 (37.7 - 62.3)	257	42.0 (35.8 - 48.2)	0.244		
Lu ^a	118	1.7 (-0.7 – 4.1)	266	3.8 (1.5 – 6.1)	0.358		
Lu ^b	113	86.7 (80.3 - 93.1)	240	67.8 (61.8 - 73.8)	0.0001		
М	92	77.2 (68.5 - 85.9)	236	80.9 (75.8 - 86.0)	0.456		
Ν	93	71.0 (61.6 - 80.4)	269	66.9 (61.2 - 72.6)	0.470		
P1	105	89.2 (83.1 - 95.3)	248	74.6 (69.1 - 80.1)	0.003		
S	125	33.6 (25.2 - 42.0)	225	31.6 (25.4 - 37.8)	0.695		
S	111	95.6 (91.7 - 99.5)	226	89.4 (85.3 - 93.5)	0.060		

Table 3.1	– Prevalen	ce of blood	group	antigens	in	HIV-infected	and	uninfec	ted	subjects
			0							

We also investigated the possible combined influence of the protective and risk-associated blood group antigens on HIV status. For this purpose, data was subjected to a stepwise logistic regression model in which analysis was repeated after the systematic removal of the least significant among the covariates. In this model, only those blood groups that were significantly associated with HIV infection in bivariate analyses were tested. The results of this analysis are shown in **Table 3.2**.

Antigen	Odds Ratio (95% Confidence Interval)	P-value
Е	0.859 (0.169 - 0.437)	0.855
\mathbf{P}_1	0.428 (0.108 - 0.169)	0.226
С	0.146 (0.047 - 0.455)	0.001
Jk ^a	1.001 (0.156 - 6.408)	1.0
Lu ^b	1.731 (0.314 – 9.541	0.529

Table 3. 2 – Logistic regression model showing the effect of blood group antigens on the risk of HIV infection

After the stepwise elimination process, only the C, Lu^b and P1 maintained statistical significance as risk factors in the model. The results are shown in **Table 3.3**. Carriage of P₁ antigen increased the odds of HIV infection by 2.2 (95% CI 1.1 – 4.5), p = 0.026 and similarly, carriage of Lu^b increased the odds ratio for HIV infection by 2.9 (95% CI 1.5 – 5.5), p = 0.001. Conversely, presence of C Rh antigen resulted in 40% decrease in HIV risk of infection.

Table 3. 3 – Blood group antigens associated with HIV infection in multivariate logistic regression analysis

Antigen	Odds Ratio (95% Confidence Interval))	p-value
С	0.6 (0.3 – 1.0)	0.047
P ₁	2.2 (1.1 – 4.5)	0.026
Lu ^b	2.9 (1.5 – 5.5)	0.001

We further sought to determine if the frequency of protective and risk-associated antigens was any different from other populations of African ethnicity or Caucasian counterparts. The frequency of blood group C in this study, though not statistically different from that published for Blacks (p=0.8389), was much lower than that for Caucasians (p<0.0001)(Makroo et al., 2013) The frequencies for the risk-associated P₁ and Lu^b were also lower in this study than in Caucasians (p=0.0446 and p<0.0001, respectively (Hellberg, 2007, Badjie and Stubbs, 2012,

Hellberg et al., 2013). However, these other studies did not report on the HIV status of the populations from which the data was derived.

Discussion

Blood groups have been implicated as factors in the epidemiology of HIV(Arendrup et al., 1991, Bolton and Garry, 2011, Lachgar et al., 1998, Onsten et al., 2013, Puri et al., 1998, Ramsuran et al., 2011). We studied the distribution of blood groups in Botswana and identified those associated with HIV infection. We therefore determined the frequencies for ABO, Rh, Kidd, Kell, Duffy, Lewis, Lutheran, MNSs, and P₁ blood group systems in both HIV-infected patients and non-HIV-infected individuals. The frequencies of Fy^b, Jk^a, Jk^b, Kp^b, Le^b, N, P₁, Lu^b and s in Botswana were statistically lower than the published literature for Blacks (Badjie and Stubbs, 2012, Makroo et al., 2013, Cooling, 2011). This is not a surprising finding since blood group variations do occur within a population (Badjie and Stubbs, 2012) and may sometimes be aggravated by cultural marriage practices that promote peculiar phenotypes (Mavanga et al., 2013). Recent publications have also demonstrated the peculiarities of the Botswana population that necessitate the use of Botswana-specific reference values (Mine et al., 2011, Maphephu and Kasvosve, 2011, Segolodi et al., 2014). Other than these, the rest of the antigen frequencies were consistent with published literature as noted.

The C and E antigens in the Rhesus blood group system were found to be rarer in HIV-infected subjects while that of Jk^a , Lu^b and P_1 , were more prevalent in this group. In comparison to individuals of African ethnicity, the Botswana population does not appear to have any higher frequency of risk-associated blood groups nor does it have any lower frequency of those associated risk mitigation. However, the prevalence of the risk-mitigating antigen C was lower in blacks (and in the Botswana population studied) than in Caucasians or Asians. Given the strength of the influence of this antigen in the model used, and given the relative rarity of its frequency in the Botswana population, we conclude that the C antigen likely plays an important role in mitigating against HIV-1 infection and could have played a role in the HIV pandemic in Botswana.

We hypothesized that a higher prevalence of an antigen in the HIV+ group than the control group possibly indicates that its presence promotes HIV infection, and vice versa. We therefore investigated the potential mitigation of the "protective" phenotypes against the "risky" ones. The odds ratios were therefore computed for risk-associated blood groups in the presence or absence of Rh C or E expression. The presence of C or E did not consistently result in obliteration of the risk-significance in the risk-associated phenotypes. This analysis was also hampered by the simultaneous carriage or alternate carriage of suspect antigens in different individuals, making it difficult to isolate the effect of each antigen.

To further elucidate how the odds ratio is impacted by interaction of all significant blood groups, the data was subjected to a logistic regression analysis. The Jk^a and E antigens, although highly significant in their association to HIV infection in the isolated Chi-square analysis, failed to maintain significance in the logistic regression model. On the other hand, the C, P₁ and Lu^b remained significant with C significantly reducing the odds of HIV infection while P₁ and Lu^b increased the odds of infection.

Our results for blood group O are not in concordance with the reports by other investigators (Nneli et al., 2004, Sayal et al., 1996) who documented an increased HIV susceptibility for this group. Neither did we find any increased risk in group B individuals as reported in Brazil (Onsten et al., 2013). There was also no difference in HIV prevalence between D-positive and D-negative subjects as reported in Nigeria (Abdulazeez et al., 2008). However, we take cognizance of the possible population differences and the effect of other factors such as viral strains in those populations that may account for the disparities observed. But for the Botswana population, the ABO groups do not appear to have any role in enhancing or diminishing HIV risk of infection.

Our findings indicate that P_1 antigen, a glycosphingolipid, is associated with a higher risk of HIV infection and thus corroborates the role of glycosphingolipids as facilitators of viral fusion to CD4+ cells (Hammache et al., 1999, Puri et al., 2004). In their experiments, Puri and colleagues demonstrated that HIV infection of CD4+ cells could be averted by addition of a glycosphingolipid inhibitor. Conversely, the presence of the glycosphingolipid increased susceptibility of CD4+ cells to HIV infection. The P_1 antigen is also known to bind several microorganisms including enteropathogenic *Escherichia coli* and parvovirus (Branch, 2010) and has a wide tissue distribution (Hellberg et al., 2013), providing multiple opportunities for viral binding. It is conceivable from its broad spectrum of binding that P_1 could also bind and act as a receptor for HIV. We conclude that our findings corroborate the role of P_1 as a potential aggravator of HIV infection.

We did not find any risks associated with the Duffy blood group system or its Duffy-null phenotype alluded to elsewhere (Thobakgale and Ndung'u, 2014, Ramsuran et al., 2011, He et al., 2008, Lachgar et al., 1998). Our results in this respect appear to support an insignificant role of Duffy blood group expression consistent with other investigators(Winkler et al., 2009, Horne et al., 2009). However, we did confirm that the Duffy-null phenotype was associated with leukopenia(Reich et al., 2009), though not neutropenia, in Botswana.

The Lutheran blood group is a member of the immunoglobulin superfamily and has been shown to function as a receptor for laminin (Parsons et al., 1995, Nemer et al.) with potential role for signal transduction and is widely distributed in various tissues (Parsons et al., 1995, Eyler and Telen, 2006) No data is available on the role of the Lutheran blood groups in HIV infection. However, its tissue-wide distribution would facilitate infection at various anatomic sites, if such a role could be confirmed.

Conclusion

We have studied the role of 23 blood group antigens in relation to susceptibility to HIV infection, this being hitherto the most comprehensive study on the role of blood groups in HIV infection. The strongest association in this study was with the Rh antigen C, which also appears to be protective, and is significantly lower in Africans than Caucasians. On the other hand, there are multiple antigens associated with increased risk that may override the protective role of C. A study of the distribution of these antigens in other populations may shed light on their roles in the HIV pandemic.

Acknowledgement:

The authors wish to express gratitude to the management and staff of the National Blood Transfusion Service, Princess Marina Hospital, Botswana Harvard HIV Reference Laboratory for their kind support, without which this study would not have been successful. Gratitude is also expressed to the University of Botswana Office of Research and Development for providing the funding for this project and to Professor Keoagile Thaga, Department of Statistics at the University of Botswana for his guidance with statistical analysis. Financial support granted to Prof OO Oguntibeju by Cape Peninsula University of Technology and National Research Foundation (NRF) of South Africa is also acknowledged.

References:

Abdulazeez, A., Alo, E. & Rebecca, S. 2008. Carriage rate of human immunodeficiency virus (HIV) infection among different ABO and Rhesus blood groups in Adamawa state, Nigeria. *Biomedical Research*, 19, 41-44.

Botswana Government . 2008. 2008 Botswana Aids Impact Survey III Statistical Report. Gaborone: Botswana.

Arendrup, M., Hansen, J. E., Clausen, H., Nielsen, C., Mathiesen, L. R. & Nielsen, J. O. 1991. Antibody to histo-blood group A antigen neutralizes HIV produced by lymphocytes from blood group A donors but not from blood group B or O donors. *AIDS*, 5, 441-4.

Badjie, K. & Stubbs, J. 2012. Molecular Analysis for Determining Blood Group Phenotypes. Available: <u>http://www.mayomedicallaboratories.com/articles/communique/2012/03.html</u> [Accessed 22 July 2015].

Barnwell, J. W., Nichols , M. E. & Rubinstein, P. 1989. In vitro evaluation of the role of the Duffy blood group in erythrocyte invasion by Plasmodium vivax. *Journal of Experimental Medicine*, 169, 1795-1802.

Bolton, M. J. & Garry, R. F. 2011. Sequence similarity between the erythrocyte binding domain of the Plasmodium vivax Duffy binding protein and the V3 loop of HIV-1 strain MN reveals a functional heparin binding motif involved in binding to the Duffy antigen receptor for chemokines. *Virology Journal*, 8, 523.

Branch, D. R. 2010. Blood groups and susceptibility to virus infection: new developments. *Current Opinion in Hematology*, 17, 558–564.

Cooling, L. 2011. ABO, H, Lewis Blood Groups and structurally related antigens. *In:* Roback, J., Grossman, B., Harris, T. & Hillyer, C. (eds.) *Technical Manual.* 17 ed. Bethesda, MD: American Association of Blood Banks.

Eyler, C. & Telen, M. 2006. The Lutheran glycoprotein: a multifunctional adhesion receptor. *Transfusion*, 46, 668-677.

GOMEZ, H., ANDERSON, G., BODIKA, S., KALUWA, O., MOYO, S., MMELESI, M., et al. 2007. 2007 Botswana ANC Second Generation HIV/AIDS Sentinel Surveillance Technical Report. Gaborone: Botswana Government.

Hadley, T. J. & Peiper, A. C. 1997. From Malaria to chemokine receptor; the emerging physiologic role of the Duffy blood group antigen. *Blood Reviews*, 89, 3077-91.

Hammache, D., Yahi, N., Maresca, M., Pie'Roni, G. & FantinI, J. 1999. Human erythrocyte glycosphingolipids as alternative cofactors for human immunodeficiency virus type 1 (HIV-1) entry: evidence for CD4-induced interactions between HIV-1 gp120 and reconstituted membrane microdomains of glycosphingolipids (Gb3 and GM3). *J Virol*, 73, 5244–5248.

He, W., Neil, S., Kulkarni, H., Wright, E. & Agan, B. K., et al. 2008. Duffy antigen receptor for chemokines mediates trans-infection of HIV-1 from red blood cells to target cells and affects HIV-AIDS susceptibility. *Cell Host Microbe*, 4, 52–62.

Hellberg, Å. 2007. Studies on the Genetic Basis of Pk, P and P1 Blood Group Antigen Expression. PhD Doctoral Thesis, Lund.

Hellberg, A., Westman, J. S., Thuresson, B. & Olsson, M. L. 2013. P1PK: the blood group system that changed its name and expanded. *Immunohematology*, 29, 25-33.

Horakova, E., Gasser, O., Sadallah, S., Inal, J., Bourgeois, G., Ziekau, I., et al. 2004. Complement mediates the binding of HIV to erythrocytes. *Journal of Immunology*, 173, 4236-41.

Horne, K., Li, X., Jacobson, L., Palella, F., Jamieson, B., Margolick, J., Et al. 2009. Duffy antigen polymorphisms do not alter progression of HIV in African Americans in the MACS cohort. *Cell Host Microbe*, *5*, 415.

International Organization for Standardisation. Annexure C: Ethics in laboratory medicine. ISO 15189:2007:C9. Geneva: ISO, 2007.

Lachgar, A., Jaureguiberry, G., Le Buenac, H., Bizzini, B., Zagury, J. F., Rappaport, J. et al. 1998. Binding of HIV-1 to RBCs involves the Duffy antigen receptors for chemokines (DARC). *Biomedicine and Pharmacotherapy Journal*, 52, 436-439.

Lund, N., Branch, D. R., Sakac, D., Lingwood, C. A., Siatskas, C., Robinson, C. J., et al. 2005. Lack of susceptibility of cells from patients with Fabry disease to productive infection with R5 human immunodeficiency virus. *AIDS*, 19, 1543-5.

Lund, N., Olsson, M. L., Ramkumar, S., Sakac, D., Yahalom, V., Levene, C., et al. 2009. The human Pk histo-blood group antigen provides protection against HIV-1 infection. *Blood Reviews*, 113, 4980-4991.

Makroo, R. N., Bhatia, A., Gupta, R. & Phillip, J. 2013. Prevalence of Rh, Duffy, Kell, Kidd & MNSs blood group antigens in the Indian blood donor population. *Indian Journal of MedicalResearch*. 137, March 2013, pp 521-526.

Maphephu, S. & Kasvosve, I. 2011. Reference intervals for serum total cholesterol, HDL cholesterol and non-HDL cholesterol in Batswana adu. *South African Medical Journal*, 101.

Mavanga, N. M., Boemer, F., Seidel, L., Malafu, A. N., Gothot, A. & Gerard., C. 2013. Blood groups, hemoglobin phenotypes and clinical disorders of consanguineous Yansi population. *World J Hematol* 2, 109-114.

Mine, M., Moyo, S., Stevens, S., Michael, K., Novitsky, V., Makhaola, K., et al. 2011. Immunohaematological reference values for HIV-negative healthy adults in Botswana. *African Journal of Laboratory Medicine*, 1, 7 pages.

Neil, S. J., Magre, S., Mcknight, A. & Weiss, R. A. 2003. *A-B-O blood groups and HIV-1 infection* [Online]. NLM Gateway; A service of the U.S. National Institutes of Health. Available: <u>http://gateway.nlm.nih.gov/MeetingAbstracts/ma?f=102262439.html</u>.

Nemer, W. E., Gane, P., Colin, Y., Bony, V., Rahuel, C., Galactéros, F., et al. 1998. The Lutheran Blood Group Glycoproteins, the Erythroid Receptors for Laminin, Are Adhesion Molecules. *Journal of Biological Chemistry*, 273, 16686-16693.

Nilsson, J., Rydell, G. E., Le Pendu, J. & Larson, G. 2009. Norwalk virus-like particles bind specifically to A, H and difucosylated Lewis but not to B histo-blood group active glycosphingolipids. *Glycoconjugate Journal*, 26, 1171–1180.

Nneli, R. O., Ekpo, B. O., Egene, J. & Ohaeri, O. C. 2004. Prevalence of Rh and ABO blood groups in HIV seropositive pregnant women in Enugu, Nigeria. *Nigerian Journal of Physiological Science*, 19, 7-9.

Onsten, T. G. H., Callegari-Jacques, S. M. & Goldani, L. Z. 2013. The Higher Frequency of Blood Group B in a Brazilian Population with HIV Infection. *Open AIDS J*, 7, 47-50.

Parsons, S. F., Mallinson, G., Holmes, C. H., Houlihan, J. M., Simpson, K. L., Mawby, W. J., et al. 1995. The Lutheran blood group glycoprotein, another member of the immunoglobulin superfamily, is widely expressed in human tissues and is developmentally regulated in human liver. *Proceedings of the National Academy of Sciences*, 92, 5496-5500.

Puri, A., Hug, P., Mun^oz-Barroso, I. & Blumenthal, R. 1998. Human erythrocyte glycolipids promote HIV-1 envelope glycoprotein-mediated fusion of CD4+ cells. *Biochemical and Biophysical Research Communications*, 242, 219–225.

Puri, A., Rawat, S. S., Lin, H.-M. J., Finnegan, C. M., Mikovits, J., Ruscetti, F. W. et al. 2004. An inhibitor of glycosphingolipid metabolism blocks HIV-1 infection of primary T-cells. *AIDS*, 18, 849-858.

Ramsuran, V., Kulkarni, H., He, W., Mlisana, K., Wright, E. J., Werner, L., et al. 2011. Duffy-Null–Associated Low Neutrophil Counts Influence HIV-1 Susceptibility in High-Risk South African Black Women. *Clinical Infectious Diseases*, 52, 1248–1256.

Reich, D., Nalls, M. A., Kao, W. H. L., Akylbekova, E. L. & Tandon, A. 2009. Reduced neutrophil count in people of African descent is due to a regulatory variant in the Duffy antigen receptor for chemokines gene. *PLoS Genetics*, 5, e1000360. doi:10.1371/journal.pgen.1000360.

Sayal, S. K., Das, A. L. & Nema, S. K. 1996. Study of blood groups in HIV seropositive patients. *Indian J Dermatol Venereol Leprol*, 62, 295-7.

Segolodi, T. M., Henderson, F. L., Rose, C. E., Turner, K. T., Zeh, C., Fonjungo, P. N., et al. 2014. Normal Laboratory Reference Intervals among Healthy Adults Screened for a HIV Pre-Exposure Prophylaxis Clinical Trial in Botswana. *PLoS One*, *9*, e93034.

Thobakgale, C. F. & Ndung'u, T. 2014. Neutrophil counts in persons of African origin. *Curr Opinions in Hematology*, 21, 50-7.

Winkler, C. A., An, P., Johnson, R., Nelson, G. W. & Kirk, G. 2009. Expression of Duffy antigen receptor for chemokines (DARC) has no effect on HIV-1 acquisition or progression to AIDS in African Americans. *Cell Host Microbe*, *5*, 411-413.

CHAPTER FOUR POTENTIAL ROLE OF LU/BCAM IN HIV-RELATED ATHEROSCLEROSIS

Modisa Sekhamo Motswaledi^{1, 2}(corresponding author), Ishmael Kasvosve¹, Oluwafemi Omoniyi Oguntibeju²

¹ Department of Medical Laboratory Sciences, Faculty of Health Sciences, University of Botswana, modisa.motswaledi@mopipi.ub.bw, <u>ishmael.kasvosve@mopipi.ub.bw</u>, Gaborone, Botswana, ² Department of Biomedical Sciences, Faculty of Health & Wellness Sciences, Cape Peninsula University of Technology, Cape Town, South Africa. <u>OguntibejuO@cput.ac.za</u> (Address correspondence to: Department of Biomedical Sciences, Faculty of Health & Wellness Sciences, Cape Peninsula University of Technology, Bellville, 7535, Cape Town, South Africa. Tel: +27219538495, Fax: +27219538495

Paper submitted to: African Journal of Laboratory Medicine

Abstract:

Atheromatous lesions are formed by macrophage and LDL-cholesterol invading the vascular intima. Here we show that increasing cholesterol levels are associated with peripheral monocyte depletion and this imbalance is aggravated by carriage of Lu/BCAM leukocyte adhesion molecule. This is true only in HIV infection and probably explains the risk of atherosclerosis observed in HIV infection.

Key words: Cholesterol, Non-DHL cholesterol, Lu/BCAM, atherosclerosis

Introduction

The pathology of atherosclerosis is tightly linked to three main factors; leukocytes, plasma lipids and vascular injury or chronic inflammation (Frostegard et al., 1990, Oh et al., 2009b). Leukocytes such as neutrophils, activated monocytes or monocyte-derived macrophages and platelets accumulate in the vascular sub-endothelial matrix by use of adhesion molecules. The Lutheran blood group, also known as basal cell adhesion molecule (Lu/BCAM), is one such molecule and is expressed on both monocytes and endothelial cells (Pedraza et al., 2000). It is the natural ligand for laminin, an extracellular basement membrane protein that facilitates both adhesion and extravasation of monocytes (Pedraza et al., 2000). In addition, monocytes and neutrophils also constitutively express $\alpha 4\beta 1$ integrin (Chaar et al., 2010), another ligand for Lu/BCAM. These molecules enable these cells to adhere to the endothelium and have been implicated in the pathology of crescentic, sickle cell disease (Chaar et al., 2010) as well as in several cancers suggesting more efficient tumour tethering and greater tumour size (Chang et al., 2017, De Grandis et al., 2013).

In HIV-infected individuals, the viral Tat protein accumulates in the extracellular matrix of blood vessels and facilitates integrin-mediated adhesion (Chiodelli et al., 2012) with the consequence that HIV-infected monocytes have enhanced extravasation (Dhawan et al., 1992). HIV infection is also characterised by chronic inflammation resulting from the activation of Tcells, macrophages, neutrophils and NK cells- all of which release pro-inflammatory cytokines that create a state of persistent inflammation (Appay and Sauce, 2008). This chronic inflammation involves neutrophils which produce reactive oxygen species thought to contribute to the oxidation of LDL-C (Hartwig et al., 2015). Cholesterol enrichment of neutrophil membranes results in more stabilized rolling and adhesion on endothelial cells (Oh et al., 2009a), where they may recruit monocytes (Hartwig et al., 2015) to promote atherogenesis. In particular, non-HDL cholesterol has been documented as one of the more important promoters of atherosclerosis in HIV-infected patients (Badiou et al., 2008). The atheromatous lesion forms by gradual accumulation of oxidized LDL-C and infiltration by macrophages, smooth muscle cells and platelets (Østerud and Bjørklid, 2003). This may ultimately narrow the vascular lumen, causing tissue hypoxia distal to the atheroma. Moreover, the release of RANTES (CCL5) by activated platelets in the atheroma leads to further recruitment of monocytes and neutrophils (von Hundelshausen et al., 2001). A high monocyte count has thus been reported to be a long term predictor of plaque-formation (Johnsen et al., 2005), while neutrophil enzymes such as elastase and myeloperoxidase in the atheroma may destabilize it (Ionita et al., 2010), leading to its fragmentation and subsequent generation of emboli that may cause vascular occlusion at distant sites.

In our previous study, the Lutheran blood group, Lu^b, was associated with a three-fold risk for HIV infection (Motswaledi et al., 2016), and it was suspected that this could be due to its adhesiveness that promotes trans-endothelial migration and spread of infected monocytes to distant sites. The current study therefore sought to determine the effect, if any, of carriage of the Lu^b antigen as an adhesion molecule on circulating monocytes and neutrophils as well as how these parameters related to cholesterol measurements. The results indicate that HIV does affect the relationship between monocytes and cholesterol, and that this relationship is accentuated by expression of Lu^b antigen.

Materials and Methods

One hundred and one (101) blood samples comprising 59 females and 42 males received at Julia Molefe Clinic in Gaborone were enrolled on the study between December 2016 and January 2017. The patients had all tested positive for HIV and were being prepared for enrolment into the antiretroviral therapy programme. A full blood count (FBC) was done using the Sysmex ® XT1800i haematology analyser (Sysmex Corporation, Kobe, Japan). The samples were phenotyped for the Lutheran antigens (Lu^a and Lu^b) using specific antisera (Fortress Diagnostics, Antrim, United Kingdom) according to the manufacturer's instructions. The reactivity of the antisera was confirmed by use of antigen-positive and negative cells selected from an antibody identification panel (DiaPanel®, Lot 45241.88.1, Bio-Rad®, Cressier FR, Switzerland). Total and HDL-cholesterol measurements were also made using the AU480 Chemistry analyser (Brea, CA). LDL cholesterol was calculated using the Friedewald equation. We also revisited a previous dataset of 261 normal subjects as a control group to examine the same relationships in healthy, HIV-uninfected subjects(Mine et al., 2011).

Ethical clearance for the study was obtained from the Office of Research and Development (University of Botswana), Botswana's Ministry of Health and Wellness's Health Research Development Committee and Faculty of Health and Wellness' Research Ethics Committee (Cape Peninsula University of Technology)[Appendices A-C]. Informed consent was not necessary since de-identified, residual samples were used (ISO, 2007).

The results were analysed using the IBM SPSS version 24 statistical software. Statistical analyses included comparison of means according to carriage of the Lutheran antigens, as well as correlation of data to explore relationship between variables. Results were considered significant only if P<0.05.

Results:

An inverse correlation was observed between total cholesterol and absolute monocyte and neutrophil counts (See Table 4.1). In HIV-infected subjects, monocyte depletion was associated with relatively high LDL-cholesterol levels (r=-0.435, p=0.003). The correlation was stronger (r=-0.479, p=0.010) in Lu^b-positive but absent (r=-0.212, p=0.414) in those negative for the antigen. Among controls (n=261), LDL cholesterol did not correlate with monocyte counts at all (r=0.031, p=0.619). Lu^b expression was associated with lower absolute monocyte counts $(0.46\pm0.19 \text{ versus } 0.54\pm0.22 \times 10^9/L, p=0.040)$.

Test on	Mean±SD (x10 ⁹ /L)	Pearson correlation coefficient (r) with TC	P-value
Lu ^b –positive and negative subjects ((n=101)		
Absolute neutrophil count	2.7±2.0	-0.207	0.038
Absolute monocyte count	0.48 ± 0.20	-0.253	0.011
Lu ^b -positive subjects only (n=76)			
Absolute neutrophil count	2.65±2.2	-0.310	0.006
Absolute Monocyte count	0.46±0.20	-0.330	0.004
Lu ^b -negative subjects only (n=25)			
Absolute neutrophil count	2.8±1.3	0.153	0.465
Absolute Monocyte count	0.54 ± 0.22	-0.047	0.822

Table 4.1: Correlation of cholesterol with phagocyte counts in Lu^b-positive and negative subjects

To further elucidate the total cholesterol (TC) component responsible for these observations, correlations were done involving phagocyte counts versus HDL-C and non-HDL-C. Non-HDL-C was consistently associated with a lower monocyte count suggesting that the leukocyte-depleting effects of cholesterol are largely due to non-HDL-C and that depletion is enhanced in Lu^b-positive subjects. Neutrophils were less affected by non-HDL-C than monocytes (See Table 4.2).

Table 4. 2: Correlation of phagocyte counts with non-HDL cholesterol in Lu^b-positive and negative subjects

Test on	Mean±SD	Non-HDL Correlation coefficient (r)	P-value
Lu ^b –positive and negative subjects (n=59)		
Absolute neutrophil count $(x10^9/L)$	2.70±2.00	-0.226	0.136
Absolute monocyte count $(x10^9/L)$	0.48±0.20	-0.401	0.006
In Lu ^b -positive subjects (n=28)			
Absolute neutrophil count (x10 ⁹ /L)	2.65±2.20	-0.369	0.053
Absolute Monocyte count $(x10^9/L)$	0.46±0.20	-0.422	0.018
In Lu ^b -negative subjects (n=31)			
Absolute neutrophil count ($x10^9/L$)	2.85±1.30	0.143	0.584
Absolute Monocyte count (x10 ⁹ /L)	0.54±0.21	-0.259	0.315

Discussion

We studied the relationship between cholesterol and phagocytes (monocytes and neutrophils) in HIV-1-infected and uninfected individuals in an attempt to determine the influence, if any, of the Lu/BCAM on phagocyte counts.

We hypothesized that carriage of the Lu^b antigen, an adhesion molecule, would enhance margination of leukocytes leading to a peripheral deficiency of the affected leukocytes. Indeed, we observed a significantly lower absolute monocyte count in Lu^b-positive individuals compared to Lu^b-negatives.

We also observed a negative correlation between cholesterol and phagocytes (neutrophils and monocytes), which corroborates the findings by other investigators that membrane cholesterol, which is in equilibrium with plasma cholesterol (Buchwald et al., 2000), increases leukocyte adhesion and extravasation (Oh et al., 2009b, Frostegard et al., 1990). Moreover, phagocyte counts were negatively correlated with non-HDL-C and LDL-C, which lipids are known to promote adhesion and atherogenesis (Frostegard et al., 1990, Østerud and Bjørklid, 2003). The monocyte depletive effect of cholesterol was accentuated in Lu^b-positive subjects but totally absent in the Lu^b-negative subjects. We conclude that higher levels of cholesterol are associated with lower peripheral monocyte and neutrophil counts and that this depletion is accentuated in individuals expressing the Lu^b antigen. This depletion is an effect of HIV infection since it was not observed in the uninfected population, and probably involves the particular role of the HIV-Tat protein's enhancement of vascular adhesion and trans-endothelial migration of monocytes reported by other investigators (Lafrenie et al., 1996). HIV-infected monocytes are known to migrate more efficiently across vessel wall while maintaining their infectivity (Dhawan et al., 1992). We conclude that the Lu^b blood group contributes to the depletion of circulating phagocytes, possibly by enhancing their margination and extravasation (Oh et al., 2009b).

Working from the established fact that cholesterol, especially LDL-C, promotes adhesion of phagocytes (Oh et al., 2009a, Frostegard et al., 1990), with possible reduction of peripheral counts, we noted that in HIV infection, even normal cholesterol levels are associated with diminishing circulating phagocytic cells. This could be due to the chronic inflammatory status occasioned by HIV infection. It appears, therefore, that HIV-infected individuals are at higher risk of developing cholesterol-related monocytopenia and neutropenia, which, in the presence of increased non-HDL cholesterol, points to the peculiar vulnerability of HIV-infected patients to cardiovascular diseases. Although these observations were generally expected considering that cholesterol promotes tethering of phagocytes to the endothelium (Oh et al., 2009b, Dhawan et al., 1992), it was interesting that only in HIV infection was the phagocyte-depleting effect of cholesterol significant.

When cholesterol components were examined, non-HDL-C negatively affected phagocyte counts consistently, especially the monocyte counts, which finding corroborated previous reports by (Badiou et al., 2008) linking non-HDL-C to atherosclerosis in HIV infection.

Conclusion

Our findings suggest that the risk of cardiovascular disease in HIV-1-infected subjects may in part be due to an interplay between cholesterol, phagocytes and Lu^b-mediated adhesion, all of which tend to recruit phagocytes out of circulation. While a genetic predisposition for atherogenesis has always been alluded to, our results suggest the Lu^b adhesion molecule to be a potentially important genetic link. The results of this study also imply a need to manage cholesterol in HIV-infected patients to achieve lower levels than the general public, especially for those who are Lu^b-positive. Moreover, the vulnerability occasioned by the Lu^b-positive status may provide opportunity for personalized care in Lu^b-positive, HIV-infected individuals.

A limitation of our study was that we did not directly measure phagocyte migration but rather, implied phagocyte migration from diminished peripheral cell counts. Future studies need to focus on direct migration measurements of phenotyped phagocytes.

Acknowledgments

The authors wish to express gratitude to the Staff and Management of the Gaborone District Health Management Team, especially those at Julia Molefe Clinic where samples were obtained and haematology and chemistry tests were performed.

Funding

This research project was supported by a research grant from the Office of Research and Development of the University of Botswana, R649, (granted to M Motswaledi), Cape Peninsula University of Technology and National Research Foundation South Africa (CPUT/RJ23), granted to Prof OO Oguntibeju. However, the funders did not contribute in any way to the design of the study, data collection, analysis, or interpretation of data nor in the writing of the manuscript.

Competing interests:

The authors declare that they have no financial or personal relationship(s) that may have inappropriately influenced them in writing this article.

Author Contributions

MSM: Conceived the idea, carried out the experiments, analysed results and drafted paper. OO and IK provided experimental guidance, participated in data analysis, critiqued the content and gave final approval of the version to be published.

References

Appay, V. & Sauce, D. 2008. Immune activation and inflammation in HIV-1 infection: causes and consequences. *The Journal of Pathology*, 214, 231-241.

Badiou, S., Thiebaut, R., Aurillac-Lavignolle, V., Dabis, F., Laporte, F., Cristol, J. P. et al. 2008. Association of non-HDL cholesterol with subclinical atherosclerosis in HIV-positive patients. *Journal of Infection*, 57, 47-54.

Buchwald, H., O'dea, T. J., Menchaca, H. J., Michalek, V. N. & Rohde, T. D. 2000. Effect of plasma cholesterol on red blood cell oxygen transport. *Clinical and Experimenral Pharmacologyand Physiology*, 27, 951-5.

Chaar, V., Picot, J., Renaud, O., Bartolucci, P., Nzouakou, R., Bachir, D., et al. 2010. Aggregation of mononuclear and red blood cells through an {alpha}4{beta}1-Lu/basal cell adhesion molecule interaction in sickle cell disease. *Haematologica*, 95, 1841-8.

Chang, H.-Y., Chang, H.-M., Wu, T.-J., Chaing, C.-Y., Tzai, T.-S., Cheng, H.-L., et al. 2017. The role of Lutheran/basal cell adhesion molecule in human bladder carcinogenesis. *Journal of Biomedical Science*, 24, 61.

Chiodelli, P., Urbinati, C., Mitola, S., Tanghetti, E. & Rusnati, M. 2012. Sialic Acid Associated with $\alpha(v)\beta(3)$ Integrin Mediates HIV-1 Tat Protein Interaction and Endothelial Cell Proangiogenic Activation. *The Journal of Biological Chemistry*, 287, 20456-20466.

De Grandis, M., Cambot, M., Wautier, M.-P., Cassinat, B., Chomienne, C., Colin, Y., et al. 2013. Jak2v617f activates Lu/BCAM-mediated red cell adhesion in polycythemia vera through an EpoR-independent Rap1/Akt pathway. *Blood*, 121, 658-665.

Dhawan, S., Vargo, M. & Meltzer, M. 1992. Interactions between HIV-infected monocytes and the extracellular matrix: increased capacity of HIV-infected monocytes to adhere to and spread on extracellular matrix associated with changes in extent of virus replication and cytopathic effects in infected cells. *Journal of Leukocyte Biology*, 52, 62-69.

Frostegard, J., Nilsson, J., Haegerstrand, A., Hamsten, A., Wigzell, H. & Gidlund, M. 1990. Oxidized low density lipoprotein induces differentiation and adhesion of human monocytes and the monocytic cell line U937. *Proceedings of the. National. Academy of. Science. USA*, 87, 904-908.

Hartwig, H., Roig, C. S., Daemen, M., Lutgens, E. & Soehnlein, O. 2015. Neutrophils in atherosclerosis. *Hamostaseologie*, 35, 121-127.

Ionita, M. G., Van Den Borne, P., Catanzariti, L. M., Moll, F. L., De Vries, J.-P., Pasterkamp, G., et al. 2010. High Neutrophil Numbers in Human Carotid Atherosclerotic Plaques Are Associated With Characteristics of Rupture-Prone Lesions. *Arteriosclerosis Thrombosis and Vascular Biology*, 30, 1842-1848.

International Organization for Standardisation. Annexure C: Ethics in laboratory medicine. ISO 15189:2007:C9. Geneva:

Johnsen, S. H., Fosse, E., Joakimsen, O., Mathiesen, E. B., Stensland-Bugge, E., Njølstad, I. et al. 2005. Monocyte Count Is a Predictor of Novel Plaque Formation A 7-Year Follow-up Study of 2610 Persons Without Carotid Plaque at Baseline The Tromsø Study. *Stroke*, 36, 715-719.

Lafrenie, R. M., Wahl, L. M., Epstein, J. S., Hewlett, I. K., Yamada, K. M. & Dhawan, S. 1996. HIV-1-Tat modulates the function of monocytes and alters their interactions with microvessel endothelial cells. A mechanism of HIV pathogenesis. *Journal of Immunology*, 156, 1638-1645.

Mine, M., Moyo, S., Stevens, P., Michael, K., Novitsky, V., Makhaola, K., et al. 2011. Immunohaematological reference values for HIV-negative healthy adults in Botswana. *African Journal of Laboratory Medicine*, 1, 5-11.

Motswaledi, M. S., Kasvosve, I. & Oguntibeju, O. O. 2016. Blood Group Antigens C, Lub and P1 may have a role in HIV infection in Africans. *PLoS ONE*, 11, e0149883.

Oh, H., Mohler, E. R., 3rd, Tian, A., Baumgart, T. & Diamond, S. L. 2009a. Membrane cholesterol is a biomechanical regulator of neutrophil adhesion. *Arteriosclerosis, Thrombosis and Vascular Biology*, 29, 1290-7.

Oh, H., Mohler, E. R., Tian, A., Baumgart, T. & Diamond, S. L. 2009b. Membrane cholesterol is a biomechanical regulator of neutrophil adhesion. *Arteriosclerosis Thrombosis and Vascular Biology*, 29, 1290-1297.

Østerud, B. & Bjørklid, E. 2003. Role of Monocytes in Atherogenesis. *Physiological Reviews* 83, 1069-1112.

Pedraza, C., Geberhiwot, T., Ingerpuu, S., Assefa, D., Wondimu, Z., Kortesmaa, J., et al. 2000. Monocytic cells synthesize, adhere to, and migrate on laminin-8 ($\alpha 4\beta 1\gamma 1$). *Journal of Immunology*, 165, 5831–5838.

Von Hundelshausen, P., Weber, K. S., Huo, Y., Proudfoot, A. E., Nelson, P. J., Ley, K. et al. 2001. RANTES deposition by platelets triggers monocyte arrest on inflamed and atherosclerotic endothelium. *Circulation*, 103, 1772-7.

CHAPTER FIVE

RHCE*C PHENOTYPE MAY EXPLAIN THE GEOGRAPHICAL DISTRIBUTION OF HIV INFECTIONS

Short Title: RHCE*C and HIV susceptibility

AUTHORS:

Motswaledi, Modisa Sekhamo¹, MS, MT(ASCP)

Kasvosve, Ishmael², PhD

Oguntibeju, Oluwafemi Omoniyi³, PhD

AFFILIATIONS:

¹ Senior Lecturer, Department of Medical Laboratory Sciences, University of Botswana, Gaborone, Botswana, Motswaledims@ub.ac.bw

² Professor, Department of Medical Laboratory Sciences, University of Botswana, Gaborone, Botswana.

³ Professor, Biomedical Science, Cape Peninsula University of Technology, Cape Town, South Africa, OguntibejuO@cput.ac.za.

CORRESPONDING AUTHOR DETAILS

Name: Oluwafemi Omoniyi Oguntibeju

Department of Biomedical Sciences, Faculty of Health & Wellness Sciences, Cape Peninsula University of Technology, PO Box 1906, Cape Town, South Africa. 7535

Phone: +277-11400428

Email: OguntibejuO@cput.ac.za

Fax: +272-1 953 8490

Article submitted to PLOS ONE journal

Abstract

BACKGROUND: Sub-Saharan Africa and the Caribbean countries bear a disproportionate burden of the HIV pandemic. This suggests a genetic predisposition arising from a common ancestry. We previously reported a 40% HIV risk-reduction associated with RHCE*C blood group. The current study sought to elucidate potential mechanisms for this risk reduction.

MATERIALS AND METHODS: Lymphocyte sub-populations and viral load measurements were achieved by routine diagnostic laboratory methods in 102 untreated, HIV-1 infected patients. Results were compared across categories of RHCE*C blood group.

RESULTS: Carriage of the RHCE*C antigen was associated with a higher proportion of T-cells (82±8% versus 75±10%, p=0.001), especially single-positive (CD8⁺) cytotoxic T-cells (64±14 versus 54±13%, P=0.004). High absolute CD8 counts were more prevalent among C-positive than C-negative subjects (90% versus 65%, p=0.023). Among C-positive subjects, the increase in CD8 count paralleled viral load compared to C-negatives (r^2 =0.630, p<0.0001 versus r^2 =0.148, p<0.001, respectively). Moreover, examination of data involving 20 countries or regions indicated that high HIV prevalence occurred only in those countries with RHCE*C frequencies lower than 30% (χ^2 =20.0, p<0.0001).

DISCUSSION: Results suggest that RHCE*C enhances CD8 counts in HIV infection, which cells are known to play a vital role in immunity against HIV. Invariably, the only geographical locations with high HIV prevalence rates are those where the frequency of this antigen is low. Results provide a possible genetic explanation for the worldwide distribution of the HIV pandemic, with a special emphasis on the vulnerability of individuals of African descent, where the prevalence of RHCE*C is consistently low across many countries.

Keywords: HIV-prevalence; RHCE*C / RHCE*C; CD8 count; CD4 count; viral load; African.

Introduction

HIV prevalence in Botswana is among the highest in the world (Weiser et al., 2006). Many studies have revealed risk factors that have contributed to the spread of the virus, some of which include poverty, multiple sexual partners, alcohol and drug abuse as well as improper use of condoms (Keetile, 2014). Excessive alcohol consumption was reported as the greatest risk since it promoted risky behaviours in general (Weiser et al., 2006). In some studies, the risk of infection in women, especially in Botswana and Swaziland, was reported to be due in part to cultural norms that predispose women to sexual abuse or assign them to lower economic status (Shannon et al., 2012).

The high prevalence of HIV in Africa as well as those communities comprising of people of African descent, such as the Caribbean strongly suggests a common genetic link. However, few such links have been alluded to. In at least two studies, the Duffy antigen receptor for chemokines, was reported to promote HIV infection (He et al., 2008, Lachgar et al., 1998) among Africans. However, these reports were strongly refuted by other investigators (Winkler et al., 2009). In yet another study, the P^k blood group was reported to protect against HIV infection (Lund et al., 2009). However, this antigen is rare among Africans (Cooling, 2014). While it is generally accepted that host genetic factors play a role in immunity against the virus (Chatterjee, 2010), no specific hereditary factors have been advanced to explain the pandemic in Africa.

Although the function of many blood groups remains unknown, some reports have demonstrated that erythrocytes bind HIV and that such HIV becomes more efficiently transferred to CD4+(Beck et al., 2009, Beck et al., 2013, Garcia et al., 2012) cells. Erythrocyte antigens are therefore logical targets for erythrocyte-virus interactions. In our previous work, we reported a 40% risk reduction for HIV-1 infection in individuals of the RHCE*C blood group (also denoted as *RHCE*C*), while blood groups P_1 and Lu^b were associated with double and triple risks, respectively (Motswaledi et al., 2016). Of epidemiological importance was the observation that this protective blood group was very rare among Africans, while the risk-associated blood groups were much more common, raising the question of whether this antigenic profile could have contributed to the peculiar susceptibility of this population to infection with HIV-1.

RHCE*C is a component of the Rh blood group system. Antigens in this system are inherited in strong linkage disequilibrium on chromosome 1. The RHD gene codes for the D antigen while the RHCE gene caries a polymorphism that leads to the production of a range of RH phenotypes that include CE, Ce, cE or ce (Ripoche et al., 2004). The Rh gene products are organized on the erythrocyte membrane as a complex of proteins that include the Rh-associated glycoproteins (RhAG), LW, CD77, Duffy and CD47, which serve as ammonia transporters (Ripoche et al., 2004, Anstee and Tanner, 1993, Pourazar, 2007).

In the current study, we performed viral load, CD4 and CD8 counts and compared them among patients expressing or negative for RHCE*C. From these results and the review of available literature, we sought to find out if the observed risk reduction for HIV infection could be corroborated by clinical laboratory data.

Materials and methods

One hundred and twenty (120) HIV-1-infected (48 males and 72 females) treatment-naïve (self-reported) individuals were enrolled in the study. Since, no drug tests were done to verify their treatment-naïve status, 19 subjects with undetectable viral load were excluded on the presumption that they could be on treatment.

We used anonymized EDTA-anticoagulated samples from the Botswana-Harvard HIV Reference Laboratory in Gaborone. Ethical clearance was obtained from the University of Botswana's Office of Research and Development, Human Research and Development Committee of the Ministry of Health and Wellness, Gaborone District Health Management Team and the Research Ethics Committee of the Faculty of Health and Wellness Sciences at Cape Peninsula University of Technology. Individual consent was not needed since the study used anonymized, residual samples (ISO, 2007).

The viral load was measured using the Cobas® Taqman 48 analyser (Pleasanton, CA., USA). CD3, CD4, CD8 and CD45 counts obtained using the FacsCalibur ® flow cytometer (Becton-Dickinson, San Jose, CA, USA). The samples were further phenotyped for the C antigen using specific anti-C and anti-c antibodies (Fortress Diagnostics, Antrim, UK). The reactivity of the antisera was confirmed with C-positive and negative cells selected from an antibody identification panel, Bio-Rad DiaPanel®, Lot 45241.88.1, (Cressier FR, Switzerland).

Results of laboratory tests were interpreted in line with the reference ranges previously determined for the Botswana population (Mine et al., 2011).

The results were analyzed using the IBM SPSS version 24 statistical software. Correlations were used to study the effect of blood RHCE*C on the relationship between CD8%, log viral load, CD8% and CD4%. Results were considered significant only if P<0.05.

Results

Results from experimental data

The mean viral load did not differ between individuals positive (n=21) or negative (n=81) for C (p=0.398). In untreated, HIV-1-infected subjects, the CD4 count was significantly lower than that of the general uninfected population. On the contrary, the absolute CD8 count was significantly higher in the infected than uninfected subjects. The results are shown in Table 5.1.

Lymphocyte population	Mean Subjects (in (n=101	HIV-infected	Mean popula	in tion	normal	P-value
	Ū			(n=261)		
CD4 ⁺	307±208			859*			< 0.0001
CD8 ⁺	927±492			540*			< 0.0001

Table 5.1: Comparison of CD4 and CD8 in untreated, HIV-infected individuals and normal controls.

*One sample (T-test) mean used from a previous study(Mine et al., 2011).

In individuals expressing the RHCE*C antigen, 19/21 had a high absolute CD8 count compared to 52/80 who were RHCE*C-negative. A high CD8 count was therefore more associated with C-positive individuals than C-negative (90% versus 65%, p=0.023). The mean CD8 count between the two categories of RHCE*C, though higher in the C-positive group, did not reach statistical significance. However, the proportion of T-lymphocytes (CD3⁺) and cytotoxic (CD8⁺) T-cells was higher in the C-positive population as reflected in Table 5.2.

Table 5.2: Comparison of mean Lymphocyte sub-populations in untreated HIV-infected patients according to RhC phenotype. The RhC phenotype was associated with higher CD8 counts.

Lymphocyte Sub- population	C-Positive (Mean±SD) n = 21	C-Negative (Mean±SD)	P-value
		n = 80	
CD3+ (%)	82±8	75.0±10	0.001
CD8+ (%)	64±14	54±13	0.004
CD8 Absolute count (/µL)	1058±444	893±500	0.171

Among C-positive subjects, the increase in CD8 counts paralleled viral load more strongly than in C-negatives ($r^2=0.630$, p<0.0001 versus $r^2=0.148$, p<0.001, respectively). These results are shown in Fig 5.1.



Figure 5.1: Relationship between viral load and CD8% in C-positive subjects shows strong positive correlation (A: $r^2=0.148$, p<0.001, versus B: $r^2=0.630$, p<0.0001)

Results from country HIV prevalence data

A literature search was also undertaken to investigate HIV prevalence in countries where the frequencies of the RHCE*C are known. Data was obtained for 20 countries or communities as shown in Table 5.3. In brief, a high HIV prevalence in any country or community was associated with an RHCE*C frequency less than 30% (Pearson χ^2 =20.0, p<0.0001).

Country/Community	RHCE*C	HIV Prevalence
	Frequency	
Botswana	29.70	21.90
South Africa (Blacks)	18.80	18.90
Nigeria	28.2	2.9*
Cote d'Ivoire	21.97	2.70
Caribbean	24.50	1.60**
Uganda	3.62	6.5***
Thailand	51.50	1.10
Ethiopia (Blacks)	41.18	1.10
South Africa (Asians)	99.53	1.00
Mali	58.8	1.00
Brazil	63	0.6
Mauritania	42.69	0.5
Argentina	62	0.4
Laos	60.30	0.30
Southern India	88.00	0.30
Northern India	84.76	0.26
Sudan	58.40	0.20
Netherlands	68.00	0.20
Iran	75.90	0.10
China	78.22	0.04

Table 5.3: Comparison of RHCE*C with HIV prevalence across geographical regions

*5.8% in 2001. Some states at some point exceeded 8%, and current rates may only reflect the effect of treatment. Nevertheless, Nigeria carries the world's heaviest HIV burden (Gwaram, 2013).

**Prevalence rates range from 1-5% for different islands, depending on the reporting dates (Coggins, 2006, UNAIDS, 2016a). RHCE*C information for Curacao used here.

***Prevalence is down from 24.8% in 1994 (Kilian et al., 1999)

Discussion

RHCE*C, CD8 Parameters and protection against HIV infection

This study sought to establish the mechanism of protection provided by carriage of RHCE*C observed in a previous study. In this study, expression of this antigen was associated with a higher proportion of cytotoxic (CD8+) T-cells (CTL). In C-positive subjects, the CD8 absolute count varied in direct proportion to the viral load, suggesting a coordinated response to viremia. This correlation was much weaker in the C-negative group. Moreover, the proportion of individuals with high CD8 counts was greater in the C-positive group. All these observations suggest a role for this antigen in cytotoxic T-cell kinetics and function, and may therefore

explain the protective role of this blood group observed in our previous study (Motswaledi et al., 2016). Our findings are also consistent with recent findings in South Africa where the odds of HIV infection were reported to be nil for individuals who are homozygous for RHCE*CC (Davison et al., 2018). This prompted us to revisit a previous dataset involving 544 subjects. In this dataset only 6 individuals were homozygous for the C antigen and none of them were infected.

CTLs are important in the control of viremia in acute HIV-1 infections (Goonetilleke et al., 2009). They also correlate negatively with infection rates(Oxenius et al., 2004) and with good prognosis (Rinaldo et al., 1995). Moreover, they have been linked to slower progression of disease in non-progressors by producing HIV-1-specific CTL responses (Betts et al., 2001, Alimonti et al., 2006) and increased perforin production in HIV-infected non-progressors (Migueles et al., 2002).

The enhancement of CTLs points to a mechanism involving major histocompatibility class I (MHC-I) response, in which a cell-mediated rather than humoral response is invoked, and this kind of response is critical for viral infections such as HIV. This phenomenon appears to be enhanced when the RHCE*C antigen is expressed on erythrocytes. Erythrocytes and erythrocyte-bound platelets harbor HIV virions (Beck et al., 2013). In this connection, platelets have been shown to engage in direct MHC-I antigen presentation to CTLs (Chapman et al., 2012), a phenomenon that has also been documented in murine megakaryocytes (Zufferey et al., 2015). The RHCE*C phenotype appears to enhance this process, possibly by engaging its associated CD47 as the activator of CTLs (Seiffert et al., 2001).

Antigen presentation is normally accompanied by co-stimulatory responses that enhance clonal expansion of the specific CTL (Watts, 2010). We propose therefore, that the virus-plateleterythrocyte-T-cell interaction may explain the connection between RHCE*C and its protective role against HIV. Our results further suggest that the presence of C, rather than the other Rh antigens associated with CD47, may be responsible for the enhanced CTL response.

CD47 is the one erythrocyte antigen most commonly associated with T-cell activation events (Reinhold et al., 1997, Ticchioni et al., 2001, Seiffert et al., 2001, Pettersen et al., 1999). This glycoprotein is widely expressed on somatic cells including erythrocytes(Ripoche et al., 2004), a characteristic which makes the direct interaction of HIV-bearing erythrocytes with T-cells and phagocytes probable.

CD47 also interacts with its ligands on monocytes and T-lymphocytes in signal-transducing events. In this regard, it binds to signal regulatory protein- α (SIRP α) in a high-affinity interaction that results in T-cell activation(Seiffert et al., 2001). Binding of CD47 to SIRP β 2 enhances adhesion of T-cells to antigen-presenting cells and therefore enhances cell-mediated immunity(Piccio et al., 2005). In activated T cells, such as occur following HIV infection, CD47 promotes Fas-mediated apoptosis(Manna et al., 2005). All these events would work to eliminate HIV-infected CD4+ cells and therefore minimize the chances for an infection to be established.

Evidence from empirical HIV prevalence data

To further investigate its potential role in the epidemiology of HIV worldwide, we compared the frequency of the RHCE*C antigen across populations with varying degrees of HIV prevalence. Invariably, countries with an antigen frequency lower than 30% were consistently associated with highest prevalence rates in their regions. In Southern Africa, antigen frequency data was available only for Botswana (Motswaledi et al., 2016) and South Africa (Tax et al., 2002). Cote d'Ivoire and Nigeria (Gwaram, 2013) likewise had the lowest antigen frequencies (Bogui et al., 2014) and highest HIV prevalence in West Africa (UNAIDS, 2016a). In contrast, other African countries with higher antigen frequencies had low HIV-prevalence rates comparable to or slightly above other non-African countries, such as Argentina (Cotorruelo et al., 2008), Brazil (Guelsin et al., 2011), Thailand (Nathalang et al., 2001), Laos (Keokhamphoui et al., 2012), India (Makroo et al., 2013), Netherlands (Tax et al., 2002), Iran (Shokouhi Shoormasti et al., 2011) and China (Ma et al., 2018), where the RHCE*C frequency is higher than 40%. These African countries included Mali (Ba et al., 2015), Mauritania (Hamed et al., 2013), Ethiopia (Tax et al., 2002), Sudan (Elfadni et al., 2014) and South Africans of Asian origin (Tax et al., 2002).

Of further note and interest was the Caribbean statistics. The Caribbean Islands represent the second highest HIV-prevalence outside Africa (UNAIDS, 2016b). Apparently, a significant proportion of this population originates from Africa and the RHCE*C antigenic profile performed in Curacao (Tax et al., 2002) is similar to that of the African countries where high HIV rates are experienced. We propose that the RHCE*C antigen is important for HIV immunology and probably explains the genetic basis for the geographical distribution of the epidemic.

Conclusion

The results of this study indicate that high HIV rates occur only in geographical locations of low RHCE*C frequencies. The lack of this antigen in Botswana and countries with similar frequencies may have contributed to the rapid spread of HIV-1 in these populations. Results also suggest an immunological role of an Rh system antigen, unreported hitherto.

Acknowledgements

The authors wish to express gratitude to the management and staff of the Botswana-Harvard HIV Reference Laboratory for their kind support in performing flow cytometry analyses, viral load and DNA/RNA PCR. Gratitude is also expressed to the Staff and Management of the Gaborone District Health Management Team, especially those at Julia Molefe Clinic where samples were obtained and haematology and chemistry tests were performed. Gratitude is expressed for the funding support through a research grant from the Office of Research and Development of the University of Botswana, R649, (granted to M Motswaledi), Cape Peninsula University of Technology and National Research Foundation South Africa (CPUT/RJ23), granted to Prof OO Oguntibeju.

References

Alimonti, J. B., Kimani, J., Matu, L., Wachihi, C., Kaul, R., Plummer, F. A. et al. 2006. Characterization of CD8+ T-cell responses in HIV-1-exposed seronegative commercial sex workers from Nairobi, Kenya. *Immunology and Cell Biology*, 84, 482-485.

Anstee, D. J. & Tanner, M. J. 1993. Biochemical aspects of the blood group Rh (rhesus) antigens. *Baillieres Clinical Haematology*, 6, 401-22.

Ba, A., Beley, S., Chiaroni, J., Bailly, P. & Silvy, M. 2015. RH diversity in Mali: characterization of a new haplotype RHD*DIVa/RHCE*ceTI(D2). *Transfusion*, 55, 1423-31.

Beck, Z., Brown, B. K., Wieczorek, L., Peachman, K. K., Matyas, G. R., Polonis, V. R., et al. 2009. Human erythrocytes selectively bind and enrich infectious HIV-1 virions. *PLoS One*, 4.

Beck, Z., Jagodzinski, L. L., Eller, M. A., Thelian, D., Matyas, G. R., Kunz, A. N. et al. 2013. Platelets and erythrocyte-bound platelets bind infectious HIV-1 in plasma of chronically infected patients. *PLoS One*, *8*, e81002.

Betts, M. R., Ambrozak, D. R., Douek, D. C., Bonhoeffer, S., Brenchley, J. M., Casazza, J. P., et al. 2001. Analysis of total human immunodeficiency virus (HIV)-specific CD4+ and CD8+ T-cell responses: relationship to viral load in untreated HIV infection. *Journal of Virology*, 75, 11983-11991.

Bogui, L. S., Dembele, B., Sekongo, Y., Abisse, S., Konaté, S. & Sombo, M. 2014. Phenotypic profile of Rh and Kell blood group systems among blood donors in Cote d'Ivoire, West Africa. *Journal of Blood Transfusion*, 2014.

Chapman, L. M., Aggrey, A. A., Field, D. J., Srivastava, K., Ture, S., Yui, K., et al. 2012. Platelets present antigen in the context of MHC class I. *Journal of Immunology*, 189, 916-923.

Chatterjee, K. 2010. Host genetic factors in susceptibility to HIV-1 infection and progression to AIDS. *Journal of Genetics*, 89, 109-16.

Coggins, P. C. 2006. Cultural Issues for HIV/AIDS Clients from the Caribbean.Myths and Realities. *Annual Grantee meeting of the Federal Government's Center for Substance Abuse Treatment Targeted Capacity Annual Expansion HIV and HIV Outreach* Washington, D.C.

Cooling, L. 2014. P Blood Groups/ Glob collection. *In:* Fung, M. K., Grossman, B. J., Hillyer, C. D. & Westhoff, C. M. (eds.) *Technical Manual.* 18 ed. Bethesda, Maryland: American Association of Blood Banks.

Cotorruelo, C. M., Fiori, S. V., Borrás, S. E. G., Racca, L. L., Biondi, C. S. & Racca, A. L. 2008. Distribution of the FYB ES and RHCE*ce(733C>G) alleles in an Argentinean population: Implications for transfusion medicine. *BMC Medical Genetics*, 9, 40.

Davison, G. M., Hendrickse, H. L. & Matsha, T. E. 2018. The relationship between immunogenic red blood cell antigens and Human Immunodeficiency Virus infection. *Transfusion and Apheresis Science*, 57, 58-62.

Elfadni, M., Yassin, H., Abushama, H., Khalid, E. & Khalil, E. 2014. Rh Blood group system phenotypes, haplotypes and probable genotypes among major sudanese tribes. *International Journal of Current Research*, 6, 7862-7866.

Garcia, M. N., Dos Ramos Farias, M. S., Fazzi, L., Grasso, D., Rabinovich, R. D. et al. 2012. Presence of IgG anti-gp160/120 antibodies confers higher HIV capture capacity to erythrocytes from HIV-positive individuals. *PLoS One*, *7*, e45808.

Goonetilleke, N., Liu, M. K., Salazar-Gonzalez, J. F., Ferrari, G., Giorgi, E., Ganusov, V. V., et al. 2009. The first T cell response to transmitted/founder virus contributes to the control of acute viremia in HIV-1 infection. *Journal of Experimental Medicine*, 206, 1253-1272.

Guelsin, G. A. S., Sell, A. M., Castilho, L., Masaki, V. L., De Melo, F. C., Hashimoto, M. N., et al. 2011. Genetic polymorphisms of Rh, Kell, Duffy and Kidd systems in a population from the State of Paraná, southern Brazil. *Revista Brasileira de Hematologia e Hemoterapia*, 33, 21-25.

Gwaram, B. 2013. Prevalence of Rh Phenotypes among Blood Donors in Kano, Nigeria. *J Med Trop*, 15, 37-39.

Hamed, C. T., Bollahi, M. A., Abdelhamid, I., Sow, A., Sidi, A. M., Habti, N. et al. 2013. Distribution of Rhesus and Kell blood group frequencies in the Mauritanian population. *Blood Transfusion*, 11, 154-155.

He, W., Neil, S., Kulkarni, H., Wright , E., Agan, B. K. 2008. Duffy antigen receptor for chemokines mediates trans-infection of HIV-1 from red blood cells to target cells and affects HIV-AIDS susceptibility. Cell Host Microbe, 4, 52–62.

International Organization for Standardisation. Annexure C: Ethics in laboratory medicine. ISO 15189:2007:C9. Geneva:

Keetile, M. 2014. High-risk behaviors among adult men and women in Botswana: Implications for HIV/AIDS prevention efforts. *Sahara Journal*, 11, 158-166.

Keokhamphoui, C., Urwijitaroon, Y., Kongphaly, D. & Thammavong, T. 2012. Blood group antigen distribution in Lao blood donors. *Immunohematology*, 28, 132-6.

Kilian, A. H., Gregson, S., Ndyanabangi, B., Walusaga, K., Kipp, W., Sahlmüller, G., et al. 1999. Reductions in risk behaviour provide the most consistent explanation for declining HIV-1 prevalence in Uganda. *AIDS*, 13, 391-398.

Lachgar, A., Jaureguiberry, G., Le Buenac, H., Bizzini, B., Zagury, J. F., Rappaport, J. et al. 1998. Binding of HIV-1 to RBCs involves the Duffy antigen receptors for chemokines (DARC). *Biomedical Pharmacotherapy*, 52, 436-439.

Lund, N., Olsson, M. L., Ramkumar, S., Sakac, D., Yahalom, V., Levene, C., et al. 2009. The human Pk histo-blood group antigen provides protection against HIV-1 infection. *Blood Reviews*, 113, 4980-4991.

Ma, T., Yang, J., Song, Y., Wang, W., Xie, X., Chen, P., et al. 2018. [Distribution of Rh blood group in 51 283 cases of inpatients and voluntary blood donors]. *Xi Bao Yu Fen Zi Mian Yi Xue Za Zhi*, 34, 70-74.

Makroo, R. N., Bhatia, A., Gupta, R. & Phillip, J. 2013. Prevalence of Rh, Duffy, Kell, Kidd & MNSs blood group antigens in the Indian blood donor population. *Indian Journal of Medical Research*, 137, March 2013, pp 521-526.

Manna, P. P., Dimitry, J., Oldenborg, P. A. & Frazier, W. A. 2005. Cd47 augments Fas/CD95mediated apoptosis. *Journal of Biological Chemistry*, 280, 29637-44.

Migueles, S. A., Laborico, A. C., Shupert, W. L., Sabbaghian, M. S., Rabin, R., Hallahan, C. W., et al. 2002. HIV-specific CD8+ T cell proliferation is coupled to perforin expression and is maintained in nonprogressors. *Nature immunology*, **3**, 1061-1068.

Mine, M., Moyo, S., Stevens, P., Michael, K., Novitsky, V., Makhaola, K., et al. 2011. Immunohaematological reference values for HIV-negative healthy adults in Botswana. *African Journal of Laboratory Medicine*, 1, 5-11.

Motswaledi, M. S., Kasvosve, I. & Oguntibeju, O. O. 2016. Blood Group Antigens C, Lub and P1 may have a role in HIV infection in Africans. *PLoS ONE*, 11, e0149883.

Nathalang, O., Kuvanont, S., Punyaprasiddhi, P., Tasaniyanonda, C. & Sriphaisal, T. 2001. A preliminary study of the distribution of blood group systems in Thai blood donors determined by the gel test. *Southeast Asian Journal of Tropical Medicine and Public Health*, 32, 204-7.

Oxenius, A., Price, D. A., Hersberger, M., Schlaepfer, E., Weber, R., Weber, M., et al. 2004. HIV-specific cellular immune response is inversely correlated with disease progression as defined by decline of CD4+ T cells in relation to HIV RNA load. *Journal of Infectious Disease*, 189, 1199-1208. Pettersen, R. D., Hestdal, K., Olafsen, M. K., Lie, S. O. & Lindberg, F. P. 1999. CD47 Signals T Cell Death. *Journal of Immunology*, 162, 7031-7040.

Piccio, L., Vermi, W., Boles, K. S., Fuchs, A., Strader, C. A., Facchetti, F., et al. 2005. Adhesion of human T cells to antigen-presenting cells through SIRPβ2-CD47 interaction costimulates T-cell proliferation. *Blood*, 105, 2421-2427.

Pourazar, A. 2007. Red cell antigens: Structure and function. Asian J Transfus Sci, 1, 24-32.

Reinhold, M. I., Lindberg, F. P., Kersh, G. J., Allen, P. M. & Brown, E. J. 1997. Costimulation of T Cell Activation by Integrin-associated Protein (CD47) Is an Adhesion-dependent, CD28-independent Signaling Pathway. *The Journal of Experimental Medicine*, 185, 1-12.

Rinaldo, C., Huang, X.-L., Fan, Z., Ding, M., Beltz, L., Logar, A., Panicali, D., Mazzara, G., Liebmann, J. & Cottrill, M. 1995. High levels of anti-human immunodeficiency virus type 1 (HIV-1) memory cytotoxic T-lymphocyte activity and low viral load are associated with lack of disease in HIV-1-infected long-term nonprogressors. *J Virol*, 69, 5838-5842.

Ripoche, P., Bertrand, O., Gane, P., Birkenmeier, C., Colin, Y. & Cartron, J.-P. 2004. Human Rhesus-associated glycoprotein mediates facilitated transport of NH3 into red blood cells. *Proceedings of the National Academy of Science USA*, 101, 17222-17227.

Seiffert, M., Brossart, P., Cant, C., Cella, M., Colonna, M., Brugger, W., et al. 2001. Signal-regulatory protein α (SIRP α) but not SIRP β is involved in T-cell activation, binds to CD47 with high affinity, and is expressed on immature CD34⁺CD38⁻hematopoietic cells. *Blood*, 97, 2741-2749.

Shannon, K., Leiter, K., Phaladze, N., Hlanze, Z., Tsai, A. C., Heisler, M., Iacopino, V. & Weiser, S. D. 2012. Gender inequity norms are associated with increased male-perpetrated rape and sexual risks for HIV infection in Botswana and Swaziland. *PLoS One*, *7*, e28739.

Shokouhi Shoormasti, R., Azimdoost, A., Saghafi, S., Movahhedi, M., Haghi Ashtiani, M. T., Pourpak, Z. & Eslami, M. B. 2011. Normal range determination of lymphocytes subsets in normal adults in Iran. *Iranian Journal of Allergy and Asthma Immunology*, 10, 295-8.

Tax, M. G., Van Der Schoot, C. E., Van Doorn, R., Douglas-Berger, L., Van Rhenen, D. J. & Maaskant-Vanwijk, P. A. 2002. RHC and RHc genotyping in different ethnic groups. *Transfusion*, 42, 634-44.

Ticchioni, M., Raimondi, V., Lamy, L., Wijdenes, J., Lindberg, F. P., Brown, E. J. et al. 2001. Integrin-associated protein (CD47/IAP) contributes to T cell arrest on inflammatory vascular endothelium under flow. *The FASEB Journal*, 15, 341-350.

UNAIDS. 2016a. *HIV Prevalence* [Online]. Geneva: UNAIDS. Available: <u>http://aidsinfo.unaids.org/</u> [Accessed 19th April 2018.

UNAIDS. 2016b. UNDAIDS fact sheet 06-Carribean [Online]. Available: http://www.who.int/hiv/mediacentre/200605-FS Caribbean en.pdf.

Watts, T. H. 2010. Staying alive: T cell costimulation, CD28, and Bcl-xL. *Journal of Immunology*, 185, 3785-7.

Weiser, S. D., Leiter, K., Heisler, M., Mcfarland, W., Percy-De Korte, F., Demonner, S. M., et al. 2006. A population-based study on alcohol and high-risk sexual behaviors in Botswana. *PLoS Med*, 3, e392.

Winkler, C. A., An, P., Johnson, R., Nelson, G. W. & Kirk, G. 2009. Expression of Duffy antigen receptor for chemokines (DARC) has no effect on HIV-1 acquisition or progression to AIDS in African Americans. *Cell Host Microbe*, *5*, 411-413.

Zufferey, A., Speck, E. R., Guo, L., Aslam, R., Kapur, R. & Semple, J. W. 2015. Murine Bone Marrow-Derived Megakaryocytes Are Capable of Antigen Cross-Presentation on Major Histocompatibility Class (MHC) I Molecules. *Blood*, 126, 3465-3465.

CHAPTER SIX

HIV-1 RNA IS DETECTABLE IN P1 ERYTHROCYTE LYSATE

Modisa Sekhamo Motswaledi, MS^{1, 2}, Ishmael Kasvosve, PhD¹, Oluwafemi Omoniyi Oguntibeju, PhD²

¹ Department of Medical Laboratory Sciences, University of Botswana Faculty of Health Sciences, Block 246 Room A215. Phone +267-355-2572, Fax: +267-307-4538). Email: <u>modisa.motswaledi@mopipi.ub.bw</u> (Co-corresponding author), <u>ishmael.kasvosve@mopipi.ub.bw</u>, ² Department of Biomedical Sciences, Faculty of Health & Wellness Sciences, Cape Peninsula University of Technology, PO Box 1906, 7535, Bellville, South Africa. Fax: +2721 953 8490. Email: <u>OguntibejuO@cput.ac.za</u> (corresponding author).

Running Title: P1 Blood Group and risk of HIV infection

Article submitted to Transfusion and Apheresis Science journal

Abstract

BACKGROUND: Many studies have implicated erythrocytes in HIV-1 pathogenesis and as viral reservoirs. We previously reported a high risk for HIV infection associated with P_1 blood group.in Africans. P_1 belongs to a group of glycolipids known to facilitate viral membrane fusion. The aim of this study was to investigate the possibility of HIV RNA inside P1 erythrocytes.

METHODS: Viral load measurements were made in 120 untreated, HIV-1 infected patients. Qualitative PCR (n=27) was done on purified erythrocytes to detect erythrocyte-associated HIV-1 RNA. Results were then compared across categories of P_1 .

RESULTS: Viral RNA was detected in both lysate and membranous fraction of purified erythrocytes, including those with undetectable plasma viral load. Plasma viral load did not differ between individuals positive or negative for P₁. However, purified P₁-positive erythrocyte lysate contained more viral RNA (Mann-Whitney U-test, P= 0.032). Erythrocyte-associated RNA was independent of plasma VL.

CONCLUSIONS: The higher level of intra-erythrocytic RNA in P_1 subjects is a novel finding consistent with viral fusion facilitated by glycosphingolipids, possibly accounting for the observed risk of infection associated with P_1 . The study also suggests a new viral reservoir hitherto undescribed.

Key words: HIV-1 RNA; P1 blood group; viral load; glycosphingolipids, erythrocytes.
Introduction

The P-blood group system is comprised to three antigens – P^k , P and P1. The antigens are glycosphingolipids (GLS) built by addition of sugars to a galactosyl ceramide, the basic sphingolipid backbone(Hellberg, 2007). The P antigen is derived by addition of sugars to the P^k antigen. The P₁ antigen synthesis follows a separate pathway resulting in a more complex structure carrying more glycosyl residues(Spitalnik and Spitalnik, 1995). In blood transfusion, the P₁-blood may occasionally cause transfusion reactions(Spitalnik and Spitalnik, 1995), although most of the cases are generally of the cold-reacting IgM class and therefore clinically insignificant. This blood group occurs among individuals of African descent with high frequency but relatively rare among Caucasians(Hellberg, 2007).

Seemingly contradictory reports have been published regarding the role of the P^k antigen with respect to predisposition to HIV infection. In some reports, GLS were reported to promote HIV membrane fusion to susceptible CD4 cells (Hammache et al., 1999, Puri et al., 1998a). In particular, the P^k blood group was thought to enhance infection with X4 viruses while P enhanced R5 strains(Puri et al., 1998a). In other reports, lipid rafts comprising of GSL and cholesterol were considered to be sufficient for viral fusion without the need for correceptors(Yang et al., 2015, Rawat et al., 2006).

In a contrary report, the P^k blood group was reported to be protective against HIV infection in vitro(Lund et al., 2009). This perspective was further strengthened by the observation that patients with Fabry Disease, in which an intracellular accumulation of the P^k antigen occurred, were resistant to HIV infection (Lund et al., 2005, Ramkumar et al., 2009). A possible explanation of this apparent contradiction was that optimal amounts of P^k are required for viral fusion. However, excess quantities of GSL protects the cell from infection by competing for the virus and thus preventing it from effectively fusing to the cell membrane(Lund et al., 2009).

To date, no reports specifically linking the P_1 antigen to the risk of HIV infection have been found. However, a number of other reports have demonstrated that HIV directly interacts with RBC, some via glycosphingolipids, and this interaction may be important in the epidemiology and pathogenesis of the virus (Beck et al., 2009). For example, in 2002, Hess and colleagues reported some HIV-1 to be associated with erythrocyte membranous ghosts in about 98% of HIV-1-infected patients (Hess et al., 2002). The erythrocyte-associated HIV was detectable in one third of the patients in whom plasma viral load was undetectable. The erythrocyteassociated HIV in some of the patients exceeded that associated with leukocytes and was associated with advanced clinical stages of the disease. Subsequent reports by Fierer and colleagues (2007), however, strongly disputed the presence of erythrocyte-associated HIV in patients with long term viral suppression while acknowledging it in those who have not achieved viral suppression.

The binding of the virus to the erythrocytes has been attributed to a number of mechanisms, including the use of complement receptors normally used for clearance of antigen/antibody (immune) complexes(Horakova et al., 2004). Although some of the virus was bound to erythrocytes via the normal mechanisms of immune-complex clearance using complement receptor 1 (CR1), these investigators also noted that the virus also bound to erythrocytes via a

non-complement receptor pathway, suggesting that molecules other than complement may be operational in this association(Horakova et al., 2004).

In 2012, Garcia and colleagues (2011) reported the presence of anti-gp160/120 antibodies on erythrocyte surfaces. In addition, they documented that such antibodies on the red cells were responsible for capturing free plasma viral particles and may account for reduction of plasma viral load. This followed their earlier report in which viral p24 was detected on erythrocytes even in patients with undetectable plasma viral load (Garcia et al., 2011).

In addition to erythrocytes, P_1 antigen is also expressed on lymphocytes (Pourazar, 2007) and therefore presents a theoretical possibility of heightened risk of HIV-infection for subjects expressing the P_1 antigen.

In the current study, we investigated the possibility of the virus being present inside erythrocytes solely on the basis of the fusogenic property of the P_1GSL . If found inside the erythrocytes, we also sought to find out if this was dependent on plasma viral load.

Methods

The study enrolled120 residual blood samples from HIV-1-infected individuals, (48 males and 72 females) prior to initiation of highly active anti-retroviral therapy at Julia Molefe Clinic in Gaborone, between May and June 2016.

. The study was approved by the University of Botswana's Office of Research and Development, Human Research and Development Committee of the Ministry of Health and Wellness, Gaborone District Health Management Team and the Research Ethics Committee, of the faculty of health and Wellness Sciences, Cape Peninsula University of Technology (Appendix A-D). Individual consent was not needed since the study used anonymized, residual samples(ISO, 2007).

The viral load was determined by use of the Cobas® Taqman 48 analyser (Pleasanton, CA., USA), while the RNA qualitative test for HIV-1 was performed using the COBAS® Ampliprep/COBAS® Taqman HIV-1 qualitative test, version 2.0 (Pleasanton, CA., USA). The samples were further phenotyped for using anti-P₁ antibody (Fortress Diagnostics, Antrim, UK) according to the manufacturer's instructions. The reactivity of the antiserum was confirmed by use of P₁-positive and P₁-negative erythrocytes selected from an antibody identification panel, Bio-Rad DiaPanel®, Lot 45241.88.1, (Cressier FR, Switzerland). CD4, CD3, CD8 and CD45 counts were enumerated using the FacsCalibur ® flow cytometer (Becton-Dickinson, San Jose, CA. USA).

Viral RNA detection was performed on purified erythrocytes. To achieve this, 1 ml of the packed red cells was diluted with an equal volume of normal saline and then overlaid on 14ml of ficoll, followed by centrifugation at 400xG for 30 minutes. The suspended leukocyte layer was removed by aspiration together with all the saline and ficoll above the erythrocytes. The remaining cells were again diluted in 1 ml normal saline and subjected to a second round of ficoll gradient centrifugation. The cells were then washed three times with 14 ml of normal saline. The supernatant of the last wash was tested for viral RNA or DNA using the COBAS®

AmpliPrep/COBAS® Taqman® (Pleasanton, CA., USA) and none was detectable, indicating a successful removal of unbound viral nucleic acids.

A sample was reconstituted for CD45 testing by adding equal volumes of the purified packed red cells and normal saline to bring the haematocrit to 41.3±1.8%. The CD45 determination was repeated to assess white cell depletion and a reduction of more than 97% after the first and over 99% after the second purification was achieved.

Subsequently, the purified cells were lysed with 1 ml of distilled water and allowed to lyse on a rotary mixer for 5 minutes before being centrifuged at 1000xG for 3 minutes to pellet erythrocytic ghosts and any virus that might have been attached to them. The top 500μ L of the RBC lysate was removed and tested for viral nucleic acids as described above, to demonstrate the presence, if any, of intra-erythrocytic viral RNA in the lysate.

The remainder of the sample was mixed by vortexing and viral RNA was also determined, which would represent the membrane-bound RNA plus that from the residual hemolysate. The number of PCR cycles taken to show a positive reaction in the membranous fraction or lysate was used as a proxy for viral concentrations in the two fractions against an internal standard. The median difference in cycles to reach a positive reaction were compared according to P₁-carriage for statistical significance (number of cycles for lysate minus the number of cycles for stroma). For the purposes of comparison in this study, a high viral load in HIV-infected patients was defined as any value above $Log_{10}VL=4.97$, the viral set-point previously determined for this population. The results were analysed using the IBM SPSS version 24 statistical software.

Results

The viral load and lymphocyte populations did not differ between individuals positive or negative for P_1 . However, the median number of additional cycles required to reach a positive reaction for erythrocyte lysate-associated viral RNA was smaller for P_1 -positive individuals than for the P_1 -negative (Mann-Whitney U-Rank-Sum test, P=0.032), indicating a higher intracellular content of viral RNA in P_1 -positive erythrocytes. The data is presented in Table 6.1.

Blood Group (n=10)	No of PCR Cycles		
	Median	IQR	
P1-positive	0.9	0.3	
P ₁ -negative	4.5	3	
P-value	0.032		

 Table 6. 1 - Comparison of RNA-PCR results in untreated HIV-1-infected patients.

Erythrocyte-associated RNA (E-RNA) was estimated in 27 patients and was detectable in 22 (81%) and undetectable in 5. Of these 5, only one had undetectable plasma viral load (pVL). On the other hand, E-RNA was demonstrable in 6 patients who had undetectable pVL. Of the 22 with demonstrable E-RNA, 10/22 (45%) demonstrated the virus on both the stroma and

lysate. Most of the E-RNA was associated with the stroma than the lysate. Only one sample yielded more RNA from the lysate than the stroma.

We also compared the prevalence of HIV across countries for which P1 frequencies were known. The results are shown in Table 6.2

Table 6.2: Comparison of P_1 frequency with HIV prevalence in various countries showing that high P_1 frequency is associated with high HIV prevalence

Country	P ₁ frequency (%)	HIV Prevalence (%)	
Botswana	74.6	21.9	
Uganda	89	6.5	
Laos	18.97	0.3	
Turkey	0.5	*	
Malaysia	40	0.4	
Thailand	30	1.1	

*Data for Turkey not available, but official reports suggest low prevalence (Ay and Karabey, 2006)

Discussion

The presence of HIV virions on erythrocytes and erythrocyte-bound platelets has been documented by various reports (Beck et al., 2013, Hess et al., 2002, Garcia et al., 2011). However, some investigators have refuted the presence of erythrocyte-bound HIV in virally suppressed patients(Fierer et al., 2007). The current study confirms the presence of viral RNA from purified erythrocytes in patients with undetectable plasma viral load. The detection of intra-erythrocytic viral RNA is another novel finding hitherto unreported.

Although the P₁ positive and P₁-negative groups did not differ with respect to plasma viral load, the PCR analysis of the erythrocyte lysate revealed the presence of more intra-erythrocytic viral RNA in P₁-positive than negative samples. This observation is consistent with the role of glycosphingolipids in enhancing viral fusion (Hammache et al., 1999, Puri et al., 2004, Puri et al., 1998b, Puri et al., 1998a) and probably explains the increased risk of HIV infection observed in P₁-positive individuals (Motswaledi et al., 2016). P₁ has a wide tissue distribution (Hellberg et al., 2013), including its expression on HIV-susceptible cells such as monocytes and lymphocytes (Dunstan, 1986, Pourazar, 2007). Our results corroborate the findings of Yang et al. (2015) who reported that lipid rafts, which are rich in sphingolipids are "necessary and sufficient for membrane targeting and fusion" in human immunodeficiency virus infection

Erythrocyte-associated HIV (E-RNA) was demonstrable in the majority of patients with or without detectable plasma viral load (pVL), consistent with other reports (Beck et al., 2009, Hess et al., 2002, Garcia et al., 2012, Garcia et al., 2011). However, E-RNA could not be demonstrated in some samples with detectable or even high pVL, suggesting that the association of the virus with erythrocytes is not simply a function of plasma viral load. The

detection of viral RNA in erythrocyte lysate suggests alternative pathways for viral entry into cells independent of CD4+ as suggested by other investigators (Garcia et al., 2012). However, this analysis was limited by the relatively small sample size for intra-erythrocytic viral RNA.

Our results suggest that the risk of infection associated with P_1 potentially emanates from its ability to promote viral fusion. Furthermore, the presence of viral RNA in erythrocyte lysate suggests an additional viral reservoir that may need to be addressed for successful eradication of HIV infection. This blood group has a high frequency among individuals of African descent and may have contributed to the high infection rates in this population.

We also looked at HIV prevalence among countries for which P1 frequencies were known. Although data was only available from five countries, the only country with a high prevalence rate was also the only country with a high P1 frequency. The presence of P_1 can therefore not be excluded as a contributing factor to HIV prevalence in those countries.

Conclusion

We report for the first time, the occurrence of HIV RNA in the lysate of erythrocytes. This phenomenon is independent of plasma viral load and potentially explains a GSL-based, receptor-independent entry of the virus into those cells possessing adequate levels of membrane GSL. Furthermore, The low frequency of P1 in countries with low HIV prevalence might corroborate our findings that P1 carriage increases vulnerability to HIV infection.

Authors' disclosure

The authors declare that there is no conflict of interest in this study.

Acknowledgement

This research project was supported by a research grant from the Office of Research and Development of the University of Botswana, R649, (granted to M Motswaledi), Cape Peninsula University of Technology and National Research Foundation South Africa (CPUT/RJ23), granted to Prof OO Oguntibeju. However, the funders did not contribute in any way to the design of the study, data collection, analysis, or interpretation of data nor in the writing of the manuscript. The authors wish to express gratitude to the management and staff of the Botswana-Harvard HIV Reference Laboratory for their kind support in performing flow cytometry analyses, viral load and DNA/RNA PCR. Gratitude is also expressed to the Staff and Management of the Gaborone District Health Management Team, especially those at Julia Molefe Clinic where samples were obtained and haematology and chemistry tests were performed.

References

Ay, P. & Karabey, S. 2006. Is there a "hidden hiv/aids epidemic" in turkey: the gap between the numbers and the facts. *Marmara Medical Journal*, 19, 90-97

Beck, Z., Brown, B. K., Wieczorek, L., Peachman, K. K., Matyas, G. R., Polonis, V. R., et al. 2009. Human erythrocytes selectively bind and enrich infectious HIV-1 virions. *PLoS One*, 4.

Beck, Z., Jagodzinski, L. L., Eller, M. A., Thelian, D., Matyas, G. R., Kunz, A. N. & Alving, C. R. 2013. Platelets and erythrocyte-bound platelets bind infectious HIV-1 in plasma of chronically infected patients. *PLoS One*, *8*, e81002.

Dunstan, R. A. 1986. Status of major red cell blood group antigens on neutrophils, lymphocytes and monocytes. *British Journal of Haematology*, 62, 301-9.

Fierer, D. S., Jr., J. V., Patel, N. & Clover, G. 2007. Absence of Erythrocyte-associated HIV-1 in vivo. *Journal of Infectious Disease* 196, 587–590.

Garcia, M., Dos Ramos Farias, M., A' Vila, M. & Rabinovich, R. 2011. Presence of p24-Antigen Associated to Erythrocyte in HIV-Positive Individuals Even in Patients with Undetectable Plasma Viral Load. *PLoS ONE*, 6, 10.1371/journal.pone.0014544.

Garcia, M. N., Dos Ramos Farias, M. S., Fazzi, L., Grasso, D., Rabinovich, R. D. & Avila, M. M. 2012. Presence of IgG anti-gp160/120 antibodies confers higher HIV capture capacity to erythrocytes from HIV-positive individuals. *PLoS One*, *7*, e45808.

Hammache, D., Yahi, N., Maresca, M., Pie'Roni, G. & Fantini, J. 1999. Human erythrocyte glycosphingolipids as alternative cofactors for human immunodeficiency virus type 1 (HIV-1) entry: evidence for CD4-induced interactions between HIV-1 gp120 and reconstituted membrane microdomains of glycosphingolipids (Gb3 and GM3). *J Virol*, 73, 5244–5248.

Hellberg, Å. 2007. Studies on the Genetic Basis of Pk, P and P1 Blood Group Antigen Expression. PhD Doctoral Thesis, Lund.

Hellberg, A., Westman, J. S., Thuresson, B. & Olsson, M. L. 2013. P1pk: The blood group system that changed its name and expanded. *Immunohematology*, 29, 25-33.

Hess, C., Klimkait, T., Schlapbach, L., Zenero, V. D., Sadallah, S. & Horakova, E. 2002. Association of a pool of HIV-1 with erythrocytes in vivo: a cohort study. *Lancet*, 359, 2230-4.

Horakova, E., Gasser, O., Sadallah, S., Inal, J., Bourgeois, G., Ziekau, I., et al. 2004. Complement mediates the binding of HIV to erythrocytes. *Journal of Immunology*, 173, 4236-41.

International Organization for Standardisation. Annexure C: Ethics in laboratory medicine. ISO 15189:2007:C9. Geneva:

Lund, N., Branch, D. R., Sakac, D., Lingwood, C. A., Siatskas, C., Robinson, C. J., et al. 2005. Lack of susceptibility of cells from patients with Fabry disease to productive infection with R5 human immunodeficiency virus. *AIDS*, 19, 1543-5. Lund, N., Olsson, M. L., Ramkumar, S., Sakac, D., Yahalom, V., Levene, C., et al. 2009. The human Pk histo-blood group antigen provides protection against HIV-1 infection. *Blood Reviews*, 113, 4980-4991.

Motswaledi, M. S., Kasvosve, I. & Oguntibeju, O. O. 2016. Blood Group Antigens C, Lub and P1 may have a role in HIV infection in Africans. *PLoS ONE*, 11, e0149883.

Pourazar, A. 2007. Red cell antigens: Structure and function. *Asian Journal of Transfusion Science*, 1, 24-32.

Puri, A., Hug, P., Jernigan, K., Barchi, J., Kim, H.-Y., Hamilton, J., et al. 1998a. The neutral glycosphingolipid globotriaosylceramide promotes fusion mediated by a CD4-dependent CXCR4-utilizing HIV type 1 envelope glycoprotein. *Proceedings of the National Academcy of Science*, 95, 14435-14440.

Puri, A., Hug, P., Mun^oz-Barroso, I. & Blumenthal, R. 1998b. Human erythrocyte glycolipids promote HIV-1 envelope glycoprotein-mediated fusion of CD4+ cells. *Biochemical and Biophysical Research Communications*, 242, 219–225.

Puri, A., Rawat, S. S., Lin, H.-M. J., Finnegan, C. M., Mikovits, J., Ruscetti, F. W. et al. 2004. An inhibitor of glycosphingolipid metabolism blocks HIV-1 infection of primary T-cells. *AIDS*, 18, 849-858.

Ramkumar, S., Sakac, D., Binnington, B., Branch, D. R. & Lingwood, C. A. 2009. Induction of HIV-1 resistance: cell susceptibility to infection is an inverse function of globotriaosyl ceramide levels. *Glycobiology*, 19, 76-82.

Rawat, S. S., Viard, M., Gallo, S. A., Blumenthal, R. & Puri, A. 2006. Sphingolipids, cholesterol, and HIV-1: A paradigm in viral fusion. *Glycoconjugate Journal*, 23, 189-197.

Spitalnik, P. F. & Spitalnik, S. L. 1995. The P blood group system: Biochemical, serological, and clinical aspects. *Transfusion Medicine Reviews*, 9, 110-122.

Yang, S.-T., Kiessling, V., Simmons, J. A., White, J. M. & Tamm, L. K. 2015. HIV gp41mediated membrane fusion occurs at edges of cholesterol-rich lipid domains. *Nature Chemical Biology*, 11, 424-431.

CHAPTER 7

GENERAL DISCUSSION AND CONCLUSION

Aims and Objectives

The aim of this project was to study the relationship between blood groups and HIV-infection. HIV infection in Botswana has not been linked to particular blood groups. In fact, the distribution of blood groups other than ABO in Botswana is not known. Furthermore, very few studies elsewhere have addressed this matter and the few that have been done only looked at ABO and Rh(D) blood groups. This is hitherto the most exhaustive study of the role of various blood groups in HIV pathology and epidemiology. The study therefore aimed to:

- Determine the frequencies of blood groups in the Botswana population with respect to the following antigens among uninfected controls and HIV-infected individuals: A, B, "O", D, C, c, E, e, Fy^a, Fy^b, Jk^a, Jk^b, Kp^a, Kp^b, Le^a, Le^b, Lu^a, Lu^b, M, N, P₁, S, and s.
- 2. Compare the occurrence of blood groups in HIV-infected versus uninfected individuals.
- 3. Determine, where possible, the underlying mechanisms.

The frequency of the selected blood groups in Botswana were determined and found to be compatible with documented frequencies for Africans in general. The occurrence of HIV infection in these blood groups was also compared and the antigens Rh2, Lu^b and P_1 were found to be associated with infection as detailed below.

ABO Blood Groups

Many studies have reported a relationship between the risk of HIV-infection and some blood groups (Abdulazeez et al., 2008, Anstee, 2010, Branch, 2010, Neil et al., 2003). In one study, naturally-occurring ABO antibodies were reported to neutralize HIV from lymphocytes of group A blood donors (Arendrup et al., 1991). According to this report, it would be expected that ABO antibodies should offer some level of protection against HIV across ABO blood groups (Neil et al., 2003, Neil et al., 2005). However, there is no consensus on the role of ABO blood groups with respect to the risk of HIV infection (Nneli et al., 2004, Abdulazeez et al., 2008).

We compared the prevalence of 23 blood group antigens in HIV-infected and uninfected individuals in Botswana. Our study did not find any relationship between risk of HIV-infection and any ABO blood group. In particular, none was found for blood group O, which produces both anti-A and anti-B, and should be expected to be least susceptible to HIV-infection if these conferred any level of protection. It is noted, however, that blood group O is the most prevalent among the subjects of the current study, and therefore infections between group O individuals would not be affected by presence of naturally occurring anti-A or anti-B. Reports concerning susceptibility of ABO blood groups to HIV infection therefore vary extensively. Considering the inadequate statistical analysis employed in some of the studies that reported such a

relationship (Ukaejiofo and Nubila, 2006, Sayal et al., 1996, Nneli et al., 2004), it is probably safe to conclude that no data exists to support the role of ABO blood groups in affecting the outcome of HIV exposure.

Blood Group C (rh', Rh2 or RHCE*C)

Of all the antigens studied, the blood groups C (Rh2) and E (Rh3) in the Rh system were associated with a reduced risk. However, with rigorous statistical analysis, the carriage of blood group E did not sustain significance as a protective surface molecule. We attributed this initial significance to its simultaneous co-expression and co-localization with C (Ripoche et al., 2004). Blood group C was associated with a 40% risk reduction for HIV infection. The association of this blood group to potentially-enhanced cellular cytotoxicity is a novel finding in this current study which has not been reported elsewhere in the literature. Other than its association to urea transporter proteins, the function of group C antigens is unknown. We provide here a potential immune functional role of an antigen in the Rh blood group system.

Further analysis revealed that the frequency of a high absolute CD8 count among HIV-infected patients was higher among C-positive subjects, indicating a heightened cell-mediated cytotoxicity. This is the major mode by which the immune system deals with cells that present non-self-antigens such as virally-infected or cancer cells (Gulzar and Copeland, 2004, Migueles et al., 2002). The evidence gleaned from this study suggests that the protective effect of blood group C emanates from its ability to eliminate the infection via cell-mediated immunity.

This immune function, hitherto unreported, is likely associated with CD47, a glycoprotein that is spatially associated with Rh antigens on the erythrocyte membrane (Dahl et al., 2003) and functions to enhance antigen presentation (Piccio et al., 2005) and cytotoxicity (Seiffert et al., 2001, Ticchioni et al., 2001). It is likely that the C-antigen accelerates the effectiveness of CD47, since other Rh antigens that are known to be spatially associated with CD47 (E, e, c, Duffy) did not show any evidence to suggest a protective role.

It is noted, however, that Rh antigens are expressed primarily on erythrocytes (Anstee and Tanner, 1993), and that erythrocytes are not considered immunologically active cells. This notwithstanding, it is an established fact that erythrocytes bind HIV by various means (Beck et al., 2013, Garcia et al., 2012, Hess et al., 2002, Horakova et al., 2004). HIV also binds to other cellular components such as platelets (Beck et al., 2013), and these antigens are presented to cytotoxic T-cells via MHC class I restriction (Chapman et al., 2012, Zufferey et al., 2015) to provoke a cell-mediated immune response. Erythrocytes and platelets are the most numerous blood cells. The amount of HIV bound to these components by various mechanisms is not only substantial, but erythrocytes preferentially capture the infectious virions (Beck et al., 2009, Beck et al., 2013) and present these via MHC class I restriction (Chapman et al., 2012, Zufferey et al., 2009, Beck et al., 2013) and present these via MHC class I restriction (Chapman et al., 2012, Zufferey et al., 2009, Beck et al., 2013). Dendritic cells also present exogenous viral antigens by a special cytotoxic MHC-I pathway (Buseyne et al., 2001, Moris et al., 2004). However, since Rh antigens are not expressed on leukocytes, the anti-viral mode associated with Rh(C) is more attributable to erythrocyte associated virions, where they are captured and concentrated on the erythrocyte membrane (Beck et al., 2009). Moreover, since erythrocytes bind the most infective form of

the virus (Beck et al., 2013, Beck et al., 2009), and then subject it to cell-mediated cytotoxicity (Alter et al., 2002, Buseyne et al., 2001), it is plausible that a blood group that facilitates cell-mediated immunity would prevent establishment of an infection.

During blood grouping with anti-C, the strongest agglutination reactions (4+) were more common among uninfected than infected individuals. Since Rh antibodies react stronger with homozygous cells, which have higher antigen density than heterozygous ones, we further examined the data for homozygous carriage of Rh(C). Here, only six individuals out of 409 were homozygous for the C antigen. Interestingly, none of these six were infected with HIV. It would be interesting in a large sample study to see if individuals who are homozygous for blood Rh2 ever get infected with HIV. A recent study in South Africa reported a zero probability for infection in homozygous C individuals (Davison et al., 2018). Particularly interesting would be the Rh2 profile of non-progressors and elite controllers. Our data suggests that they would be mostly Rh2-positive and more likely to be homozygous for the antigen.

This observation suggests that the protection is related to the density of antigens on the erythrocytes. From a population genetics perspective, this is a blood group with a low frequency among individuals of African origin (27%), compared to Northern Indians (85%) or Caucasians (68%). These findings are important from an epidemiological perspective as the rarity of Rh2 among Africans may have contributed to the severity of the epidemic in the continent (Motswaledi et al., 2016). In fact a close examination of the world statistics demonstrates that HIV infections are high only in those regions of the world where Rh(C) frequency is low (Table 7.1).

Duffy Blood Group

Another blood group system implicated in HIV-infection is the Duffy blood group system. A fierce debate has previously raged between investigators whose data suggested an increased risk (Lachgar et al., 1998, He et al., 2008) while others suggested a protective role (Ramsuran et al., 2011) and others argued for no effect at all (Winkler et al., 2009). Our study did not find any relationship of the Duffy antigens to the risk of HIV infection. However, we note that the Duffy antigen co-localizes on the erythrocyte membrane with the C antigen (Ripoche et al., 2004, Anstee and Tanner, 1993), which was found to be protective in this study (Motswaledi et al., 2016).

Lutheran Blood Group

The Lutheran antigens are also known as basal cell adhesion molecules (BCAM) (Chang et al., 2017). Lu^b, but not Lu^a, was associated with a three-fold risk for HIV-infection (Motswaledi et al., 2016). This antigen has wide tissue distribution and is present on monocytes, erythrocytes and endothelial cells (Parsons et al., 1995). It is a natural ligand for laminin, an intercellular matrix protein that has been implicated in vascular occlusion occasioned by adhesion of sickled erythrocytes to vascular endothelia. The role of adhesion molecules in cardiovascular disease is an active area of research (Østerud and Bjørklid, 2003, Lafrenie et al., 1996, Pedraza et al., 2000, Weerasinghe et al., 1998). Monocytes not only adhere to laminin, but also migrate on it into the tissues (Pedraza et al., 2000). For this reason, Lutheran blood groups are an important

consideration in cardiovascular disease. The most significant finding in this study was a lower absolute monocyte count in individuals positive for Lu^b antigen.

The correlation between total cholesterol and phagocytic cells was also studied. As reported by others, cholesterol levels correlated negatively with phagocytes (Solomkin et al., 2007, Frostegard et al., 1990). This cholesterol-related phagocyte depletion was present only in HIV infection and probably reflects the other aspect of atherosclerosis – inflammation, resulting from chronic HIV infection. Considering that atherogenesis requires both macrophages (transformed monocytes), cholesterol and inflammation, we observe here a combination of these factors in a fashion that logically favours atherogenesis. Moreover, we note that this interaction is enhanced in Lu^b-positive subjects but absent in Lu^b-negatives. This evidence strongly suggests that Lu/BCAM likely plays a role in endothelial retention of monocytes that leads to atherogenesis, and that such adhesion may be a function of non-HDL cholesterol.

Moreover, since HIV-infected monocytes have enhanced epithelial transmigration, it is probable that the presence of the Lu/BCAM adhesion molecule enhances this process and results in the introduction of the virus to anatomical sites such as the lymph nodes, where HIV-susceptible cells are concentrated. This would explain the observed, increased risk associated with this blood group.

Since the negative correlation between cholesterol and phagocyte counts was not observed in uninfected individuals, we conclude that HIV or its effects, especially its inflammatory effects, are responsible for promoting leukocyte depletion in the presence of cholesterol. This, if confirmed by further investigations, could explain the increased risk of cardiovascular diseases in HIV infected patients.

The P₁ Blood Group.

A novel finding regarding the P1 blood group in this study was its association with risk of HIV infection and the presence of HIV RNA in the lysate of P_1 erythrocytes. The P_1 blood group is comprised of glycosphingolipids, which are known to facilitate viral fusion independent of coreceptor expression (Hammache et al., 1999) or enhance cofactor efficiency (Puri et al., 1998).

The current study corroborated the risk associated with the glycosphingolipid P_1 . This blood group was associated with a two-fold odds ratio for HIV infection. Moreover, the study confirmed association of the virus with erythrocytes even in those patients with undetectable plasma viral load. The erythrocyte-associated virus appears to be independent of the viral load as it was detectable in those with undetectable plasma viral load and absent in some with high viral load. The detection of viral RNA in the lysate of erythrocytes suggests presence of the virus inside the erythrocytes, which was another novel finding of the study.

Combination of risk of HIV-1 Infection based on the ratio of the risk to protective blood group.

Table 7.1 presents frequencies of the protective RHCE*C and the risky P_1 . In Africa, P_1 frequencies were only available for Uganda and Botswana. The ratio of the frequencies of the risky over the protective blood groups indicates Uganda as the most HIV-risk-prone population followed by Botswana. For comparison sake, the frequency ratio for China is used as the denominator. The least vulnerable, according to this model would be Turkey, Laos, Malaysia and China. Other countries have a ratio less than one, indicating a protective advantage. The Lu^b blood group was excluded from this analysis since it occurs in high frequency across most ethnic groups (Karamatic et al., 2003).

Country/Community	RHCE*C	P ₁	P1/RHCE*C	Risk	HIV
	Frequency	Frequency	ratio	Factor	Prevalence
				based	
				on	
				China	
Botswana	29.70	74.6	2.51	5.8	21.90
Uganda	3.62	89	24.59	57	6.5*
China	88.1	37.82	0.43	1	0.04
Laos	60.30	18.97	0.31	0.72	0.30
Malaysia	96	40	0.42	1	0.4
Northern India	82.02	71.92	0.88	2.0	0.26
Thailand	51.50	30	0.58	1.3	1.10
Turkey	43	0.5	0.01	0.02	**

Table 7.1: Comparison of the risk of HIV-1 Infection based on the ratio of the risk to protective blood group.

*1994 prevalence data = 24.8% (Kilian et al., 1999)

**Official statistics not available, but HIV not considered a threat(Ay and Karabey, 2006).

Conclusion

Taking all data into account, human blood groups, especially Lu^b , and P_1 , were observed to increase susceptibility to HIV infection with data consistent with their known biochemical properties. Particularly novel in this study is the finding of viral RNA in erythrocyte lysate since HIV has not been reported to enter erythrocytes. Moreover, Rh(C) was found to be protective against HIV and was for the first time linked to an immunological function, particularly involving CD8+ lymphocytes. Our data also shows that a combination of the risky blood groups and the protective activity may influence the severity of the HIV epidemic in a region, especially exemplified by analysis of reports on the prevalence of HIV and the distribution of the blood groups in Africa and other parts of the world. This, to our knowledge is the first comprehensive documentation of the role of blood group systems in HIV immunology.

The study was limited in that while it documented the association of RH2 with cell-mediated immunity, no specific CD functions were investigated. Future experiments will need to include cytotoxicity studies that include measurement of granzyme and interferon.

References

Abdulazeez, A., Alo, E. & Rebecca, S. 2008. Carriage rate of human immunodeficiency virus (HIV) infection among different ABO and Rhesus blood groups in Adamawa state, Nigeria. *Biomedical Research*, 19, 41-44.

Alter, G., Merchant, A., Tsoukas, C. M., Rouleau, D., Leblanc, R. P., Côté, P., et al. 2002. Human Immunodeficiency Virus (HIV)-Specific Effector CD8 T Cell Activity in Patients with Primary HIV Infection. *Journal of Infectious Disease*, 185, 755-765.

Anstee, D. J. 2010. The relationship between blood groups and disease. *Blood*, 115, 4635-4643

Anstee, D. J. & Tanner, M. J. 1993. Biochemical aspects of the blood group Rh (rhesus) antigens. *Baillieres Clinical Haematology*, 6, 401-22.

Arendrup, M., Hansen, J. E., Clausen, H., Nielsen, C., Mathiesen, L. R. & Nielsen, J. O. 1991. Antibody to histo-blood group A antigen neutralizes HIV produced by lymphocytes from blood group A donors but not from blood group B or O donors. *AIDS*, 5, 441-4.

AY, P. & KARABEY, S. 2006. Is there a "hidden hiv/aids epidemic" in turkey: the gap between the numbers and the facts. *Marmara Medical Journal*, 19, 90-97

Beck, Z., Brown, B. K., Wieczorek, L., Peachman, K. K., Matyas, G. R., Polonis, V. R., Rao, M. & Alving, C. R. 2009. Human erythrocytes selectively bind and enrich infectious HIV-1 virions. *PLoS One*, 4.

Beck, Z., Jagodzinski, L. L., Eller, M. A., Thelian, D., Matyas, G. R., Kunz, A. N., et al. 2013. Platelets and erythrocyte-bound platelets bind infectious HIV-1 in plasma of chronically infected patients. *PLoS One*, *8*, e81002.

Branch, D. R. 2010. Blood groups and susceptibility to virus infection: new developments. *Current Opinion in Hematology*, 17, 558–564.

Buseyne, F., Le Gall, S., Boccaccio, C., Abastado, J. P., Lifson, J. D., Arthur, L. O., et al. 2001. MHC-I-restricted presentation of HIV-1 virion antigens without viral replication. *Nature Medicine*, 7, 344-9.

Chang, H.-Y., Chang, H.-M., Wu, T.-J., Chaing, C.-Y., Tzai, T.-S., Cheng, H.-L., et al. 2017. The role of Lutheran/basal cell adhesion molecule in human bladder carcinogenesis. *Journal of Biomedical Science*, 24, 61.

Chapman, L. M., Aggrey, A. A., Field, D. J., Srivastava, K., Ture, S., Yui, K., Topham, D. J., Baldwin, W. M. & Morrell, C. N. 2012. Platelets present antigen in the context of MHC class I. *Journal of Immunology*, 189, 916-923.

Dahl, K. N., Westhoff, C. M. & Discher, D. E. 2003. Fractional attachment of CD47 (IAP) to the erythrocyte cytoskeleton and visual colocalization with Rh protein complexes. *Blood*, 101, 1194-1199.

Davison, G. M., Hendrickse, H. L. & Matsha, T. E. 2018. The relationship between immunogenic red blood cell antigens and Human Immunodeficiency Virus infection. *Transfusion and Apheresis Science*, 57, 58-62.

Frostegard, J., Nilsson, J., Haegerstrand, A., Hamsten, A., Wigzell, H. & Gidlund, M. 1990. Oxidized low density lipoprotein induces differentiation and adhesion of human monocytes and the monocytic cell line U937. *Proceedings of the. National. Academy of. Science. USA*, 87, 904-908.

Garcia, M. N., Dos Ramos Farias, M. S., Fazzi, L., Grasso, D., Rabinovich, R. D. & Avila, M. M. 2012. Presence of IgG anti-gp160/120 antibodies confers higher HIV capture capacity to erythrocytes from HIV-positive individuals. *PLoS One*, 7, e45808.

GULZAR, N. & COPELAND, K. F. 2004. CD8+ T-cells: function and response to HIV infection. *Current HIV Research*, 2, 23-37.

Hammache, D., Yahi, N., Maresca, M., Pie'Roni, G. & Fantini, J. 1999. Human erythrocyte glycosphingolipids as alternative cofactors for human immunodeficiency virus type 1 (HIV-1) entry: evidence for CD4-induced interactions between HIV-1 gp120 and reconstituted membrane microdomains of glycosphingolipids (Gb3 and GM3). *Journal of Virology*, 73, 5244–5248.

He, W., Neil, S., Kulkarni, H., Wright, E. & Agan, B. K., Et Al 2008. Duffy antigen receptor for chemokines mediates trans-infection of HIV-1 from red blood cells to target cells and affects HIV-AIDS susceptibility. *Cell Host Microbe*, *4*, 52–62.

Hess, C., Klimkait, T., Schlapbach, L., Zenero, V. D., Sadallah, S. & Horakova, E. 2002. Association of a pool of HIV-1 with erythrocytes in vivo: a cohort study. *Lancet*, 359, 2230-4. Horakova, E., Gasser, O., Sadallah, S., Inal, J., Bourgeois, G., Ziekau, I., et al. 2004. Complement mediates the binding of HIV to erythrocytes. *Journal of Immunology*, 173, 4236-41.

Karamatic, C. V., Carole, G. & Geoff, D. 2003. Molecular bases of the antigens of the Lutheran blood group system. *Transfusion*, 43, 1729-1737.

Kilian, A. H., Gregson, S., Ndyanabangi, B., Walusaga, K., Kipp, W., Sahlmüller, G., et al. 1999. Reductions in risk behaviour provide the most consistent explanation for declining HIV-1 prevalence in Uganda. *AIDS*, 13, 391-398.

Lachgar, A., Jaureguiberry, G., Le Buenac, H., Bizzini, B., Zagury, J. F., Rappaport, J. & Zagury, D. 1998. Binding of HIV-1 to RBCs involves the Duffy antigen receptors for chemokines (DARC). *Biomedicine and Pharmacotherapy*, 52, 436-439.

Lafrenie, R. M., Wahl, L. M., Epstein, J. S., Hewlett, I. K., Yamada, K. M. & Dhawan, S. 1996. HIV-1-Tat modulates the function of monocytes and alters their interactions with microvessel endothelial cells. A mechanism of HIV pathogenesis. *Journal of Immunology*, 156, 1638-1645.

Migueles, S. A., Laborico, A. C., Shupert, W. L., Sabbaghian, M. S., Rabin, R., Hallahan, C. W., et al. 2002. HIV-specific CD8+ T cell proliferation is coupled to perforin expression and is maintained in nonprogressors. *Nature immunol*, *3*, 1061-1068.

Moris, A., Nobile, C., Buseyne, F., Porrot, F., Abastado, J.-P. & Schwartz, O. 2004. DC-SIGN promotes exogenous MHC-I–restricted HIV-1 antigen presentation. *Blood*, 103, 2648-2654.

Motswaledi, M. S., Kasvosve, I. & Oguntibeju, O. O. 2016. Blood Group Antigens C, Lub and P1 may have a role in HIV infection in Africans. *PLoS ONE*, 11, e0149883.

Neil, S. J., Magre, S., Mcknight, A. & Weiss, R. A. 2003. *A-B-O blood groups and HIV-1 infection* [Online]. NLM Gateway; A service of the U.S. National Institutes of Health. Available: <u>http://gateway.nlm.nih.gov/MeetingAbstracts/ma?f=102262439.html</u>.

Neil, S. J., Mcknight, A., Gustafsson, K. & Weiss, R. A. 2005. Hiv-1 incorporates ABO histoblood group antigens that sensitize virions to complement-mediated inactivation. *Blood*, 105, 4693-9.

Nneli, R. O., Ekpo, B. O., Egene, J. & Ohaeri, O. C. 2004. Prevalence of Rh and ABO blood groups in HIV seropositive pregnant women in Enugu, Nigeria. *Nigerian Journal of Physiological Science*, 19, 7-9.

Østerud, B. & Bjørklid, E. 2003. Role of Monocytes in Atherogenesis. *Physiological Reviews* 83, 1069-1112.

Parsons, S. F., Mallinson, G., Holmes, C. H., Houlihan, J. M., Simpson, K. L., Mawby, W. J., et al. 1995. The Lutheran blood group glycoprotein, another member of the immunoglobulin superfamily, is widely expressed in human tissues and is developmentally regulated in human liver. *Proceedings of the National Academy of Science*, 92, 5496-5500.

Pedraza, C., Geberhiwot, T., Ingerpuu, S., Assefa, D., Wondimu, Z., Kortesmaa, J., Tryggvason, K., Virtanen, I. & Patarroyo, M. 2000. Monocytic cells synthesize, adhere to, and migrate on laminin-8 ($\alpha 4\beta 1\gamma 1$). *Journal of Immunology*, 165, 5831–5838.

Piccio, L., Vermi, W., Boles, K. S., Fuchs, A., Strader, C. A., Facchetti, F., et al. 2005. Adhesion of human T cells to antigen-presenting cells through SIRPβ2-CD47 interaction costimulates T-cell proliferation. *Blood*, 105, 2421-2427.

Puri, A., Hug, P., Jernigan, K., Barchi, J., Kim, H.-Y., Hamilton, J., et l. 1998. The neutral glycosphingolipid globotriaosylceramide promotes fusion mediated by a CD4-dependent CXCR4-utilizing HIV type 1 envelope glycoprotein. *Proceedings of the National Academy of Science*, 95, 14435-14440.

Ramsuran, V., Kulkarni, H., He, W., Mlisana, K., Wright, E. J., Werner, L., et al. 2011. Duffy-Null–Associated Low Neutrophil Counts Influence HIV-1 Susceptibility in High-Risk South African Black Women. *Clinical Infectious Disease*, 52, 1248–1256.

Ripoche, P., Bertrand, O., Gane, P., Birkenmeier, C., Colin, Y. & Cartron, J.-P. 2004. Human Rhesus-associated glycoprotein mediates facilitated transport of NH3 into red blood cells. *Proceedings of the National Academy of Science, USA*, 101, 17222-17227.

SAYAL, S. K., DAS, A. L. & NEMA, S. K. 1996. Study of blood groups in HIV seropositive patients. *Indian Journal of Dermatology, Venereology and Leprology*, 62, 295-7. Seiffert, M., Brossart, P., Cant, C., Cella, M., Colonna, M., Brugger, W., et al. 2001. Signal-regulatory protein α (SIRPα) but not SIRPβ is involved in T-cell activation, binds to CD47

with high affinity, and is expressed on immature CD34⁺CD38⁻hematopoietic cells. *Blood*, 97, 2741-2749.

Solomkin, J. S., Robinson, C. T., Cave, C. M., Ehmer, B. & Lentsch, A. B. 2007. Alterations in membrane cholesterol cause mobilization of lipid rafts from specific granules and prime human neutrophils for enhanced adherence-dependent oxidant production. *Shock*, 28, 334-338.

Ticchioni, M., Raimondi, V., Lamy, L., Wijdenes, J., Lindberg, F. P., Brown, E. J., et al. 2001. Integrin-associated protein (CD47/IAP) contributes to T cell arrest on inflammatory vascular endothelium under flow. *The FASEB Journal*, 15, 341-350.

Ukaejiofo, E. O. & Nubila, T. 2006. Association between ABO, Rhesus blood group systems and hemoglobin genotype among confirmed HIV/AIDS TB-coinfected patients in Enugu Urban, Nigeria. *West African Journal of Medicine*, 25, 61-64.

Weerasinghe, D., Mchugh, K. P., Ross, F. P., Brown, E. J., Gisler, R. H. & Imhof, B. A. 1998. A Role for the $\alpha\nu\beta3$ Integrin in the Transmigration of Monocytes. *The Journal of Cell Biology*, 142, 595-607.

Winkler, C. A., An, P., Johnson, R., Nelson, G. W. & Kirk, G. 2009. Expression of Duffy antigen receptor for chemokines (DARC) has no effect on HIV-1 acquisition or progression to AIDS in African Americans. *Cell Host Microbe*, *5*, 411-413.

Zufferey, A., Speck, E. R., Guo, L., Aslam, R., Kapur, R. & Semple, J. W. 2015. Murine Bone Marrow-Derived Megakaryocytes Are Capable of Antigen Cross-Presentation on Major Histocompatibility Class (MHC) I Molecules. *Blood*, 126, 3465-3465.

CHAPTER 8

RECOMMENDATIONS

Blood Group C

- This blood group was protective and showed an increased proportion of cytotoxic T cells. Moreover, the frequency of this antigen was consistently low in countries with high HIV rates of infection. However, HIV-specific CD8 T-cell responses were not measured in this study to confirm enhanced cytotoxicity. A study is hereby recommended to investigate HIV-1-specific T cell responses such as Interferon gamma response to stimulation with HIV-specific peptides. This would confirm the functional role of the Rh2 antigen in the immunological response to HIV.
- 2. Most countries in Africa do not have data relating to these antigens that relate to HIV susceptibility. A multi-national study is needed to corroborate our findings, similar to the South African study cited in this document.
- 3. Another pertinent study would be to determine the RhC frequency among HIV-infected non-progressors and elite controllers.
- 4. Other profitable studies could investigate how erythrocyte-bound virions are presented to T-cells and the role, if any, of CD47 in enhancing the presentation of such antigens.
- 5. The use of homozygous Rh2 hematopoietic stem cells may benefit those who are Rh2 homozygotes. These would not reject the Rh2 cells but would increase the antigen expression and consequent resistance to HIV. Moreover, a more permanent vaccine may be possible by introducing hematopoietic stem cells to developing foetuses who are likely to take adopt the immune hematopoietic cells and acquire HIV-resistance.

Lutheran Blood group (Lu/BCAM, LU2 or Lu^b)

- 1. The results concerning this blood group showed that the Lutheran blood group, in the presence of high cholesterol levels, promotes depletion of circulating monocytes. This phenomenon constitutes an atherogenic environment. Recommended studies include the use of a larger sample size to study the relationship between high lipids, inflammation, LU2 antigen carriage, and carotid intima/medial thickness. This could help in establishing the genetic link of the Lutheran blood group to cardiovascular disease in HIV-infected individuals.
- 2. Other studies could also focus on the effect of the blood group on monocyte tethering, migration, as well as the combined atherogenic effect of HIV infection and Lu^b.
- 3. Furthermore, measures of monocyte activation such as soluble CD163 (sCD163) could be used to study monocyte activation between Lu^b-positive and negative individuals.

P₁ Blood Group

1. The P₁ blood group was associated with a high risk for HIV infection. This was attributed to its ability to facilitate viral fusion to erythrocytes. However, this study used

only a small sample size to demonstrate viral RNA in erythrocytes. Large scale studies are needed to explore the *in vitro* fusogenic effect of the blood group.

2. If corroborated by such studies, the findings would pave the way for further studies to determine if such intra-erythrocytic RNA has potential to infect splenic macrophages when they phagocytose senescent erythrocytes, which could eventually lead to the development of strategies for dealing with this viral reservoir. In particular, the results hold promise for using RNase-laden glycosphingolipid liposomes to capture and lyse circulating virions as a way to effect a rapid drop in the viral load.

APPENDICES

APPEDNDIX A - University of Botswana IRB Permit



UNIVERSITY OF BOTSWANA

Office of the Deputy Vice Chancellor (Academic Affairs)

Office of Research and Development

Corner of Notwane and Mobuto Road, Gaborone, Botswana

Pvt Bag 00708 Tel: [267] 355 2900 Gaborone, Botswana Fax: [267] 395 7573 E-mail: research@mopipi.ub.bw

Ref: URB/IRB/1365

30th November 2012.

The Permanent Secretary Ministry of Health Private Bag 0038 Gaborone, Botswana

RE: APPLICATION FOR A RESEARCH PERMIT: MODISA S. MOTSWALEDI

Since it is a requirement that everyone undertaking research in Botswana should obtain a Research Permit from the relevant arm of Government, The Office of Research and Development at the University of Botswana has been tasked with the responsibility of overseeing research at UB including facilitating the issuance of Research permits for all UB Researchers inclusive of students

I am writing this letter in support of an application for a research permit by Mr Modisa Motswaledi an academic staff member in the Faculty of Health Sciences. Mr Motswaledi has proposed to conduct a study titled "The role of blood groups in preventing or enhancing HIV infection in Botswana."The main aim of this project is to study the relationship between human blood groups and susceptibility to HIV-infection. It will also compare the viral binding ability of specific blood groups and how such binding impacts haematological and biochemical prognostic indicators in HIV-infected patients. The findings of this study may be used in future to improve the care of people who are at risk of, or are infected with HIV

The Office of Research and Development is satisfied with the process for data collection, analysis and the intended utilisation of findings from this research. We will appreciate your kind and timely

We thank you for your usual cooperation and assistance

Sincerely,

Dr M.B.M. Sekhwela

For Director, Office of Research & Development Encls: Completed Application for Research Permit Informed Consent document Data collection tools Comments from the UB IRB

www.ub.bw

APPENDIX B – Ministry of Health Permit

Telephone: (267) 363200 FAX (267) 353100 TELEGRAMS: RABONGAKA TELEX: 2818 CARE BD



MINISTRY OF HEALTH PRIVATE BAG 0038 GABORONE

REPUBLIC OF BOTSWANA

REF NO: PPME-13/18/1 Vol VII (698)

05 December 2012

Mr Modisa Sekhamo Motswaledi Pvt Bag 00708 University of Botswana

Dear Sir/Madam

PERMIT: THE ROLE OF BLOOD GROUPS IN PREVENTING OR ENHANCING HIV INFECTION IN BOTSWANA

Reference is made to your application submitted to the Health Research Unit (HRU) for permission to conduct above mentioned study. The HRU reviewed and approved the study.

Permission is therefore granted to conduct the above study. This approval is valid for a period of 1 year effective 05 December 2012.

You are requested to submit at least one hardcopy and an electronic copy of the report to the Health Research and Development Division, Ministry of Health within 3 months of completion of the study. Copies should also be submitted to all other relevant authorities.

Yours sincerely

her n

P. Khulumani
FOR PERMANENT SECRETARY



1

APPENDIX C – CPUT ETHICS PERMIT



HEALTH AND WELLNESS SCIENCES RESEARCH ETHICS COMMITTEE (HW-REC)

Registration Number NHREC: REC- 230408-014

P.O. Box 1906 • Bellville 7535 South Africa Symphony Road Bellville 7535 •Tel: +27 21 959 6352 • Fax +27 21 953 8490 Email: danielso@cput.ac.za

> 26 April 2013 CPUT/HW-REC 2013/H17

Faculty of Health and Wellness Sciences - Biomedical Sciences Department

Dear Modisa Sekhamo Motswaledi

APPLICATION TO THE HW-REC FOR ETHICAL CLEARANCE

Approval was granted on 24 April 2013 by the Health and Wellness Sciences-REC to Modisa Sekhamo Motswaledi for your Ethical Clearance application (pending corrections that have been received and reviewed-if-applicable). This approval is for research-activities related to a DTech: Biomedical Technology at this institution.

TITLE: The role of blood groups in preventing or enhancing HIV infection in Botswana.

Internal Supervisor:	Prof O Oguntibeju
External Co-supervisor:	Prof I Kasvosve

Comment:

Approval will not extend beyond 26 April 2014. An extension should be applied for 6 weeks before this expiry date should data collection and use/analysis of data, information and/or samples for this study continue beyond this date.

Note:

The investigator(s) should understand the conditions under which they are authorized to carry out this study and they should be compliant to these conditions. It is required that the investigator(s) complete an annual progress report that should be submitted to the HW-REC in December of that particular year, for the HW-REC to be kept informed of the progress and of any problems you may encounter.

Kind Regards

Zuleika Nortjé

CHAIRPERSON – ETHICS RESEARCH COMMITTEE FACULTY OF HEALTH AND WELLNESS SCIENCES TELEPHONE: (+267) 3974 482

FAX : (+267)3974 494



National Health Laboratory Private Bag 0057 GABORONE Botswana

REPUBLIC OF BOTSWANA

TEL: 3686700

30th January 2013

Ref: NBTS 007/01/13

Mr Modisa S. Motswaledi University of Botswana Department of Medical Laboratory Sciences

Dear Sir

RE: PERMISSION TO USE RESIDUAL BLOOD DONOR SAMPLES

Permission is granted to you to use residual blood samples in National Blood Transfusion Centre (EDTA samples) however you will be collecting them on the seventh day when they are due to be discarded.

Thank you

gmm

Yours faithfully

Mr K. Mosenki / for Ag. Medical Director

APPENDIX E – BOTSWANA-HARVARD HIV REFERENCE LABORATORY PERMIT



29th January 2013

Mr Modisa Motswaledi Department of Medical Laboratory Science Faculty of Health Sciences University of Botswana P/Bag UB 00712 Gaborone

Dear Mr Motswaledi,

Re: The Role of Blood Groups in Preventing or Enhancing HIV Infection in Botswana.

Thank you for your detailed proposal to undertake a study on **"The Role of Blood Groups in Preventing or Enhancing HIV Infection in Botswana"**. The BHP recognizes the importance of studying various factors that are important in HIV Infection and how those can inform new strategies for prevention of HIV transmission in Botswana.

This letter is in full support of the proposed research study being conducted at the Botswana Harvard HIV Reference Laboratory.

This research will be conducted under the agreement for collaborative research on HIV/AIDS in Botswana between the Botswana Harvard Partnership and The University of Botswana.

I take this opportunity to wish you success on your study and look forward to finalizing the specific details relating to conducting the study.

Yours Sincerely,

sa er

Dr M. J. Makhema CEO Botswana Harvard Aids Institute Partnership

Private Bag BO 320 Gaborone, Botswana Phone: +267 - 3902671 Fax: +267 - 3901284 Website: www.bhp.org.bw

ADDENDUM: RESEARCH OUTPUT

Published articles

- MOTSWALEDI, M. S., KASVOSVE, I. & OGUNTIBEJU, O. O. 2013. The Role of Red Blood Cells in Enhancing or Preventing HIV Infection and Other Diseases. BioMed Res Intern.Volume 2013, Article ID 758682, 7 pages. Available at: <u>http://dx.doi.org/10.1155/2013/758682</u>.
- 2. MOTSWALEDI, M. S., KASVOSVE, I. & OGUNTIBEJU, O. O. 2016. Blood Group Antigens C, Lub and P1 may have a role in HIV infection in Africans. PLoS ONE, 11, e0149883.
- 3. Motswaledi[,] MS, Kasvosve, I & Oguntibeju OO (2019). Rh2 Phenotype Affects Cytotoxic T- Cell Counts. *Jordan Journal of Biological Sciences* 12(3). In Press

Articles submitted to journals

- 1. MOTSWALEDI, M. S., KASVOSVE, I. & OGUNTIBEJU, O. O. 2018. Potential role of Lu/BCAM in HIV-related atherosclerosis. Submitted to *African Journal of Laboratory Medicine*
- 2. MOTSWALEDI, M. S., KASVOSVE, I. & OGUNTIBEJU, O. O. 2018. HIV-RNA is detectable in P₁ erythrocyte lysate. Submitted to *Transfusion and Apheresis Science*

Conference Presentations

- MOTSWALEDI, M. S., KASVOSVE, I. & OGUNTIBEJU, O. O. 2013. Blood Groups in HIV infection. Botswana Institute for Clinical Laboratory Professionals. 2nd Scientific Conference 2016.
- MOTSWALEDI, M. S., KASVOSVE, I. & OGUNTIBEJU, O. O. 2017: Potential Mechanisms for HIV risk of infection in subjects carrying blood groups C, Lub and P₁. Botswana Institute for Clinical Laboratory Professionals – 3rd Scientific Conference, 2017.
- MOTSWALEDI, M.S., KASVOSVE, I. & OGUNTIBEJU, O. O.2018. Lack of Rh2 as a risk factor for HIV among Africans. 4th Biennial Conference of the African Society for Laboratory Medicine. ASLM Abstract Control ID: 3054843, Abuja Nigeria.